

An additional meeting of the above Committee will be held at Shire Hall, Market Place, Warwick on Monday 13 May 2024, at **4.30pm**.

Councillor K Gorman (Chair) Councillor J Sinnott (Vice-Chairman)

Councillor D Armstrong Councillor A Barton Councillor K Dray Councillor C Gifford Councillor D Harrison Councillor C King Councillor P Kohler Councillor M Luckhurst Councillor J Matecki Councillor P Redford Councillor W Roberts Councillor G Rosu Councillor S Syson

Agenda

1. **Apologies & Substitutes**

- (a) to receive apologies for absence from any Councillor who is unable to attend; and
- (b) to receive the name of any Councillor who is to act as a substitute, notice of which has been given to the Chief Executive, together with the name of the Councillor for whom they are acting.

2. **Declarations of Interest**

Members to declare the existence and nature of interests in items on the agenda in accordance with the adopted Code of Conduct.

Declarations should be disclosed during this item. However, the existence and nature of any interest that subsequently becomes apparent during the course of the meeting must be disclosed immediately. If the interest is not registered, Members must notify the Monitoring Officer of the interest within 28 days.

Members are also reminded of the need to declare predetermination on any matter.

If Members are unsure about whether or not they have an interest, or about its nature, they are strongly advised to seek advice from officers prior to the meeting.

3. Minutes of the Licencing & Regulatory Committee

To confirm the minutes of the Licensing & Regulatory Committee held on 11 March 2024.

(Pages 1 to 6)







4. Minutes of Licencing & Regulatory Panels

To confirm the minutes of the Licensing & Regulatory Panel 21 March 2024.

(Pages 1 to 4)

5. **Request to classify and permit the showing of an unclassified film.**

To consider a report from Safer Communities, Leisure and Environment.

(Pages 1 to 244)

6. **Public and Press**

To consider resolving that under Section 100A of the Local Government Act 1972 that the public and press be excluded from the meeting for the following item by reason of the likely disclosure of exempt information within the paragraphs of Schedule 12A of the Local Government Act 1972, following the Local Government (Access to Information) (Variation) Order 2006, as set out below.

Item	Paragraph	Reason
Numbers	Numbers	
7	1	Information relating to an Individual
7	2	Information which is likely to reveal the identity of an
		individual

7. Confidential Minutes of the Licensing & Regulatory Panels

To confirm the confidential minutes of the Licensing & Regulatory Panel meetings held on:

a) 29 February 2024	(Pages 1 to 5) (Not for publication)
b) 10.00am 13 March 2024	(Pages 1 to 7) (Not for publication)
c) 11.00am 13 March 2024	(To Follow)

Published Tuesday 30 April 2024

General Enquiries: Please contact the Committee Services team via email at. Alternatively, you can contact us at:

Warwick District Council, Town Hall, Parade, Royal Learnington Spa, CV32 4AT or telephone 01926 456114.

For enquiries about specific reports, please contact the officers named in the reports. You can e-mail the members of the Committee at <u>LandRCommittee@warwickdc.gov.uk</u>

Details of all the Council's committees, Councillors and agenda papers are available via our website on the <u>Committees page</u>

We endeavour to make all of our agendas and reports fully accessible. Please see our <u>accessibility statement</u> for details.

The agenda is available in large print on request, prior to the meeting, by emailing <u>committee@warwickdc.gov.uk</u> or telephoning (01926) 456114

Licensing and Regulatory Committee

Minutes of the meeting held at Shire Hall, Warwick on Monday 11 March 2024 at 4.30pm.

Present: Councillors Gorman (Chair), Barton, Dray, D Harrison, King, Kohler, Rosu and Sinnott

13. Apologies and Substitutes

- (a) apologies for absence were received from Councillors Armstrong, Syson, C Gifford, Redford and Roberts; and
- (b) there were no substitutes.

14. **Declarations of Interest**

There were no declarations of interest.

15. **Proposed Hackney Carriage Fare Increase**

The Committee considered a report from Safer Communities, Leisure, and Environment regarding requests that had been received from the taxi trade for the current Hackney Carriage fare tariffs to be increased. The current tariff had been in place since 2014 and should be reviewed in line with the current economic climate.

The ability of the Council to set Hackney Carriage fares was derived from S65 of the Local Government (Miscellaneous Provisions) Act 1976. The setting of Hackney Carriage fares was a Cabinet function, but the Cabinet were seeking the views of this Committee based upon its knowledge of licensing Hackney Carriage Drivers.

The Council should apply this power reasonably as the fixing of a market price for Hackney Carriage fares within the Council district had a direct impact on the ability for a hackney driver to make a living.

Tariff rates only applied to Hackney Carriages (HC), there was no similar power to set fares for Private Hire vehicles. Private Hire Vehicles had discretion to set whatever charges they consider reasonable, given normal market forces and business competition.

Hackney carriages could set lower fares and offer special discounts should they wish but were unable to charge more than the set fares and tariff maximum. However, when the journey goes beyond the boundary of the District, the tariffs did not apply.

Tariff rates differed according to the time of day, the number of passengers, and whether the journey took place on a bank holiday. Tariffs directly related to the unsociable hours worked by drivers, therefore ensuring as far as possible the provision of Hackney Carriage transport was always available.

The current tariff was introduced in 2014. Table 1 - Current tariff:

	Tariff 1	Tariff 2	Tariff 3	
WARWICK	4 or less	4 or less passengers	5+passengers	
DISTRICT	passengers	(Evening **)	(Evening**)	
COUNCIL	(Day*)	5+ passengers (Day*)		
Distance	£3.40	£5.10	£6.80	
First 7/10 th Mile				
Subsequent Distance	£0.20	£0.30	£0.40	
1/10 th Mile				
Waiting Time	£0.10	£0.15	£0.20	
30 Seconds				
Soiling Charge		£75		
Day* = 06:00 to 22:00				
Evening = 22:00 to 06:00 and Bank Holidays **				
Except on Christmas Eve and New Years Eve = 18:00 to 06:00				

Warwick District Council currently licenced 143 vehicles as Hackney Carriage vehicles. 100 of these vehicles (69.9%) had five or more passenger seats and would use tariff 3 in the evening and on bank holidays.

Warwick District Council currently sat at number 310 out of 355 other local authorities on the National Hackney Carriage Fare Table for a two-mile fare at Tariff 1. With No.1 on the table being the most expensive and 355 the cheapest set fare in the country. Below showed a table with how WDC currently compared with our neighbouring Local Authorities.

Position in National table (As of Feb 24)	Local Authority	2 mil.Fare (in the day)	Last HC fare increase
53	Coventry	£8.00	2022
135	Nuneaton Bedworth	£7.25	2023
158	Rugby	£7.10	2022
197	North Warwickshire	£6.80	2022
235	Stratford	£6.60	2023
310	Warwick	£6.00	2014

The cost-of-living crisis, including increase in fuel and running costs (maintenance and insurance etc) was considered as the main reason why the Council had received several enquiries by the taxi trade for a fare increase.

On 11 December 2023, Cabinet agreed to a fare increase consultation. This consultation was with the owners of Hackney Carriage vehicles only as these vehicles had to use the fare structure decided by WDC. WDC Private Hire vehicles could charge whatever they felt reasonable. An online consultation took place between 18 December 2023 to 15 January 2024. 57% of the licensed HC trade responded. The results of the survey were attached as Appendix 1 to the report.

When asked what percentage increase HC vehicle owners would consider appropriate the answers ranged from 20% to over 50%.

When considering a suitable percentage increase, the office for national statistics, inflation, and prices indices (all items) data had been used as a comparison. Between 2014 and 2023 the sum of the overall Consumer Price Index- including Housing (CPIH) annual rate increase was 27.7%. Inflation had averaged 3.2% a year meaning a two-mile fare in the day that cost £6.00 in 2014 would cost £7.94 in 2023.

Following the results of the survey, the proposed new tariff was set out as Appendix 2 to the report. This would equate to a 30% increase on the current tariffs, which recognised the CPIH increase between 2014 and 2023, as well as allowing a small amount of growth for 2024. There were no proposed changes to the soiling charge or the tariff times. Below, a table of how the revised tariffs would compare with our neighbouring Local Authorities and the National Hackney Carriage Fare Table if the proposed 30% increase was implemented.

Position in National table (As of Jan 24)	Local Authority	2 mil.Fare (in the day)
53	Coventry	£8.00
83	Warwick	£7.65
135	Nuneaton Bedworth	£7.25
158	Rugby	£7.10
197	North Warwickshire	£6.80
235	Stratford	£6.60

Current Neighbouring Local Authorities Hackney Carriage tariffs were attached as Appendix 3 to the report.

If a fare increase was agreed at Cabinet, the table of fares would be advertised in the local newspaper as well as being advertised on the Council website and at the published Council's Head Office (Town Hall) Recognizing the change in locations at present, the Council would also ensure that the notice was placed on the public noticeboard outside the Town Hall and outside Riverside House.

If there were no objections, the proposed fare increase would take effect in May 2024.

If there were objections, these would need to be considered by Cabinet. At that point Cabinet could choose to make modifications to the proposed table of fares in light of the objections received or not.

For future Hackney Carriage fare increases, it was proposed that the trade would be consulted by means of a survey every 18 months from when the

last increase took place to determine if the HC trade felt a fare increase was appropriate at that time.

In terms of alternative options, the Committee could recommend to Cabinet that the proposed fares were not accepted or they could propose an alternative fare. Both of these would need to include reasons for the proposal, based on the evidence within the report.

In response to questions from Members, the Licensing Team Leader explained that:

- there were 143 Hackney Carriage vehicles in the District, and this number had stayed relatively stable over the last few years. Whilst the Council were not losing many Hackney Carriages, there had been a lot of change since Covid and the popularity of Private Hire apps;
- the proposal was to increase the current starting fare by 30%, with amounts being rounded for ease. She would look at the figure in sections 1.11 and 1.12 to ensure that they had been calculated correctly before the report went to Cabinet;
- the last fare increase was in 2014, and at the time these tariffs were relatively high. Since that time, other local authorities had increased their fares, but Warwick District Council (WDC) had not been petitioned by drivers to increase the fares too. However, going forward, increases in fares would be implemented more frequently than every decade to keep up with cost of living;
- these proposed rates were the maximum that drivers could charge, so they could operate at cheaper rates if they wished. Even with the increase, these fees were still competitive with that of Private Hire vehicles as operators were able to set their own fees;
- if a driver is WDC licensed, they were required to use the set fares. However, the set rate no longer applied once a journey left the boundary of the District;
- WDC had not had any correspondence with Stratford-on-Avon District Council (SDC) regarding their Hackney Carriage fees. SDC had an increase in 2023, but WDC's process might prompt them to review these; and
- officers would be happy to explain to the public the reasons behind the 30% increase to pre-empt any questions regarding such a large increase.

It was proposed by Councillor Kohler, seconded by Councillor King and

Recommended to Cabinet that

(1) the outcome of the fare increase survey as set out in Appendix 1 to the report be noted; and

- (2) the proposed fare tariff increase should be approved, subject to the following internal notes for officers:
 - a) the figures contained within the report, especially in sections 1.11 and 1.12, should be reviewed and made clearer ahead of the next Cabinet meeting; and
 - b) any communications via the Council's social media accounts should explain in detail to the public the reasons behind the fare increase.

16. Minutes

The minutes of the meeting of 16 October 2023 meeting were taken as read and signed by the Chairman as a correct record.

17. Minutes of the Licensing & Regulatory Panels

The minutes of following meetings were taken as read and signed by the Chairman as a correct record:

- a) 14 September 2023;
- b) 27 September 2023;
- c) 24 October 2023;
- d) 21 November 2023;
- e) 7 February 2024; and
- f) 14 February 2024.

18. **Public and Press**

Resolved that under Section 100A of the Local Government Act 1972 that the Public and Press be excluded from the meeting for the following item by reason of the likely disclosure of exempt information within paragraphs 1 & 2 of Schedule 12A of the Local Government Act 1972, following the Local Government (Access to Information) (Variation) Order 2006.

19. Confidential Minutes of the Licensing & Regulatory Panels

The minutes of the following Licensing and Regulatory Panels were taken as read and signed by the Chairman as a correct record:

- a) 5 October 2023;
- b) 10.00am 14 December 2023;
- c) 11.00am 14 December 2023;
- d) 11 January 2024;
- e) 16 January 2024;
- f) 10.00am 1 February 2024; and
- g) 11.00am 1 February 2024.

(The meeting ended at 4.56pm)

CHAIR 13 May 2024

Licensing & Regulatory Panel

Minutes of the additional Licensing & Regulatory Panel held in Saltisford One, Warwick on Thursday 21 March 2024 at 10.00am.

Present: Councillors Dray, Gorman and Syson.

Also Present: Mia Matthews (Committee Services Officer); Alaa Cronin (Council's Solicitor); Lewis Apaya (Licensing Administration Apprentice) – observing only and Emma Morgan (Licensing Enforcement Officer).

1. Appointment of Chairman

Resolved that Councillor Dray be appointed as Chairman for the hearing.

2. Substitutes

There were no substitutes.

3. Declarations of Interest

There were no declarations of interest made.

4. An application for a new Street Trading Consent

The Panel considered a report from Safer Communities, Leisure and Environment which brought forward a new application for a Street Trading Consent to trade outside the hours and restrictions set out within Warwick District Council's Street Trading Policy.

Anyone who exposes goods for sale within the District needed a Street Trading Consent issued by the Council.

CJ's Events Limited had applied for a Street Trading Consent to trade from The Pump Room Gardens on Saturday 11 May 2024 at a Eurovision Screening event from 10.00am to 12.00am. The application was for a group trader consent for up to 75 stalls. However, only the hot food and drink stall would be trading after 6:00pm. The application, a plan of the site and list of traders were attached as Appendix 1 to the report.

Upon receiving the application officers referred to the current Street Trading Consent Policy which permitted trading between the hours of 6.00am and 6.00pm. As the application was to trade beyond these hours the application had been referred to the Licensing and Regulatory Panel for a decision.

The applicant had received permission from the Landowner (Warwick District Council) to use the site and had also obtained a Temporary Event Notice issued under the Licensing Act 2003 to serve alcohol at the event with no objection received by either Warwickshire Police or Environmental Health for the Temporary Event Notice.

The event had also been through Warwick District Councils Safety Advisory Group.

The Council's Street Trading Consent Policy (2016) stated (at page 3) that:

"In considering applications for the grant or renewal of a street trading consent the following factors should be considered:

a) Public Safety

Whether the proposed activity represents or could present a risk to the public from the point of view of obstruction, fire hazard, unhygienic conditions.

b) Public Order

Whether the proposed activity presents or could present a risk to public order.

c) The Avoidance of Public Nuisance

Whether the proposed activity presents or could present a risk of nuisance to the public from noise or misbehavior, particularly in residential areas.

d) Local Area Needs

Consideration will be given to the character of the area (e.g. conservation area)"

The Street Trading Consent Policy also stated (at page 5) that:

"Before a street trading consent is granted or renewed the Licensing Team will consult with the following agencies as considered appropriate:

- Police
- Highways Department
- Environmental Health

The Council shall consider any representation made in writing to the Authority in respect of any application. The applicant will be provided with a copy of any representation received and given an opportunity to comment before a final decision is taken."

The consultation period ended on 21 March 2024.

Details of the procedure adopted by the Licensing Committee for Panel Hearings had been supplied to the applicant. The procedure was explained fully by the Council's Solicitor at the commencement of the hearing.

The Chairman introduced the Panel and officers present. The applicant introduced himself as Mr Walker, Events Organiser at CJ's Events.

At the Chairman's request, the Licensing Enforcement Officer introduced the report and confirmed that no objections had been received.

Mr Walker was invited to give a statement. Mr Walker explained that:

- CJs Events were experienced at organising events, including food festivals, across the District;
- Warwick was an inclusive and diverse District and they would like to hold an event that reflected this;
- the event would be family orientated between the hours of 10.00am -6.00pm. There would be arts and craft stalls, fair rides, and a market. These would close at 6.00pm in order to minimise noise levels;

- From 8.00pm to 12.00am the screening would start, and 27 food traders would remain. These were traders that CJs Events had worked with before and they knew to be respectful of the area;
- the Event should be cleared up completely by 1.00am;
- letters to any residents or businesses within a two-mile area would be issued informing them of the schedule for the event and including a phone number for the duty manager onsite should any issues arise; and
- CJs Events were working with Environmental Health regarding sound levels and decibel recorders would be situated around the site to record level of sound.

In response to questions from the Panel, the applicant explained that:

- work with Environmental Health was ongoing, so the exact decibels expected were not yet known. However, the system could go up to 105 decibels;
- from 11.00pm the sound would be reduced;
- the screen would be positioned to face away from residential properties;
- the traders would be expected to take all rubbish away with them. Bins from Fortress had been procured and would be collected on Monday;
- during the day the bins would be separated across the whole site, then after 6.00pm they would be moved closer to the food stalls; and
- the event would be fully staffed, and litter picks would be done during the event to ensure the site was clean. CJs Events understood the importance of keeping the Town clean and ensured they left the Town tidy following any food festival events.

The Licensing Enforcement Officer stated the event required Street Trading Consent as the event was free, if there had been a charge implemented they would not require this. They also needed to apply for consent due to requesting to trade beyond the usual hours.

The Chairman asked the applicant if there he wished to make a closing statement and Mr Walker confirmed he had nothing further to add.

At 10.18am, the Chairman asked all parties other than the Panel, the Council's Legal Advisor, and the Committee Services Officer to leave the meeting, in order to enable the Panel to deliberate in private and reach its decision. The decision would be communicated in writing via email to the applicant later the same day, followed by a written notice with a full decision within seven days.

Resolved that the consent be granted.

The Licencing Panel received a report from Warwick District Council's Licensing Department in respect of an application from CJ's Events Limited for a Multiple Trader/Special Event Street Trading Consent for up to 75 stalls to trade from The Pump Room Gardens, Leamington Spa on Saturday 11 May 2024 at a Eurovision Screening event from 10:00 hours to 00:00 hours.

In reaching a decision the Panel considered the written report, the Licensing Officer's presentation of the report and the oral representations made at the meeting by the Event Manager, Mr Walker who attended on behalf of CJ's Events Limited. The Panel also had sight of a Noise Management Plan which was submitted by the applicant to the Council's Environmental Health department (though not appended to the Licensing Officer's report) and considered the Council's "Street Trading Consent Policy" dated 2016 as well as the provisions of Schedule 4 Local Government (Miscellaneous Provisions Act) 1982.

The Panel was concerned that the event could lead to public nuisance particularly noise nuisance and increased littering. However, it gave significant weight to the fact that no representations were received from any of those consulted on the application, including the Police and Environmental Health. It was also noted that CJ's Events Limited were in on-going discussion with the Council's Environmental Health department to ensure that appropriate noise management levels are implemented at the event.

Accordingly, the Panel considered that, at the time of their decision, there were no relevant grounds on which to refuse the applications. In the circumstances, the Panel resolved to grant the Street Trading Consent. A copy of this Record of Decision will be served on all relevant parties.

Under Schedule 4 Local Government (Miscellaneous Provisions) Act 1982 there is no statutory right of appeal against this decision. An aggrieved applicant may apply to the High Court for a judicial review to challenge any errors in the administrative process. Alternatively, applicants have recourse to the Council's complaints procedure.

Cllr Dray (Chair); Cllr Syson; and Cllr Gorman

(The meeting ended at 10.26am)

CHAIRMAN 13 May 2024

Agenda Item No 5 Licensing and Regulatory Committee

Title: Request to classify and permit the showing of an unclassified film. Lead Officer: Emma Morgan <u>Emma.morgan@warwickdc.gov.uk</u> Portfolio Holder: Councillor Jim Sinnott Wards of the District directly affected: Leamington Clarendon

Approvals required	Date	Name
Portfolio Holder	25/4/2024	Jim Sinnott
Finance	25/4/2024	Chloe Evans
Legal Services	25/4/2024	Ross Chambers
Chief Executive	25/4/2024	Chris Elliott/Darren Knight
Director of Climate Change	25/4/2024	Dave Barber
Head of Service(s)	22/4/2024	Marianne Rolfe
Section 151 Officer	25/4/2024	Andrew Rollins
Monitoring Officer	25/4/2024	Graham Leach
Leadership Co-ordination Group		

Final decision by this Committee or rec to another Cttee / Council?	Yes
Contrary to Policy / Budget framework?	No
Does this report contain exempt info/Confidential? If so, which paragraph(s)?	Νο
Does this report relate to a key decision (referred to in the Cabinet Forward Plan)?	No
Accessibility Checked?	Yes

Summary

Warwick District Council Licensing Authority has received a request to permit the showing of an unclassified film.

Under the Licensing Act 2003 any film which has not been given a classification by the British Board of Film Classification (BBFC) must receive permission from the Local Authority before it may be shown in that Local Authority.

Recommendation(s)

- (1) Officers recommend that permission be given for the showing of the films, 'School Run' and 'A Green Network' in Warwick District and that the classification of 'U' is given to both films.
- (2) Subject to permission being granted the Committee classify the film in accordance with the BBFC guidelines.

1 Reasons for the Recommendation

- 1.1 On 12 April 2024 Warwick District Council Licensing Department received a request from Grand Union Productions to permit the showing of two films at The Royal Spa Centre, Newbold Terrace, Royal Learnington Spa, CV32 4HN.
- 1.2 The application form submitted by Mr Michael Clifford on behalf of Grand Union Productions is attached as **Appendix 1.**
- 1.3 The BBFC performs a national Film Classification role on behalf of the Licensing Authority, classifying films shown in Warwick District for public exhibition under the Licensing Act 2003. However, when a film is not classified by the BBFC the decision falls to the Licensing and Regulatory Committee to authorise a film to be shown in Warwick District, and to classify the film under the protection of children from harm licensing objective.
- 1.4 Section 182 Guidance issued under the Licensing Act contains some information which is particularly relevant to this decision, and this is highlighted below:
- 1.5 The protection of children from harm includes the protection of children from moral, psychological and physical harm. This includes not only protecting children from the harms associated directly with alcohol consumption but also wider harms such as exposure to strong language and sexual expletives (for example, in the context of exposure to certain films or adult entertainment). Licensing authorities must also consider the need to protect children from sexual exploitation when undertaking licensing functions. (Paragraph 2.28)
- 1.6 Under Section 20 of the Licensing Act 2003, a mandatory condition is applied to all premises issued with a premises licence which authorises the exhibition of films. The mandatory condition relates to the restriction of the admission of children. Children are defined as persons aged under 18. The condition is as follows:

The admission of children under the age of 18 to film exhibitions permitted under the terms of this licence shall be restricted in accordance with any recommendations made:

A. By the British Board of Film Classification (BBFC) where the film has been classified by that board, or

B. By the Licensing Authority where no classification certificate has been granted by the BBFC, or, where the licensing authority has notified the licence holder that section 20(3) (b) of the Licensing Act 2003 applies to the film.

The Licensee shall comply with such provisions of the Cinematograph (Safety) Regulations 1955 as restrict the use of the premises.

- 1.7 Premises must restrict the admission of children to the exhibition of any film to either the film classification recommended by the BBFC or the Licensing Authority.
- 1.8 A copy of the film has been made available to the Committee to view prior to the meeting. However, as the films total less than 12 minutes in length it is recommended that the Committee adjourn to view the films during the Licensing and Regulatory Committee.
- 1.9 In terms of guidance on how to classify a film, it is recommended that the Committee use 'The British Board of Film Classification Guidelines'. A copy of this document is attached as **Appendix 2.**
- 1.10 The classifications are listed and categorised as:

U Universal – Suitable for all

U-rated content is suitable for audiences of all ages, although not all U-rated content is aimed at children. Children's content should be set within a positive framework and should offer reassuring counterbalances to any violence, threat or scary scenes. Generally a U is unlikely to unsettle a four-year-old, although it is impossible to predict what might affect any particular child.

PG Parental Guidance – General viewing, but some scenes may be unsuitable for young children

PG-rated content is suitable for general viewing. A PG should generally not unsettle a child aged around eight, although parents and caregivers should be aware that some scenes may be unsuitable for more sensitive children.

12A/12 - Suitable for 12 years and over

No one younger than 12 may be permitted to attend a 12A cinema screening unless they are accompanied by an adult. Adults planning to take a child under 12 to a 12A should consider whether the main feature is suitable for that child. To help them decide, we recommend that they check the BBFC's content advice for the film in advance.

No one younger than 12 may rent or buy a 12-rated video.

15 – Suitable only for 15 years and over

No one younger than 15 may see 15-rated content in a cinema. No one younger than 15 may rent or buy a 15-rated video.

18 – Suitable only for adultsNo one younger than 18 may see 18-rated content in a cinema.No one younger than 18 may rent or buy an 18-rated video.

R18

R18 is a legally restricted classification for explicit pornographic works of consenting sex or strong fetish material involving adults. R18-rated cinema content may only be shown to adults in specially licensed cinemas, and R18-rated video works may be supplied to adults only in licensed sex shops and may not be supplied by mail order.

2. Alternative Options

- 2.1 The Committee could decide not to give authorisation for the films to be shown in Warwick District.
- 2.2 The Committee could decide to give this an alternative classification to that set out in the recommendations of the report.

3. Legal Implications

3.1 The Council's Statement of Licensing Policy provides that the authority will take an objective view on all applications and will ensure the promotion of the four licensing objectives. Each application will be judged on its own merits.

4 Financial Services

4.1 There are no financial implications in relation to any decision made.

5. Corporate Strategy

- 5.1 Warwick District Council has adopted a Corporate Strategy which sets three strategic aims for the organisation.
- 5.2 Delivering valued, sustainable services –Low cost, low carbon energy across the district.
- 5.3 Creating vibrant, safe and healthy communities of the future.

6 Environmental/Climate Change Implications

6.1 There are no adverse environmental impacts.

7 Analysis of the effects on Equality

7.1 Any decision made by the Committee would not impact on equality.

8 Data Protection

8.1 To ensure data protection of the films provided all Members have been asked not to distribute the films to any other person.

8.2 The Licensing Department will keep a copy of the films securely, in line with their retention policy.

9. Health and Wellbeing

9.1 There are no health and wellbeing implications in this report.

10. Risk Assessment

10.1 Arriving at the most appropriate film classification is an important judgement. Placing a film in an inappropriate classification may expose children to harm or conversely, artificially restrict a viewing audience for whom the content is acceptable.

11. Consultation

11.1 There is no requirement for a public consultation.

Background papers:

None.

Supporting documents:

Statement of Licensing Policy – Appendix 3

Revised Guidance issued under section 182 of the Licensing Act 2003 - Appendix 4

APPLICATION FOR A FILM CLASSIFICATION

I -Michael Clifford-----Grand Union Productions Ltd -----Grand Union

Hereby apply to the licensing authority to reclassify a film already classified by the BBFC or to classify a film that is not currently classified; to be shown within Warwick District.

S
Michael Clifford
additional sheets if necessary)
'School Run' and 'A Green Network Pt1'
6mins 37 And 5 mins 14
'School Run' - A new 'bike bus', a safe way for parents and children to ride to school, starts in Wellesbourne, guided by the Warwick District cycling mayor, Simon Storey. 'Green Network Part 1' - Volunteers and health professionals reveal the potential for greenways in Warwickshire and a commuter tells her story of being enabled by the Lias Line to cycle to work.

HAS YOUR FILM BEEN CLASSIFIED BY THE BBFC?:	No		
IF YES WHAT RATING WAS AWARDED?			
DATE(S) OF FILM EXHIBITION:	21.5.24		
ADDRESS(ES) OF FILM EXHIBITION:	Spa Centre, Leamington Spa		
POST CODE:	CV32 4AA		
LANGUAGE(S) SPOKEN IN FILM:	English		
CERTIFICATION SOUGHT: (i.e. "U", "PG", "12", "15", "18" etc)	U	6	
DECLARATION (Please r below)	ead carefully, tick each box to confirm you	agree and sign	
The information in this of my knowledge and b	form is accurate and correct to the best belief.	V	
	been provided to Warwick District Officers and Members in the classification)	
foreign language, I may	film is either partly or entirely in a y also incur additional costs for an er before Warwick District Council will	J	
not submitted correctly	application and all supporting items are y and within good time then Warwick guarantee it will be able to classify the exhibited.	\checkmark	
SIGNED:			
PRINT NAME(S):	MICHAEL CLIFFOR	2D	
DATED:	12.4.24		

Appendix 2





Classification Guidelines

Age Ratings. Shaped By You.



87%

of parents agree with BBFC age rating decisions all or most of the time

85%

trust BBFC age ratings to indicate whether content is suitable for themselves or their family

Trust in BBFC age ratings has grown to



among parents/caregivers in the last five years 99%

recognise at least one BBFC age rating symbol

(78% recognise them all)

97%

of people see a benefit to age ratings for some or all audiences

81%

of people want consistent age ratings across cinema, DVD and Blu-ray and Video on Demand/ streaming services

Item 5 / Page 9

Contents

2	Introduction	20	• 12A/12
5	Guiding principles	22	• 15
6	General classification considerations	24	• 18
8	Specific classification considerations	26	R18
		28	Intervention and appeals
12	Other classification considerations		
		31	Content advice and other information for audiences
14	The classification categories		
16	▲ U	34	Annexe: legal considerations
18	A PG		

The British Board of Film Classification (BBFC) is the independent film and video regulator for the UK. We are here to help everyone choose age-appropriate films, videos and websites, wherever and however they watch or use them. We do this primarily through our widely recognised and trusted age ratings and content advice.

Every BBFC age-rating decision and every piece of content advice is based on our Classification Guidelines, which we refresh every four to five years through an extensive consultation with thousands of people across the UK. This gives us an unmatched insight into what audiences find acceptable for different age groups.

The result is a transparent, well-understood and trusted classification system that gives people the best information to help them choose what to view and to protect children and vulnerable adults from harmful content.

We are not-for-profit and are funded through the fees we charge for our services.

We classify:

- films, trailers, advertisements and other content for theatrical (cinema) release on behalf of local authorities who licence cinemas under the UK's licensing legislation¹
- video works distributed on physical media under the Video Recordings Act 1984 (VRA)
- video content distributed on Video on Demand and streaming services (VoD) under a voluntary, self-regulatory service
- commercial and internet content distributed via the UK's Mobile Networks under a voluntary, self-regulatory service

As well as being based on consultation with thousands of people across the UK, our guidelines incorporate other relevant research, expert advice and our accumulated experience. The guidelines, and our practice in applying them, pay particular attention to changes in audience expectations, attitudes and concerns and to changes in the law.

Here, and throughout the guidelines, we take video content to include films, programmes and other content released on DVD and Blu-ray and on VoD services.

We take responsibility for applying the guidelines to all content submitted to us, to ensure their application and interpretation is consistent, fair and reasonable. From time to time, issues may arise that are not addressed directly in the guidelines. Where such issues occur, we will deal with them in line with the standards expressed and implied in these guidelines. The guidelines are not a legal document and should be interpreted in the spirit of what is intended as well as in the letter.

We will provide, on request, guidance on the interpretation and application of these guidelines.

Since 31 March 2016 the BBFC has sub-contracted the assessment of cinema advertisements for commercial goods and services to the Cinema Advertising Association, while retaining responsibility for classifying, on behalf of local authorities, films, trailers, public information films and charity campaigns, and other cinema content.

Shaped by you.

Our latest guidelines consultation involved 12,000 people.

Item 5 / Page 12



Guiding principles

Our guiding principles are:

- to protect children and vulnerable adults from potentially harmful content
- to empower consumers, particularly parents and those with responsibility for children, to make informed viewing decisions
- to enable content to reach the widest possible audience appropriate for the theme and treatment
- to support the right of adults to choose their own entertainment as long as it remains within the law and is not potentially harmful

We fulfil these principles chiefly by providing and publishing age ratings and content advice for films and videos. Content advice gives a detailed breakdown of the issues that result in a particular age rating, as well as other issues likely to be of relevance to audiences.

Our extensive research into UK public opinion guides us in ensuring that our classification decisions reflect public sensibilities and expectations, especially as these change over time. We also consult regularly with experts on a range of issues. Where input relating to harm is inconclusive, our own experience and expertise help us to make a judgement as to the suitability of content for classification at a particular age category, taking into consideration whether the availability of the material to the age group concerned would run contrary to broad public opinion.

In relation to harm, we will consider whether the material, either on its own or in combination with other content of a similar nature, may cause any harm at the category concerned. This includes not just any harm that may result from the behaviour of potential viewers but also any moral or societal harm that may be caused by, for example:

- desensitising a potential viewer to the effects of violence
- degrading a potential viewer's sense of empathy
- encouraging a dehumanised view of others
- encouraging antisocial attitudes
- reinforcing unhealthy fantasies
- eroding a sense of moral responsibility

Especially with regard to children, harm may also include impairing social and moral development, distorting a viewer's sense of right and wrong and limiting their capacity for compassion.

We will not classify material that is in conflict with the criminal law, and we will have regard to whether the material has arisen from the commission of an unlawful act. As the UK does not have a single legal system, we take account of the different legal systems that coexist. Further details about the applicable legislation can be found in the 'Annexe: legal considerations'.

5

General classification considerations

There are general factors that may influence a classification decision at any level and in connection with any issue. These factors are of particular importance when content lies on the borderline between two age categories.

Context

We consider the context in which an issue (such as sex, language or violence) occurs. In doing this we take account of factors such as:

- the setting (historical, fantasy, realistic, contemporary)
- the apparent intentions or manner of presentation
- the original production date (for example, outdated attitudes in some older content might be considered less offensive and consequently classified at a lower category)
- the expectations of the likely audience
- any special merits or value in the content as a whole

Theme

Classification decisions take into account the overall theme, including its treatment and the sensitivity of its presentation.

Filmmakers and other content producers are free to explore controversial themes and subjects, and we will assess them in accordance with our guidelines. Certain challenging themes (for example, suicide, substance addiction, sexual violence) are unlikely to be appropriate at U and PG. In principle, however, most themes can be presented in a manner that allows classification at 18 or, where suitable, at lower categories.

We are likely to require a less restrictive age rating where difficult themes are handled in a reassuring and age-appropriate manner, or where there is a positive outcome. We are likely to require a more restrictive age rating where difficult themes are handled in a manner likely to create or reinforce anxiety.

Tone and impact

The overall tone may also affect the classification decision. While the presentation of specific issues, such as sex and violence, may not be problematic at a particular category, a bleak or unsettling tone may result in a higher age rating. Other tonal considerations that influence classification include the extent to which the content presents a despairing view of the world, or the extent to which transgressive or harmful behaviour is condoned or normalised.

Classification decisions take into account the potential impact of the content, how it may make the audience feel. This may include the presentation of credible real-world scenarios about which audiences, especially younger audiences, are likely to be anxious (for example, terrorism, abduction, suicide, self-harm). People's expectations of genre are also important, for example regarding horror films, where threat may be more significant than the level of violence, or regarding action films, where people might expect certain types of choreographed and unrealistic violence.

The presence of multiple classification issues may result in a cumulative impact that makes a higher age rating more appropriate.



Specific classification considerations

This section identifies concerns that apply, to a greater or a lesser degree, at all classification levels, and sets out the general approach that we take. The concerns are listed in alphabetical order.

Pages 16 to 27 provide specific guidance for U through to R18 with regard to such concerns. This should be read together with the 'General classification considerations'.

Dangerous behaviour

Classification decisions will take into account any detailed portrayal of criminal and violent techniques, and glamorisation of easily accessible weapons such as knives. Content that portrays antisocial behaviour uncritically is likely to receive a higher age rating. Content that, taken as a whole, promotes illegal behaviour may be cut or found to be unsuitable for classification.

Portrayals of potentially dangerous behaviour that children and young people might copy will be cut if a higher age rating is not appropriate. The relative ease and likelihood of imitation will also be considered.

Classification decisions may be less restrictive where the risks of any dangerous behaviour are likely to be obvious to the intended audience, or where content may carry valuable messages for younger viewers about dangers. Classification decisions will be more restrictive where novel information (for example, a specific criminal technique) is presented, harmful behaviour is glamorised or risks are not made clear.

Discrimination

Classification decisions will take account of potentially offensive material relating to matters such as race, gender, religion, disability or sexuality, and of the strength or impact of their inclusion. The context in which such material may appear also has a bearing. A lower age rating may be given where:

- discriminatory language, behaviour or attitudes are implicitly or explicitly criticised or challenged
- there is a historical setting in which people would anticipate outdated attitudes or expressions
- attitudes or expressions in older content are a product of their time
- there is value to younger audiences in the representation of the past or the treatment of discriminatory behaviour
- there is no clear discriminatory intent behind the language or behaviour
- discriminatory language is used in a reclaimed or self-referential manner

A higher age rating may be given where:

- discriminatory language, behaviour or attitudes are unchallenged or normalised
- discriminatory language and behaviour are accompanied by threat or violence
- there is a clear power imbalance

Where discriminatory language or behaviour occurs, this will normally be indicated in content advice.

Drugs

Content that normalises or glamorises drug misuse is likely to receive a higher age rating than content that emphasises the dangers of drug misuse. We may cut highly detailed, promotional scenes of drug misuse if the portrayal is likely to encourage illegal behaviour or presents a harm risk at the category concerned.

Where alcohol, smoking or substance abuse feature to a significant extent in content that appeals to children, this will be indicated in content advice or may result in a higher age rating being given. Classification decisions will also take into account any promotion or glamorisation of such activities.

Language

The extent of offence that a word, expression or gesture may cause can depend upon the context in which it is used. Responses can vary according to the age, gender, race, background, beliefs or expectations of the viewer.

The advice at different categories, therefore, provides general guidance, taking account of the views expressed in our public consultations.

Nudity

Nudity with no sexual context is in principle acceptable at all categories, but at U this will be infrequent.

Sexualised nudity will receive a higher age rating. Nudity in a pornographic context – that is, where the content taken as a whole is designed primarily for the purpose of sexual arousal – will only be classified at 18 or R18.



Specific classification considerations continued

Sex

We will apply these guidelines in relation to sex to the same standard regardless of the sexual orientation of the activity portrayed.

Sexual activity can range from kissing to unsimulated sex. Progressively stronger sexual content may be permitted at each successive age rating. Overtly sexualised behaviour is a concern at U and PG.

Pornography – content that, taken as a whole, is designed primarily for the purpose of sexual arousal – will only be classified at 18 or R18.

Sexual violence

The stronger forms of sexual violence, including rape, are not permitted at U and PG, and at 12A/12 should only be implied.

When considering scenes of sexual violence, factors that may lead to a higher age rating include:

- the presence of a gang dynamic (for example, a 'gang rape' scene)
- a focus on the victim being overpowered or powerless
- prolonged, detailed or gratuitous depictions
- an emphasis on nudity
- an emphasis on the pleasure of the attacker
- a strong emphasis on the distress and fear of the victim
- a credible real-world setting
- a protracted build-up of sexual threat

Factors that may lead to a lower age rating include:

- brevity and lack of detail
- a valuable message aimed at young people
- a strong narrative justification

We may find content unsuitable for classification if it makes rape or other non-consensual sexually violent behaviour look appealing or acceptable, reinforces the suggestion that victims enjoy such behaviour, or invites viewer complicity in such behaviour.

We are likely to treat references to sexual violence less restrictively than depictions of sexual violence, although any references at the lower categories will generally be oblique or undetailed.

Suicide and self-harm

When considering portrayals of suicide or self-harm, we assess the relative ease and likelihood of imitation, and may find material unsuitable for classification if a higher age rating is not appropriate.

Classification decisions may be less restrictive where the risks are likely to be obvious to the intended audience, or where material may carry valuable messages for younger viewers about mental health, suicide or self-harm. Classification decisions will be more restrictive where novel information is presented (for example, about specific suicide techniques), where harmful behaviour is glamorised or where risks are not made clear.



Threat and horror

Where content is targeted at a younger audience, classification decisions will take into account factors such as the frequency, length and detail of scary or otherwise unsettling scenes as well as factors such as the impact of music and sound, and whether there is a swift and reassuring outcome.

The classification of threat and horror will take account of the general tone, impact, realism and supernatural elements of the content overall, as well as the level of detail in individual scenes. Fantasy settings or the inclusion of humour may lead to a lower age rating. The presentation of real-world issues and fears may lead to a higher age rating.

Violence

Classification decisions will take account of the degree and nature of violence. Content that features the following is likely to receive a more restrictive classification:

- the portrayal of violence as a normal solution to problems
- heroes who inflict pain and injury
- callousness towards victims
- the encouragement of aggressive attitudes
- characters taking pleasure in pain or humiliation

- the suffering of a victim of violence
- the glorification or glamorisation of violence
- gratuitous violence
- violence presented in a credible and realistic context (for example, gang violence, discriminatory violence, domestic abuse)

Content that features the following is likely to be treated less restrictively:

- violence in a historical context
- violence in an action or fantasy context
- violence that lacks detail
- violence that looks unreal, fake or overly staged
- comedic violence
- violence that is challenged or punished

Other classification considerations

Education videos

When classifying an education video, including a sex education video, we will take account of the educational purpose of the video and the context in which it is to be viewed (for example in the classroom mediated by a teacher).

Music videos

Where music videos are short and selfcontained, material is less likely to be justified by context, and so we take account of this when classifying them.

Photo or pattern sensitivity, motion sickness and reactions to low-frequency sound

A small number of viewers are sensitive to flashing and flickering light, or some shapes and patterns, and may experience seizures or other serious physical effects. Some viewers experience feelings of motion sickness or other symptoms when viewing content that features handheld or otherwise moving camerawork, or which feature very low-frequency sounds.

Where clear examples of such imagery or sounds are obvious upon viewing, we will advise the customer of the need to ensure that appropriate warnings are in place. Where necessary, we may require assurances regarding the display of appropriate warnings as a condition of classification.

Reclassification of older content

Our classification standards change over time in line with public opinion. Older content with an existing BBFC age rating may therefore require a higher or lower age rating if it is submitted to us again.

To ensure the age rating is in line with current audience expectations, we strongly encourage customers to submit such content to us again in order for us to reassess it under current standards. Customers will be required to resubmit the content if the existing rating is a historic rating and therefore no longer valid.

In some cases, previously issued cuts are no longer required, because of changes in our classification standards or amendments to relevant laws. Changes to UK law and its interpretation may also result in cuts being required to content previously passed uncut.

Release format

Classification decisions may be stricter on video works than for cinema content. This is because of the increased possibility of underage viewing as recognised in the VRA (see 'Annexe: legal considerations'), as well as the increased possibility of works being replayed or sections viewed out of context. Accordingly, a video (either packaged or online) may occasionally receive a higher age rating than for cinemas, or require new or different cuts. The screen format or visual presentation of a submission may also alter a classification, for example, if the image has been processed in the 3D format, or is shown with an altered aspect ratio such as on an IMAX screen, or if the content is experienced as immersive linear virtual reality (VR). A film may also receive a different rating than before because it is a different edit.

Titles

We will require changes as a condition of classification if the title of a film or video incites hatred or other criminal behaviour or encourages an interest in abusive or illegal sexual activity.

If the title is likely to cause significant offence to a significant number of people if it were displayed in a public place, we will advise the customer to consider carefully the places in which the title is likely to be seen and to take appropriate action, for example, by obscuring certain words on packaging or marketing materials. (This advice is not given in relation to pornographic video works classified R18 as such works may only be supplied or offered for supply in a licensed sex shop.)

Trailers and advertisements

Audiences may choose to see a full-length feature based on expectations of the particular genre at the given classification and on the published content advice. In contrast, audiences have no choice, and often no expectation, about the accompanying trailers or advertisements, which may be very different in tone and content to the main feature. In addition, because trailers and advertisements are short and self-contained, material is less likely to be justified by context.

For these reasons, classification decisions for trailers and advertisements may be higher than for equivalent material in a main feature.

Video games

With a few limited exceptions, we do not classify video games. We consider for classification those video games contained on discs that feature primarily linear video content and any pornographic video games.

We also advise the Games Rating Authority on the classification of linear video footage contained in games that is not integral to the game. This includes, for example, rewards and video content in games that is designed to be viewed in its own right, without taking forward the narrative drive of the game.

Virtual reality

The BBFC is responsible for classifying linear VR content whereas the Games Rating Authority is responsible for classifying non-linear VR content.

The classification categories




The following guidance should be read in combination with the general approach set out earlier under 'Guiding principles', 'General classification considerations', 'Specific classification considerations' and 'Other classification considerations'.

Because content from time to time presents issues in ways that cannot be anticipated, we will not apply these criteria in an over-literal way if such an interpretation would lead to an outcome that would confound audience expectations.

Universal Suitable for all



U-rated content is suitable for audiences of all ages, although not all U-rated content is aimed at children.

Children's content should be set within a positive framework and should offer reassuring counterbalances to any violence, threat or scary scenes. Generally a U is unlikely to unsettle a four-year-old, although it is impossible to predict what might affect any particular child.

Dangerous behaviour

Potentially dangerous or antisocial behaviour that young children may copy should be clearly disapproved of or wholly unrealistic. There should be no emphasis on realistic or easily accessible weapons.

Discrimination

Discriminatory language or behaviour is unlikely to be acceptable unless clearly disapproved of.

Drugs

References to drugs must be infrequent and innocuous, or have valuable messaging for young children.

Language

There may be very mild bad language. Frequent use may result in a higher age rating.

Nudity

There may be occasional nudity with no sexual context.

Sex

There may be very mild sex references or sexual behaviour (for example, kissing).

Threat and scary scenes

Threat and scary scenes that may unsettle young children should be very mild and the outcome should be reassuring.

Violence

Violence should be very mild and justified by context (for example, comedic, animated, wholly unrealistic). Injury detail should be very mild, brief and undetailed.





Parental Guidance

General viewing, but some scenes may be unsuitable for young children



PG-rated content is suitable for general viewing. A PG should generally not unsettle a child aged around eight, although parents and caregivers should be aware that some scenes may be unsuitable for more sensitive children.

Dangerous behaviour

Potentially dangerous behaviour that young children are likely to copy should not be detailed or presented as safe or fun. Realistic or easily accessible weapons such as knives should not be glamorised. Any focus on antisocial behaviour that young children are likely to copy is unlikely to be acceptable.

Discrimination

Discriminatory language or behaviour should be clearly disapproved of. Some language or behaviour may be permitted if justified by context (for example, reclaimed, historical, lack of discriminatory intent).

Drugs

References to illegal drugs or drug misuse should be brief and carry a suitable cautionary message.

Language

There may be mild bad language only.

Nudity

There may be nudity with no sexual context.

Sex

Sexual activity may be implied, but should be discreet and infrequent. There may be mild sex references and innuendo only.

Threat and scary scenes

Threat and scary scenes should not be prolonged or intense.

Violence

Violence will be mild, with more impactful moments permitted if justified by context. Injury detail, including sight of blood, should be brief.





12A/12 Suitable for 12 years and over



Cinema content classified 12A and video content classified 12 contain material that is not generally suitable for children aged under 12.

No one younger than 12 may be permitted to attend a 12A cinema screening unless they are accompanied by an adult. Adults planning to take a child under 12 to a 12A should consider whether the main feature is suitable for that child. To help them decide, we recommend that they check the BBFC's content advice for the film in advance.

No one younger than 12 may rent or buy a 12-rated video.

Dangerous behaviour

Potentially dangerous behaviour that children are likely to copy should not be encouraged. Realistic or easily accessible weapons such as knives should not be glamorised. Antisocial behaviour should not be endorsed.

Discrimination

Discriminatory language or behaviour must not be endorsed by the content as a whole. Such behaviour is unlikely to be acceptable if accompanied by violence, unless clearly condemned.

Drugs

Misuse of drugs or other substances will generally be infrequent. There should be no glamorisation or instructional detail.

Language

There may be moderate bad language. Strong language may be permitted, depending on context and frequency within the content as a whole.

Nudity

There may be nudity, but in a sexual context it should be brief and discreet.

Sex

Sexual activity may be portrayed briefly and discreetly. Moderate sex references are permitted, but frequent crude references are unlikely to be acceptable.

Sexual violence

There may be verbal references to sexual violence provided they are not graphic. The stronger forms of sexual violence, including rape, may only be implied and any sexual threat or abusive behaviour should be brief and negatively presented.

Suicide and self-harm

There may be references to suicide and self-harm, although stronger details are unlikely to be permitted unless justified by context. Portrayals of suicide and self-harm should only be brief and discreet.

Threat and horror

There may be scenes of moderate physical and psychological threat. Although some scenes may be disturbing, the overall tone should not be. Horror sequences should not be frequent or sustained.

Violence

There may be moderate violence, but it should not dwell on detail. Occasional blood and injuries may be shown, but gory images should be infrequent and justified by context (for example, medical detail, fantasy).







No one younger than 15 may see 15-rated content in a cinema. No one younger than 15 may rent or buy a 15-rated video.

Dangerous behaviour

Highly dangerous behaviour should not dwell on detail that could be copied. Whether the depiction of easily accessible weapons is acceptable will depend on factors such as realism, context and setting.

Discrimination

Discriminatory language, themes and behaviour are permitted, but must not be endorsed by the content as a whole.

Drugs

Drug and substance misuse may be shown, but should not be promoted or encouraged by the content as a whole.

Language

There may be strong language. Very strong language may be permitted depending on the context and frequency.

Nudity

Sexualised nudity may be permitted, but graphic detail should be brief and justified by context.

Sex

Sexual activity may be portrayed, but usually without graphic detail. There may be strong sex references, but frequent crude references are unlikely to be acceptable. Content that, when taken as a whole, has the primary purpose of sexual arousal is unacceptable.

Sexual violence

There may be detailed verbal references to sexual violence or abusive behaviour, but graphic descriptions of specific acts are unlikely to be acceptable. The stronger forms of sexual violence, including rape, must not be detailed or prolonged. An intense and sustained focus on sexual threat is unacceptable.

Suicide and self-harm

Scenes of suicide and self-harm should not be frequent or endorsed. Glamorisation of suicide or self-harm may result in a higher rating. Graphic depictions or detailed references are unlikely to be permitted.

Threat and horror

There may be strong threat and horror. A sustained focus on sadistic threat is unlikely to be acceptable.

Violence

Violence may be strong but there should not be sustained focus on the infliction of pain or victims' suffering. The strongest gory images may occur if justified by context (for example, war, medical detail). Prolonged sadistic violence is unlikely to be acceptable.





No one younger than 18 may see 18-rated content in a cinema. No one younger than 18 may rent or buy an 18-rated video.

Adults should be free to choose their own entertainment. Exceptions are most likely in the following areas:

- where material is in breach of the criminal law, or has been created through the commission of a criminal offence
- where material risks harm to individuals or, through their behaviour, to society. This may include portrayals of sadistic violence, rape or other non-consensual sexually violent behaviour, particularly where this is made to look appealing, reinforces the suggestion that victims enjoy it, or invites viewer complicity. Other examples may include scenes of drug misuse or suicide/self-harm where the portrayal is sufficiently detailed, glamorised or promotional as to represent a non-trivial harm risk
- where there are more explicit images of sexual activity in the context of pornography (see right)

In the case of videos, which may be more accessible to younger viewers, intervention may be more frequent than for cinema films.

Pornography at 18

Pornography is content that, taken as a whole, is designed primarily for the purpose of sexual arousal. Pornography containing only material that may be simulated is generally passed 18. Pornography containing clear images of real sex, strong fetish material, sexually explicit animated images, or other very strong sexual images will be confined to the R18 category.

Material that is unacceptable in pornography at R18 is also unacceptable in pornography at 18.



Item



To be shown only in specially licensed cinemas, or supplied only in licensed sex shops, and to adults only



R18 is a legally restricted classification for explicit pornographic works of consenting sex or strong fetish material involving adults. **R18**-rated cinema content may only be shown to adults in specially licensed cinemas, and **R18**-rated video works may be supplied to adults only in licensed sex shops and may not be supplied by mail order.

The following content is not acceptable:

- material that is in breach of the criminal law, including material judged to be obscene under the current interpretation of the Obscene Publications Act 1959 (see 'Annexe: legal considerations')
- material (including dialogue) likely to encourage an interest in sexually abusive activity that may include adults role-playing as non-adults
- the portrayal of sexual activity that involves real or apparent lack of consent. Any form of physical restraint that prevents participants from indicating a withdrawal of consent
- the infliction of pain or acts that are likely to cause serious physical harm, whether real or (in a sexual context) simulated.
 Some allowance may be made for nonabusive, consensual activity
- penetration by any object likely to cause physical harm
- sexual threats, humiliation or abuse that do not form part of a clearly consenting role-playing game

Item 5 / Page 36

Intervention and appeals

Where possible we will carry out our responsibilities through appropriate use of the classification categories, our primary focus always being the protection of children and vulnerable adults from potential harm. In some cases, we require assurances, cuts or other changes (for example, the addition of warning captions) as a condition of classification, or as a condition of classifying at a particular category. If necessary, however, we may cut a film or video or determine that it is unsuitable for classification.

Advice viewings

A customer may submit content for advice at any stage of the production process. We will inform them of the likely age rating their content will receive, and where appropriate any changes required to achieve their preferred age rating.

Advice given in such circumstances is not binding and we reserve the right to reach a different decision when the final version is submitted for classification. If the final version differs in any significant respect from the advice version, and if those changes appear to reflect advice we have given, then details of the changes will appear on our website.

Optional cuts

As well as providing our advice service, we work with customers to help them achieve their preferred age rating when they submit their content for classification. Securing a lower age rating may be possible through relatively simple changes. If so, we will offer the customer a choice of accepting either the higher age rating without cuts or the lower age rating with changes as necessary.

Optional cuts are unlikely to be available if the required changes would be very extensive or complex, or would not address a tonal or thematic issue running throughout.

Compulsory cuts

If a film or video raises issues or concerns that cannot be addressed by classification at a particular category, we may require cuts or other changes as a condition of classification.

Such intervention is most likely when the content contains:

- material that may promote criminal activity
- material that is obscene or otherwise illegal
- material created by means of the commission of a criminal offence
- portrayals of children in a sexualised or abusive context
- material that makes rape, other nonconsensual sexually violent behaviour or sadistic violence look appealing
- graphic images of real injury, violence or death presented in a salacious or sensationalist manner that risks harm by encouraging callous or sadistic attitudes
- material that reinforces the suggestion that victims enjoy rape or other non-consensual sexually violent behaviour
- material that invites viewer complicity in rape, other non-consensual sexually violent behaviour or other harmful violent activities
- pornography that contains material listed as unacceptable at R18

When the issue relates to the circumstances of filming (for example, in relation to animal cruelty or public indecency), the customer will be given an opportunity to present evidence before a final decision is reached.

Unsuitable for classification

As a last resort, we may find the content as a whole unsuitable for classification, in line with the objective of preventing non-trivial harm risks to potential viewers and, through their behaviour, to society.

We may do so, for example, where a central concept is unacceptable, such as a sustained focus on rape, other non-consensual sexually violent behaviour or sadistic violence. Before determining that a film or video is unsuitable for classification, we will consider whether the problems could be adequately addressed through intervention such as cuts. In considering whether content is unsuitable for classification, we will keep in mind the inherent difficulty of using behavioural research to draw conclusions about real-world risks, and will have regard to the full range of available evidence, including the views of audiences and our own knowledge and experience.

Appeals

We offer a formal reconsideration process, open to any customer wishing to appeal our classification decision. The reconsideration is free of charge and will normally take fewer than 10 working days. A customer may also appeal directly to an independent authority. Such an appeal may take place following, or instead of, our reconsideration. In the case of cinema films, the customer (or any member of the public) may contact the local authority that licenses cinemas in a particular area. In the case of video works that we have classified under the VRA, a customer may appeal to the Video Appeals Committee (VAC).

The VAC is independent of the BBFC and can be contacted as follows:

The Secretary The Video Appeals Committee 3 Soho Square London W1D 3HD Email: secretary@vacappeal.co.uk

Customers should note that a reconsideration or an appeal involves looking at the issues afresh. This means that the outcome could, in some circumstances, be more restrictive than the original decision.





Content advice and other information for audiences

We publish information about the content of every film and series that we classify. Aimed particularly at parents and caregivers, content advice includes:

- a short description of the issues contained in the film or the episodic content. For cinema releases, this appears on the black card shown on screen before the film starts
- an extended version, designed for people who want a more detailed idea of the issues – such as bad language, drugs, sex, violence or the use of discriminatory language – that they are likely to find within the content, before they or their children watch it

We publish content advice on our website and app the moment a film or video is classified. Our extended content advice for cinema films may only be available from 10 days before the film opens. We encourage people to check our content advice before they decide what is right to view for themselves or their family. All our extended content advice is published with a spoiler warning.

bbfc.co.uk

Our website provides a comprehensive database of all the content we classify, including age ratings, content advice and information about cuts. We publish the Classification Guidelines, as well as our press releases, research and Annual Reports. We also publish case studies – including on many films that are set texts for Film Studies courses in the UK – a timeline of key events in our history, and podcasts.

cbbfc.co.uk

CBBFC contains content for younger children to help them learn more about our work. It includes activities such as the chance to apply age ratings to trailers for children's films. Parents and caregivers can also find advice about choosing content for children on VoD platforms and information about our education programme.







Our ratings quick guide



A U film should be suitable for audiences aged four years and over, although it is impossible to predict what might upset any particular child.





BBFC app

Our free app for iOS and Android devices lets audiences check the latest age-rating decisions and content advice, watch trailers, find local showtimes and read our guidelines.

Social media

We operate official BBFC social media accounts on a number of different platforms, including Facebook, Instagram and X. We share all our news and latest classification decisions, along with topical and relevant resources including articles and case studies. You can ask us quick questions there too. For more detailed questions, you can email us on feedback@bbfc.co.uk.

Podcasts

Our BBFC podcast is available to stream from our website, as well as from Apple Podcasts, Soundcloud and Spotify. Each episode focuses on a particular theme or on age-rating decisions, highlighting key issues we took into account and any interesting facts about the classification process, for example any advice we gave, or changes made by the filmmaker or customer to secure a particular age rating.

Feedback

Your feedback is welcome. Please contact us on feedback@bbfc.co.uk or write to us at:

Chief Executive's Office British Board of Film Classification 3 Soho Square London W1D 3HD





Annexe: legal considerations

The following legislation is not listed according to chronology or importance. Instead, it reflects a useful way of explaining the structure of the legal framework that applies to our work.

The Licensing Act 2003 England and Wales The Cinemas Act 1985 Scotland The Cinemas (Northern Ireland) Order 1991 Northern Ireland

Cinemas require a licence from the local authority in which they operate. The licence must include a condition requiring the admission of children (anyone under 18) to any film to be restricted in accordance with our recommendations or those of the licensing authority. One of the key reasons for the licensing requirement is the protection of children, including from potentially harmful content in films.

The Video Recordings Act 1984

Video works (including films, TV programmes and some video games) that are supplied on a disc, tape or any other device capable of storing data electronically must have a BBFC classification unless they fall within the definition of an exempted work.

When considering whether to give an age rating to a work, or whether a work is suitable at a particular category, we are required by the Act to have special regard to the likelihood of works being viewed in the home, and to any harm that may be caused to potential viewers or, through their behaviour, to society by the manner in which the work deals with:

- criminal behaviour
- illegal drugs
- violent behaviour or incidents

- horrific behaviour or incidents
- human sexual activity

In considering these issues, we have in mind the possible effect not only on children but also on other vulnerable people.

The Obscene Publications Acts 1959 and 1964 England and Wales The Civic Government (Scotland) Act 1982 Scotland The Obscene Publications Act 1857

The Obscene Publications Act 1857 Northern Ireland

It is illegal to publish a work that is obscene. A work is obscene if, taken as a whole, it has a tendency to deprave and corrupt a significant proportion of those likely to see it. Under the Obscene Publications Act 1959, no offence is committed if publication is justified as being for the public good on the grounds that it is in the interests of science, art, literature or learning or other objects of general concern.

In Scotland, case law implies a similar test would be applied. In Northern Ireland, while there is no express defence of public good it is likely that English law would be taken into consideration. The Criminal Justice and Immigration Act 2008 England, Wales and Northern Ireland The Criminal Justice and Licensing (Scotland) Act 2010/The Civic Government (Scotland) Act 1982 Scotland

It is illegal to be in possession of an extreme pornographic image. Under the Criminal Justice and Immigration Act 2008 an extreme pornographic image is one that is pornographic and grossly offensive, disgusting or otherwise of an obscene character, which features an apparently real person and which portrays, in an explicit and realistic way, an act that:

- threatens a person's life
- results, or is likely to result, in serious injury to a person's anus, breasts or genitals
- involves sexual interference with a human corpse
- involves intercourse or oral sex with an animal
- involves non-consensual penetration of a vagina, anus or mouth with a penis or non-consensual sexual penetration of a vagina or anus by another part of a person's body or anything else

Under the Civic Government (Scotland) Act 1982 an extreme pornographic image is one that is pornographic and obscene, and which depicts, in an explicit and realistic way, an act that:

- takes or threatens a person's life
- results, or is likely to result, in a person's severe injury

- involves rape or other non-consensual penetrative sexual activity
- involves sexual activity involving (directly or indirectly) a human corpse
- involves sexual activity between a person and an animal

Works we have classified under the VRA are excluded from the scope of the offence unless images have been extracted from such works for the purpose of sexual arousal.

The Protection of Children Act 1978 England and Wales The Civic Government (Scotland) Act 1982 Scotland The Protection of Children (Northern Ireland) Order 1978

Northern Ireland

It is illegal to make, distribute, show or possess indecent photographs or pseudo-photographs of a child. It is also illegal to make, distribute, show or possess indecent images of children that have been derived from a photograph or pseudo-photograph (for example, by tracing). Offences relating to the possession of such images are contained within the Criminal Justice Act 1988 (England, Wales and Scotland), and the Criminal Justice (Evidence, Etc.) (Northern Ireland) Order 1988. A child is defined as a person under the age of 18.

Annexe: legal considerations continued

The Coroners and Justice Act 2009 England, Wales and Northern Ireland

It is illegal to be in possession of a prohibited image of a child. A prohibited image of a child is a non-photographic or non-pseudo-photographic image that is pornographic and grossly offensive, disgusting or otherwise of an obscene character, and which focuses solely or principally on a child's genitals or anal region, or which portrays specified sexual acts by, of, or in the presence of a child, including masturbation, oral sex or penetration, including sexual acts with animals. A child is defined as being under 18 and an image of a child or other person can include imaginary representations. Works we have classified under the VRA are excluded from the scope of the offence unless images have been extracted from such works for the purpose of sexual arousal.

The Sexual Offences Act 2003 England and Wales The Sexual Offences (Scotland) Act 2009 Scotland The Sexual Offences (Northern Ireland) Order 2008 Northern Ireland

It is illegal to expose oneself with intent to cause alarm or distress – this offence augments the common law misdemeanour of indecent exposure. It is also prohibited for a person to record the private act of another, where the intention of the recording is for the sexual gratification of himself or a third party and where the recorded party has not consented to so being filmed. The Criminal Justice and Courts Act 2015 England and Wales The Abusive Behaviour and Sexual Harm (Scotland) Act 2016 Scotland The Justice Act (Northern Ireland) 2016 Northern Ireland

It is an offence to disclose a private sexual photograph or film without the consent of any individual who appears in the photograph or film, if it is done with the intention of causing that individual distress (or, in Scottish law, causing them fear, alarm or distress).

The Public Order Act 1986

England, Wales and Scotland
The Public Order (Northern Ireland) Order 1987
Northern Ireland

It is illegal to distribute, show or play to the public a recording of visual images or sounds that are threatening, abusive or insulting if the intention is to stir up racial hatred, or if racial hatred is likely to be stirred up. It is also illegal to distribute, show or play to the public a recording of visual images or sounds that are threatening if the intention is to stir up religious hatred or hatred on the grounds of sexual orientation.

In Northern Ireland the relevant group of persons may be defined not only by colour, race, nationality or ethnic or national origins but also by religious belief, sexual orientation or disability.

The Cinematograph Films (Animals) Act 1937 England, Wales and Scotland

It is illegal to show any scene organised or directed for the purposes of the film that involves actual cruelty to animals. This Act applies to the exhibition of films in public cinemas but we also apply the same test to video works. For the purposes of this legislation and the Animal Welfare Act 2006, only vertebrates that are domesticated or otherwise under the control of man are defined as animals.

The Animal Welfare Act 2006 England and Wales The Animal Health and Welfare (Scotland) Act 2006 Scotland The Welfare of Animals Act (Northern Ireland) 2011 Northern Ireland

It is illegal to supply, publish, show or possess with intent to supply a video recording of an animal fight that has taken place within the UK from the commencement date of the relevant legislation.

The Tobacco Advertising and Promotion Act 2002

It is illegal, in the course of a business, to publish a tobacco advertisement.

Blasphemy

In Scotland and Northern Ireland, the common law crime of blasphemy exists but has not been utilised for prosecution in modern times. The offences of blasphemy and blasphemous libel under the common law of England and Wales were abolished in the Criminal Justice and Immigration Act 2008.

The Human Rights Act 1998

The Act permits such restrictions on freedom of expression as are prescribed by law and are necessary in a democratic society, in the interests of national security, territorial integrity or public safety, for the prevention of disorder or crime, for the protection of health or morals, for the protection of the reputation or rights of others, for preventing the disclosure of information received in confidence, or for maintaining the authority and impartiality of the judiciary.

Other unlawful material

In carrying out its responsibilities the BBFC will have regard to whether the material itself appears to be unlawful in the UK, or has arisen from the commission of an unlawful act.



Get more info on our app **bbfc.co.uk/app**

British Board of Film Classification 3 Soho Square London W1D 3HD

T 020 7440 1570

bbfc.co.uk

Copyright © British Board of Film Classification 2024



Appendix 3



STATEMENT OF LICENSING POLICY

(Required by section 5 of the Licensing Act 2003)

2021 - 2026

IMPORTANT NOTE

In producing this Statement Of Licensing Policy the Licensing Authority is aware that the Government may amend the Licensing Act 2003, subordinate legislation and statutory guidance.

Any such amendments made in the future may not be incorporated into this policy document and readers of this document are advised to check the Home Office/Gov.uk website to ensure they have the latest information.

CONTENTS	Page
Introduction	3
Consultation	4
Fundamental Rights	5
Licensing Conditions	5
Operating Hours	6
Late Night Levy and Early Morning Restriction Order	6
Cumulative Impact	7
Promotion of the Licensing Objectives	8
Mandatory Licensing Conditions	12
Other Considerations	12
Best Practice Schemes	14
Integrating Strategies and Avoidance of Duplication	14
Enforcement	16
Administration, Exercise and Delegation of Functions	17
Comments on this policy	19

Appendix

- 1. Model Conditions
- 2. Responsible authorities List

STATEMENT OF LICENSING POLICY

INTRODUCTION

- 1.1 Warwick District Council ('the Council') has a duty under the terms of the Licensing Act 2003 ('the Act') to carry out its functions as the Licensing Authority with a view to promoting the following licensing objectives:
 - The prevention of crime and disorder
 - Public safety
 - The prevention of public nuisance
 - The protection of children from harm
- 1.2 The promotion of these objectives is the paramount consideration when determining an application and any conditions attached to an authorisation.
- 1.3 Warwick District Council (WDC) is situated in the south of Warwickshire in the centre of England. Appropriately for England's heartland, Warwick District Council's boundaries are roughly heart-shaped, embracing an area of some 28,253 hectares with a population exceeding 138,400 people. The District covers four towns, Royal Learnington Spa, Warwick, Kenilworth and Whitnash as well as a large rural area with 18 Parish Councils. It is acknowledged that the town centres have a large proportion of residential premises.
- 1.4 This statement of licensing policy relates to all those licensing activities identified as falling within the provisions of the act, namely:
 - The sale by retail of alcohol
 - The supply of alcohol by clubs
 - The provision of regulated entertainment
 - The provision of late night refreshment

For the purposes of this document any reference to an 'authorisation' means a Premises Licence, Club Premises Certificate, Temporary Event Notice (TEN) and where appropriate to the context a Personal Licence.

1.5 The Licensing Authority recognises that the licensing function is only one means of promoting delivery of the above objectives and should not therefore be seen as a means for solving all problems within the community. The Licensing Authority will therefore continue to work with the Responsible Authorities, the South Warwickshire Community Safety Partnership, local businesses and local people to promote the common objectives as outlined. In addition the Licensing Authority recognises its duty under s.17 of the Crime and Disorder Act 1998 with regard to the prevention of crime and disorder.

- 1.6 This policy statement has been prepared in accordance with the provisions of the Act and the guidance issued under s.182 of the Act. The policy statement is valid until 5th January 2026. This policy statement will be subject to review and further consultation prior to any substantial changes.
- 1.7 A list of contact details for the Responsible Authorities authorised under the act is attached to this policy statement as Appendix 2.
- 1.8 The Licensing Authority has recognised Warwickshire County Council as the local body competent to advise it on the protection of children from harm and has designated it as a responsible authority for the purposes of s.13 of the Act.
- 1.9 The Licensing Authority will, when acting as a responsible authority, act in accordance with the guidance issued under s.182 of the Act wherever possible. In particular, it will not normally intervene in applications where the issues are within the remit of another responsible authority and will ensure an appropriate separation of responsibilities between the officer administering an application and an officer acting as a responsible authority.

Public Health as a Responsible Authority

- 1.10 There is not a specific licensing objective related directly to health within the current legislation. When making a representation, the Director of Public Health will be required to relate such representations and available data to the other licensing objectives. This may include underage drinking, prevention of accidents, injuries and other immediate harms that can result from alcohol consumption.
- 1.11 Health bodies hold valuable information which may not be recorded by other agencies, including analysis of data on attendance at emergency departments and the use of ambulance services following alcohol related incidents. Sometimes it may be possible to link ambulance callouts and attendance to irresponsible practices at specific premises. Anonymised data can be collated about incidents relating to specific premises and presented to Licensing Sub-Committees when representations are made.

CONSULTATION

- 2.1 Before publishing this policy statement the Licensing Authority has consulted and given proper consideration to the views of the following in line with the statutory guidance.
 - The Responsible Authorities
 - Representatives of current licence and certificate holders
 - Representatives of local businesses
 - Representatives of local residents

FUNDAMENTAL RIGHTS

- 3.1 Under the terms of the Act any person may apply for a variety of authorisations and have each application considered on its individual merits. Equally, any person has a right to make relevant representations on an application or to seek a review of a licence or certificate where provision has been made for them to do so in the Act.
- 3.2 Applicants and those making relevant representations in respect of applications to the Licensing Authority have a right of appeal to Warwickshire Magistrates' Court against the decisions of the Licensing Authority.

LICENSING CONDITIONS

- 4.1 Licensing is about regulating the carrying on of licensable activities on licensed premises, by qualifying clubs and at temporary events within the terms of the Act. 'Premises' includes open spaces. Conditions attached to various authorisations will be focused on matters that are within the control of the individual licensees and others in possession of relevant authorisations. Accordingly, these matters will centre on the premises being used for licensable activities and the vicinity of those premises. If there is an incident or other dispute, the Licensing Authority will primarily focus on the direct impact of the activities taking place at licensed premises on members of the public living, working or engaged in other normal activities in the area concerned.
- 4.2 The Licensing Authority can impose conditions if it has received a relevant representation or if such conditions are consistent with the operating schedule.
- 4.3 When considering any conditions, the Licensing Authority acknowledges that licensing law should not be seen as the primary mechanism for the general control of nuisance and anti-social behaviour by individuals once they are away from licensed premises and therefore beyond the direct control of the individual club or business holding the licence, certificate or authorisation concerned. Nonetheless, it is a key aspect of general control and licensing law will always be part of a holistic approach to the management of the evening and night time economy. For example, applicants should note that stricter conditions to control noise are likely to be imposed in the case of premises situated in largely residential areas.
- 4.4 The Licensing Authority will not impose standard licensing conditions on licences or other relevant types of authorisation across the board. Therefore, the Licensing Authority will attach conditions to relevant authorisations which are tailored to the individual style and characteristics of the premises and events concerned and that are appropriate to promote the licensing objectives in the light of any representations received.

4.5 The Licensing Authority has produced a set of model conditions, which is aimed at assisting and supporting applicants through the application process. The model conditions would also assist the Licensing Authority and Responsible Authorities in deciding which conditions would be appropriate to add to a licence. The model conditions may be found at the end of this policy as Appendix 1.

OPERATING HOURS

- 5.1 The Licensing Authority welcomes the opportunities afforded to the local economy by the Act and will strive to balance this with the rights of local residents and others who might be adversely affected by licensable activities based on the principles laid down in this document.
- 5.2 When dealing with licensing hours, each application will be dealt with on its individual merits. The Licensing Authority recognises that longer licensing hours with regard to the sale of alcohol can assist to avoid concentrations of customers leaving premises simultaneously thereby reducing the friction at late night fast food outlets, taxi ranks and other forms of transport which can lead to disorder and disturbance. The Licensing Authority does not wish to unduly inhibit the development of thriving and safe evening and night time local economies which are important for investment and employment locally and in the main welcomed by residents and visitors to the District.
- 5.3 The Licensing Authority will not set fixed trading hours within designated areas. However, an earlier terminal hour and stricter conditions with regard to noise control than those contained within an application, may be appropriate in residential areas where relevant representations are received and such measures are deemed appropriate to uphold the licensing objectives.
- 5.4 Shops, stores and supermarkets will generally be free to provide sales of alcohol for consumption off the premises at any times when the retail outlet is open for shopping, unless there are relevant representations giving good reasons based on the licensing objectives for restricting those hours.

LATE NIGHT LEVY AND EARLY MORNING RESTRICTION ORDERS

6.1 The Licensing Authority, having not been presented with any evidence to the contrary, does not consider that the application of a Late Night Levy or Early Morning Restriction Order are appropriate for the Licensing Authority's area at the present time. The Licensing Authority will keep these matters under review and accordingly reserves the right, should the need arise, to introduce these measures during the life of this statement of licensing policy.

CUMULATIVE IMPACT ASSESSMENT

- 7.1 For the purposes of this document 'cumulative impact' means the potential impact on the promotion of the licensing objectives of a significant number of licensed premises concentrated in one area. Cumulative impact is a proper matter for the Council to consider in developing its licensing policy. This should not be confused with 'need', which concerns the commercial demand for another particular type of premises. Government guidance states that 'need' is not a matter for the Licensing Authority but is a matter for the Planning Authority and the free market.
- 7.2 The Licensing Authority has not been presented with sufficient evidence to consider any area within the District to currently have a particular concentration of licensed premises causing a cumulative impact on one or more of the licensing objectives. The Licensing Authority will keep this matter under review and accordingly reserves the right, should the need arise, to introduce a special policy concerning cumulative impact during the life of this statement of licensing policy.
- 7.3 The absence of a special policy does not prevent any responsible authority or any other party from making representations on a new application for the grant of a licence on the grounds that the premises will give rise to a negative cumulative impact on one or more of the licensing objectives. The Act allows for such consideration but the individual merits of each application must always be considered.

PROMOTION OF THE LICENSING OBJECTIVES

Prevention of Crime and Disorder

- 8.1 Licensed premises, especially those offering late night/early morning entertainment or alcohol and refreshment can sometimes be associated with elevated levels of crime and disorder.
- 8.2 The Licensing Authority expects individual licence/certificate holders, new applicants and temporary event organisers, to regularly review their arrangements in addressing crime and disorder issues pertinent to their particular licensable activities, location and/or premises. Information and advice can be obtained from the Police, business network groups and other sources. The Licensing Authority also encourages local residents and other businesses, where appropriate, to discuss issues of concern directly with individual businesses, or, to contact the Police or the Licensing Authority.
- 8.3 The Licensing Authority will, through its Community Safety Partnership devise and help deliver strategies to tackle the misuse of alcohol, which has been identified as being a precursor to crime and anti-social behaviour. The Licensing Authority expects existing licence/certificate holders, new applicants and the organisers of temporary events, to be able to demonstrate the measures they

use, or propose to adopt, to prevent and actively discourage the sale/supply of alcohol to children and to individuals who are already drunk.

- 8.4 The risk assessment approach remains fundamental in the operation of all licensed premises. Licence/certificate holders and applicants are strongly recommended to work closely with the Police in particular, in bringing into effect appropriate control measures to overcome established and potential problems. A combination of short and longer term strategies may need to be deployed by holders of authorisations to sustain and promote the prevention of crime and disorder.
- 8.5 The Licensing Authority will expect new applicants, existing licence/certificate holders and organisers of temporary events to adopt recognised good practice in whatever area of operations they are engaged. The Licensing Authority will regard each responsible authority as the expert in their respective field and in some cases as the primary source of advice in relation to a particular licensing objective.
- 8.6 Queues at late night take-aways can be a source of disorder and applicants for premises licences for this type of premises are expected to address this in their operating schedule.
- 8.7 The Licensing Authority has specific duties under s.17 of the Crime and Disorder Act 1998 that underpins any control strategy that is employed. The Licensing Authority will continue to work in partnership with the Police in addressing crime and disorder issues.
- 8.8 The Licensing Authority is of the view that generally, in order to promote the licensing objectives, all licensed premises within the District are encouraged to be members of the relevant local Pubwatch Scheme, or any similar scheme, where one exists.
- 8.9 The Licensing Authority and Police have a zero tolerance of drug use in licensed premises but recognise that drug use is not something that is relevant to all licensed premises. However, it is recognised that special conditions may need to be imposed for certain venues to reduce the likelihood of drugs being sold and consumed and to create a safer environment for those who may have taken them.
- 8.10 Once away from licensed premises a minority of consumers may behave badly and unlawfully. There are other mechanisms both within and outside the licensing regime that are available for addressing such issues. The Licensing Authority will address a number of these issues through the Community Safety Partnership in line with the strategic objectives for crime and disorder reduction and drug and alcohol misuse within the District.
- 8.11 In relation to premises seeking or holding a premises licence and where alcohol will be sold under the terms of that licence, the Licensing Authority expects that

(a) any Designated Premises Supervisor (DPS) will have been given sufficient management authority and be able to exercise effective day-to-day control of the premises and (b) authority to make alcohol sales when given by the DPS or any other Personal Licence holder should be clearly evidenced in writing. This is to ensure that premises selling alcohol are properly managed in accordance with the Act and that premises operate in a way that promotes the prevention of crime and disorder. This will also benefit operators themselves through being able to demonstrate a commitment to the proper management of premises, particularly if enforcement action becomes necessary.

Promotion of Public Safety

- 8.12 Public safety is not defined within the act, but the Government guidance advises that it is concerned with the physical safety of people using the premises and not with public health, which is covered by other legislation.
- 8.13 Applicants and event organisers will be expected to assess not only the physical environment of the premises or site but also operational practices, in order to protect the safety of members of the public visiting the premises or site, those who are employed in the business, those who are engaged in running an event or anyone else that could be affected by site activities. This assessment would normally take place within a risk assessment framework.
- 8.14 Holders of premises licences and club certificates, or those organising temporary events, should interpret 'public safety' widely to include freedom from danger or harm.
- 8.15 For licensed or certificated premises and for temporary events, public safety must be kept under review and where changes to operational practices/events occur, or the customer profile changes, a review of risk assessments must be undertaken.
- 8.16 Fire safety is governed by the Regulatory Reform (Fire Safety) Order 2005 and is not something with which the Licensing Authority will normally become involved.

Where a Responsible Authority has recommended a safe capacity limit on all or part of a premises the Licensing Authority will normally expect an applicant/authorisation holder to follow such a recommendation unless there are good reasons for not doing so.

Prevention of Public Nuisance

8.17 The Licensing Authority remains sensitive to the expectations and needs of different parts of the community in respect of leisure and cultural pursuits, and will view applications accordingly. The impact of those activities on people who have to live, work and sleep within the local vicinity of a licensed premises or event will also be considered. If the impact of licensed activities is

disproportionate and unreasonable or markedly reduces the amenity value of the area to local people, then the Licensing Authority will take account of this when exercising its functions.

- 8.18 The Licensing Authority considers that the potential for public nuisance can be prevented or much reduced by good design and planning during new or ancillary construction works, by the provision of good facilities and effective management. This will require appropriate advice at the planning and development stages of new projects. Applicants should consider carefully the suitability of the premises for the type of activity to be undertaken, particularly in terms of ventilation, cooling, noise breakout and noise/vibration transmission to adjoining premises.
- 8.19 Licence holders already in receipt of complaints should seek an early remedy to any confirmed problem. The organisers of temporary events should seek to pre-empt potential nuisance, especially if complaints have previously arisen at the same venue.
- 8.20 The Licensing Authority expects holders of authorisations to use their risk assessment and operating schedules to review and, if need be, to make necessary improvements to the premises or to operational practices, in order to prevent public or statutory nuisance.
- 8.21 Where the provisions of existing legislation prove inadequate or inappropriate for control purposes, the Licensing Authority will consider imposing licence conditions. Any condition deemed appropriate and imposed by the Licensing Authority to promote the prevention of public nuisance will focus on measures within the direct control of the licence holder or designated premises supervisor.
- 8.22 Any such conditions imposed will be tailored to the style and characteristics of the premises and the type of activities expected to take place. Any conditions added will be precise and enforceable and will be unambiguous and clear in what they intend to achieve.

Protection of Children from Harm

- 8.23 The Act details a number of legal requirements designed to protect children in licensed premises. The Licensing Authority is concerned to ensure that holders of authorisations, including organisers of temporary events, create safe environments (in terms of physical, moral and psychological welfare) for children who may be on the premises. Children should be unable to access alcohol or drugs and be subject to an appropriate level of care and supervision at all times.
- 8.24 The Act prohibits children under the age of 16 years old and unaccompanied by an adult, to be present in licensed premises (including premises operating under a TEN) being used primarily or exclusively for consumption of alcohol.

8.25 The admission of children to any premises will otherwise normally be left to the discretion of the individual licensee/event organiser, as the Act does not prohibit children from accessing licensed premises. Where children are accompanied and supervised by a responsible adult, additional measures, should not normally be necessary. The Licensing Authority supports the view that children should enjoy access to a range of licensed premises, but cannot impose conditions requiring the admission of children to any premises.

The Licensing Authority will judge the merits of each application before deciding whether or not to impose conditions restricting access by children. Conditions which may be relevant in this respect are outlined in the Government Guidance.

- 8.26 In premises where alcohol is sold or supplied it is a mandatory condition that premises licence holders will operate a recognised Age Verification Scheme. The Licensing Authority supports the Challenge 25 scheme and where this is not proposed within the operating schedule, alternative and similarly rigorous controls should be detailed. The Licensing Authority recommends that the premise licence holder operates a method for recording when a sale is refused as part of any age challenge scheme (also known as a refusals book).
- 8.27 The Licensing Authority expects that customers should be confronted by clear and visible signs on the premises that underage drinking constitutes an offence in law and that they may well be required to produce proof of their age to a member of staff. Organisers of temporary events should apply similar safeguards in their undertakings.
- 8.28 Venue operators seeking premises licences and club premises certificates can volunteer prohibitions and restrictions in their operating schedules because their own risk assessments have determined that the presence of children is undesirable or inappropriate. These will become conditions attached to the licence or certificate where no relevant representations are received by the Licensing Authority.
- 8.29 The Licensing Authority regards Warwickshire County Council as being the primary source of advice and information on children's welfare and would normally expect any advice/recommendations from the County Council to be followed unless there are good reasons for not doing so. The Licensing Authority will attach appropriate conditions where these appear necessary to protect children from moral, psychological or physical harm. It is also reasonable for the licensing authority to expect the responsible authorites to intervene where the basis for the intervention falls within the remit of that other authority. For example, the police should take appropriate steps where the basis for the review is concerned about crime and disorder or the sexual exploitation of children.
- 8.30 The Licensing Authority will consider the need to protect children from sexual exploitation when undertaking licensing functions.

- 8.31 In order to prevent children from seeing films incompatible with their age, licence holders who exhibit films will be expected to impose and enforce viewing restrictions in accordance with the recommendations of the British Board of Film Classification. In exceptional cases e.g. where the BBFC has not classified a film then the Licensing Authority may specify viewing restrictions which an authorisation holder will be expected to comply with.
- 8.32 It is expected that authorisation holders will ensure that, whenever children are in the vicinity of a film or exhibition that is being shown/staged in a multipurpose premises, sufficient ushers/stewards (minimum 18 years old) will be in attendance at the entrance the viewing rooms at all times to ensure children cannot enter or view the film or exhibition.
- 8.33 Children have access to a range of regulated public entertainment venues and may be present as members of a viewing audience or as performers in their own right. The Licensing Authority expects authorisation holders including those organising temporary events, to make proper provision for child safety and welfare during such events. Notwithstanding public safety issues, supervisory arrangements must be reflected within operating schedules. Suitable monitoring strategies should also be in place to ensure that supervisory levels are appropriate.
- 8.34 Where a large number of children are likely to be present on any licensed premises, for example, a children's show or pantomime, the Licensing Authority may require that there is an adequate number of adult staff at places of entertainment to control access and egress of children and to protect them from harm. Children present at events as entertainers will be expected to have a nominated adult responsible for each child performer.

MANDATORY LICENSING CONDITIONS

9.1 The Government has introduced a range of mandatory conditions aimed at establishing minimum standards for the way certain licensable activities are conducted. The conditions apply to all appropriate premises. A full list of the mandatory conditions can be found on the Gov.uk website.

OTHER CONSIDERATIONS

Relationship with Planning

- 10.1 The planning and licensing regimes involve consideration of different (albeit related) matters. The Council's Licensing and Regulatory Committee and Sub

 Committees are not bound by decisions made by the Council's Planning Committee and vice versa.
- 10.2 The grant of any application or variation of a licence which involves a material alteration to a building would not relieve the applicant of the need to apply for planning permission or building control approval where appropriate.
- 10.3 There are also circumstances when as a condition of planning permission; operating hours are set for the use of the premises for commercial purposes. Where these hours are shorter than the licensing hours, the applicant must observe the planning restrictions. Premises operating in breach of their planning consent would be liable to enforcement action under planning law.
- 10.4 The Local Planning Authority may also make representations against a licensing application in its capacity as a Responsible Authority, where such representations relate to one or more of the licensing objectives.

Applications

- 10.5 An applicant may apply under the terms of the Act for a variety of authorisations and any such application will be considered on its individual merits. Any person may make representations on an application or seek a review of a licence or certificate where provision has been made for them to do so in the Act.
- 10.6 The Licensing Authority expects each and every applicant for a premises licence, club premises certificate or variation to address how they intend to promote the licensing objectives.
- 10.7 In determining a licence application the Licensing Authority will take each application on its merits. Licence conditions will only be imposed following a hearing or in order to promote the licensing objectives and will only relate to matters within the control of the applicant. Licence conditions will not normally be imposed where other regulatory provisions are in force (e.g. planning, health and safety at work, fire safety and building control legislation) so as to avoid confusion and duplication, except where they can be exceptionally justified to promote the licensing objectives.
- 10.8 The Licensing Authority will impose only such conditions as are proportionate towards promoting the licensing objectives and which do not impose unnecessary burdens and which are appropriate to the individual size, style and characteristics of the premises and events concerned.
- 10.9 In considering applications, the Licensing Authority will primarily focus on the direct impact of activities taking place at the licensed premises on members of the public living, working or engaged in normal activity in the area concerned. The Licensing Authority recognises that licensing law is not the primary mechanism for the general control of nuisance and anti-social behaviour by individuals once they are away from the licensed premises and, therefore beyond the direct control of the individual, club or business holding the licence, certificate or authorisation concerned.
- 10.10 Conditions include any limitations or restrictions attached to a licence, certificate or other authorisation and essentially are the steps or actions the holder of the authorisation will be required to take or refrain from taking at all times when licensable activities are taking place at the premises in question.

Deregulated Public Entertainment

10.11 As a result of the Live Music Act 2012 and The Legislative Reform (Entertainment Licensing) Order 2014 most public entertainment taking place between 08:00 and 23:00 hrs has, subject to certain conditions, been deregulated and removed from the scope of Licensing Authority control. No authorisation is required where public entertainment is being provided under these statutory exemptions. However, if exempt public entertainment is or will be taking place as well as licensable activities (e.g. selling alcohol) then an authorisation covering the licensable activities will still be required. Operators of 'on-licensed' premises should also note that it is possible to re-introduce full licensing controls over public entertainment where a premises licence or a club premises certificate has been reviewed and a Licensing and Regulatory Sub-Committee determines that it is appropriate for such controls to be re-introduced.

Public Spaces Protection Order

10.12 At the time of writing, The Warwick District Public Spaces Protection Order (PSPO) (Introduced by The Anti Social Behaviour, Crime and Policing Act 2014) are under review. Regardless of the outcome of the PSPO review the local authority must have regard to section 62 and 63 of the 2014 Act which limits what can be restricted in relation to alcohol. For example, where a PSPO covers alcohol prohibition, section 62 of the 2014 Act lists a number of premises to which an Order cannot apply – such as on licensed premises (or within its curtilage), premises which by virtue of Part 5 of the Licensing Act 2003 may at the relevant time be used for the supply of alcohol, or facilities or activities relating to the sale or consumption of alcohol which are at the relevant time permitted by virtue of a permission granted under section 115E of the Highways Act 1980 (highway-related uses). Section 63 makes it an offence only when a person refuses or fails to comply with a reasonable requirement from an authorised person not to consume or surrender alcohol.

BEST PRACTICE SCHEMES

11.1 The Licensing Authority supports best practice schemes for licensed premises. Premises in an area covered by a scheme are encouraged to become members of the scheme.

INTEGRATING STRATEGIES AND THE AVOIDANCE OF DUPLICATION

12.1 By consulting widely prior to this policy statement, the Licensing Authority has taken full account of local policies covering crime prevention, anti-social behaviour, culture, transport, planning and tourism as part of an integrated strategy for the Licensing Authority, Police and other agencies. Many of these strategies may not be directly related to the promotion of the licensing objectives, but indirectly impact upon them.

- 12.2 There are a number of wider issues which may need to be given due consideration when dealing with applications. The Council's Licensing and Regulatory Committee can request reports, where it thinks it is appropriate on the following areas:-
 - The needs of the local tourist economy, to ensure that these are reflected in their considerations;
 - The employment situation and the need for new investment and employment where appropriate; and
 - The general impact of alcohol related crime and disorder.
 - The general impact of alcohol related harms to health.

Crime Prevention Strategies

- 12.3 Crime prevention and drug and alcohol misuse policies and the input of the South Warwickshire Community Safety Partnership (SWCSP) will be reflected in licence conditions as far as possible.
- 12.4 The SWCSP is committed to making South Warwickshire a safe place in which to live work and visit. It is the role of the SWCSP to strategically plan, commission and oversee services that tackle crime and disorder and address drug and alcohol misuse.

Duplication

12.5 When considering any application the Licensing Authority will avoid duplication with other regulatory regimes as far as possible. Therefore the Licensing Authority will not attach conditions to a licence in relation to a matter covered by another regulatory regime unless going beyond such a regime is considered appropriate for the promotion of the licensing objectives in the particular circumstances.

Promotion of Equality

- 12.6 The Licensing Authority in carrying out its functions under the Act is obliged to have 'due regard' to the need to eliminate unlawful discrimination, harassment and victimisation, to advance equality of opportunity and to foster good relations between persons with different protected characteristics. The protected characteristics are age, disability, gender reassignment, pregnancy and maternity, race, religion or belief, sex and sexual orientation.
- The Government guidance advises that conditions should not be attached to authorisations which would duplicate existing statutory requirements. The Licensing Authority therefore takes this opportunity to remind operators of premises of their duties towards disabled persons (including performers) on their premises under the Building Regulations and the Equalities Act 2010. This includes a duty that any person who provides a service to the public must make

reasonable adjustments to any physical feature that makes it impossible or unreasonably difficult for a disabled person to access a service, or to provide the services by a reasonable alternative means.

ENFORCEMENT

- 13.1 The Licensing Authority has an established working relationship with the Police and other responsible authorities on enforcement issues through the Multi Agency Licensing Enforcement Meeting. This provides a more efficient deployment of resources targeting high risk premises and activities.
- 13.2 This enforcement regime follows the Government's Regulators' Code in that it follows the basic principles of openness, helpfulness, proportionality and consistency. The Licensing Authority has a separate enforcement policy in respect of licensing.
- 13.3 Licensed premises are visited by the Responsible Authorities and the Licensing Authority to carry out targeted inspections to check that the premises licence or certificate is being complied with, to check compliance with other legislation and/or deal with complaints that have been received.
- 13.4 On some occasions a multi-agency group (representing a number of Responsible Authorities) will visit premises. The officers will check the premises/activities relevant to their particular role.
- 13.5 There are several enforcement options that will be used as appropriate and in line with the Licensing Authority's licensing enforcement policy. These options include:
 - Verbal advice this covers minor complaints/infringements where advice is seen as the most appropriate way to deal with the issue.
 - Written warning this is a step-up from verbal advice and holders of authorisations are given a letter recording the warning given and containing the details of any necessary remedial action.
 - Action planning this plan will be written down and given to the holder of the authorisation and designated premises supervisor. It explains what actions are required, within a timescale, for compliance with the licensing objectives, specific legislation or conditions. It will be regularly reviewed and if compliance has been achieved it will be terminated. If areas of noncompliance remain a more formal enforcement option further up the scale may be selected in order to achieve compliance.
 - Review any person may call for a review of a licensed premises where there is evidence that the licensing objectives are not being promoted. The holder of the authorisation will have to attend a review hearing in front of the Licensing Sub Committee who may decide, based on the evidence

submitted to them, to take no action, to remove the DPS, to revoke, suspend, or amend the licence or apply additional conditions.

- Prosecution Under the Licensing Act 2003 certain offences can be prosecuted by the Licensing Authority/Director Of public Prosecutions/Weights and Measures Authority (Trading Standards). In addition, Responsible Authorities have a wide range of powers to institute prosecution under other specific legislation.
- Closure several of the Responsible Authorities have the power to close licensed premises if they deem it necessary. The Licensing Authority also has powers to request closure through the Magistrates court for continuing unauthorised alcohol sales.

ADMINISTRATION, EXERCISE AND DELEGATION OF FUNCTIONS

Licensing and Regulatory Committee

- 14.1 The majority of powers given to the Licensing Authority by the Act have been delegated by the Council to the Licensing and Regulatory Committee and Officers. The Licensing and Regulatory Committee has in turn established Sub-Committees to determine some matters under the Act.
- 14.2 The Council's Constitution defines those responsibilities and is available for inspection on the Council's website, but a summary of responsibility is set out in Table 1 below.

14.3 Table 1:

Matters to be dealt with	Full Committee	Sub Committee	Officers
Application for personal licence		If a police objection	If no objection made
Application for premises licence/club premises certificate		If a relevant representation made	If no relevant representation made
Application for provisional statement		If a relevant representation made	If no relevant representation made
Application to vary premises licence/club premises certificate		If a relevant representation made	If no relevant representation made
Application to vary designated premises supervisor		If a police objection	All other cases
Request to be removed as designated premises supervisor			All cases
Application to transfer of premises licence		If a police objection	All other cases
Application for interim authorities		If a police objection	All other cases
Application to review premises licence/club premises certificate	All cases		
Decision on whether a complaint is irrelevant frivolous vexatious etc.			All cases
Decision to object when local authority is a consultee and not the relevant authority considering the application	All cases		
Determination of a police/EHO objection to a temporary event notice		All cases	
Determination of a Minor Variation application			All cases
Removal of the requirement for a designated premises supervisor at community premises		If a police objection	All other cases

14.4 However Council has retained the power to set the Council's Licensing Policy Statement, but it will seek the views of the Licensing & Regulatory Committee before determining any amendments.

Application forms And Process

- 14.5 All application forms will be in the prescribed format. The operating schedule will form part of the completed application form for a premises licence and a club premises certificate. The applicant will have to detail the steps that will be taken to promote the licensing objectives. Applicants should carry out a risk assessment before they apply for a licence.
- 14.6 Applicants are encouraged to fully consult the police and other statutory services well in advance of carrying out their risk assessments and submitting their applications. Application forms and guidance leaflets will be available from the Licensing Authority, including contact names for each of the responsible authorities. Most applications will require additional documentation and a fee to be included with the form. Incomplete applications will not be considered and will be returned to the applicant.
- 14.7 Where national guidance permits, on line applications will be accepted providing the necessary documentary attachments are uploaded into the application and the appropriate fee paid.
- 14.8 Applicants are encouraged to make themselves aware of any relevant planning and transportation policies, tourism and cultural strategies and local crime, alcohol, drug and disorder strategies in order to take these into account, where appropriate, when formulating their operating schedule.

COMMENTS ON THIS POLICY

15.1 The statement of licensing policy will be reviewed on a regular basis. Individuals and organisations that wish to comment on the policy are invited to send their comments in writing to:

Warwick District Council, Licensing Team, Riverside House, Milverton Hill, Royal Leamington Spa, CV32 5HZ

Email: licensing@warwickdc.gov.uk

APPENDIX 1 – Set of Model Conditions

Warwick District Council has produced this document to assist and support applicants and existing licence holders through the application process. It has also been designed for the consideration of responsible authorities and the Council's Licensing and Regulatory Committee.

When deciding to grant or vary a premises licence under the Licensing Act 2003, the licensing authority may do so subject to conditions which it considers are appropriate for the promotion of one or more of the licensing objectives.

Those applying for a premises licence, club certificate, variation of a premises licence or variation of a club certificate may also wish to consider those conditions which would promote the licensing objectives when completing the operating schedule.

In determining what conditions are appropriate, it will be necessary to consider the individual circumstances of the premises, including:-

- The nature and style of the venue,
- The activities being conducted there,
- The location, and,
- Anticipated clientele

Guidance for operating schedule

Conditions on a premises licence or club premises certificate are important in setting the parameters within which premises can lawfully operate.

Licensing authorities should be satisfied that it is appropriate to impose conditions to promote one or more of the four licensing objectives. Conditions should be tailored to the particular circumstance of an individual licensed premises and determined on a case-by-case basis.

Under no circumstances should licensing authorities regard pools of conditions as standard conditions to be automatically imposed in all cases.

Prevention of Crime and Disorder

- 1. There shall be no sales of alcohol for consumption off the premises after (time).
- 2. No single cans or bottles of beer or cider or spirit mixtures shall be sold at the premises.
- 3. No miniature bottles of spirits of 20 cl or below shall be sold from the premises.
- 4. No beer, lager, cider, ale or spirit mixers with an alcohol by volume content above (insert percentage) will be sold or offered for sale.
- 5. Each self-serve pump must be covered, in full, by the CCTV system.
- 6. Only craft beer or ale is permitted to be dispensed from the self-service pumps and will only be available in measures of (measure).
- 7. Only wine is permitted to be dispensed from the self-service wine dispenser and will only be available in a maximum measure of (measure).
- 8. When a self-service dispenser is in use a notification system must be in place to alert a member of bar staff.
- Regular meetings will take place between the Designated Premises Supervisor, Warwickshire Police and Licensing authority. The meetings will take place every (number) months.
- 10. All door supervisors, and other persons engaged at the premises, for the purpose of supervising or controlling queues or customers, must wear (high visibility jackets or vests or armbands).
- 11. Door supervision must be provided on (specify days). Door supervisors must be on duty from (insert hours) and must remain on duty until the premises are closed and all the customers have left.
- 12. Door supervisors must be provided with radios to enable them to contact each other and the duty manager at the premises.
- 13. On/at (specify days/hours) at least (insert number) of SIA registered door supervisors must be on duty at the premises (may specify location at the premises or as shown on the plan).
- 14. Where SIA registered door supervisors are used at the premises, a record must be kept of their SIA registration number and the dates and times when they are on duty. That register shall be available for inspection on reasonable request Authorised Officer of the Council, the Security Industry Authority or a Police Constable and shall contain the following details:-
 - the door supervisor's name, date of birth and home address;
 - his / her Security Industry Authority licence number;
 - the time and date he / she starts and finishes duty;

- each entry shall be signed by the door supervisor.
- 15. Any door staff register shall be available for inspection on demand by an Authorized Officer of the Council, the Security Industry Authority or a Police Constable and will be retained on the premises for a period of 12 months from the date of the last entry.
- 16. The Premises Licence holder / Designated Premises Supervisor will operate to a written dispersal policy which ensures the safe and gradual dispersal of customers from the premises. The Premises Licence holder / Designated Premises Supervisor will ensure that staff receive training on the policy.
- 17. CCTV to be installed and the premises licence holder must ensure that:
 - a. CCTV cameras are located within the premises to cover all public areas.
 - b. The system records clear images permitting the identification of individuals.
 - c. The CCTV system is able to capture a minimum of 12 frames per second and all recorded footage must be securely retained for a minimum of 28 days.
 - d. The CCTV system operates at all times while the premises are open for licensable activities'. All equipment must have a constant and accurate time and date generation.
 - e. The CCTV system is fitted with security functions to prevent recordings being tampered with, i.e. password protected.
 - f. Downloads will be provided to the Police upon reasonable request in line with the Data Protection Act 2018.
 - g. Signed off by Warwickshire Police Design Out Crime Officer.
- 18. The Premises Licence holder / Designated Premises Supervisor is to provide the Police with the contact details of at least two members of staff (or other person(s)) who are trained and familiar with the operation of the equipment so that, at the expense of the Premises Licence holder, they are able to check that the equipment is operating properly and that they are able to provide copies of recorded data upon request and within no more than 12 hours from the time of the request.
- 19. The Premises Licence holder / Designated Premises Supervisor must notify the Licensing Office or the Police in the event of CCTV breakdown or malfunction as soon as is reasonably practicable and in any event within 24hrs.
- 20. No open vessels to leave the premises at any time.
- 21. No open vessels to be taken outside the curtilage of the premises at any time.
- 22. Empty bottles must be placed into locked bins so as to prevent them from being used as weapons.

- 23. Toughened recycled and recyclable polycarbonate glasses or bottles (or similar compostable/reusable alternative) to be used in the outside areas of the premises.
- 24. Toughened recycled and recyclable polycarbonate glasses or bottles (or similar compostable/reusable alternative) will be used when requested by Warwickshire Police.
- 25. Drinks must only be served in toughened recycled and recyclable polycarbonate glasses or bottles (or similar compostable/reusable alternative) containers.
- 26. No customers carrying glassware shall be admitted to the premises at any time that the premises are open to the public.
- 27. SIA door staff will be required to remove all alcohol from customers who are queuing to enter the premise or entry to be refused. This alcohol must then be disposed of immediately in a bin provided at the premises.
- 28. A Personal Licence holder must be on the premises at all times when open to the public.
- 29. A Personal Licence holder must be on the premises on (state days) (time) between (time) hours and close of business.
- 30. The designated premises supervisor will ensure that he/she gives written authorisation to individuals whom they are authorising to sell alcohol in their absence. This should be maintained and made available for viewing by Authorised Officers.
- 31. Details of the names, addresses and up-to-date contact details for the Designated Premises Supervisor and all Personal Licence holders shall be maintained and kept on the premises.
- 32. Premises will participate in any Police or responsible authority awareness campaign or training that is relevant to the sale of alcohol, use of drugs or entertainment.
- 33. The Premises Licence holder shall have a written policy in relation to drugs which will include search, seizure and disposal of drugs and weapons. Staff will be provided with training on the policy, including drugs awareness.
- 34. A suitable receptacle for the safe retention of illegal substances will be provided and arrangements made for the safe disposal of its contents as agreed with Warwickshire Police.
- 35. There shall be displayed on the premises, information regarding drugs awareness. / Zero tolerance policy.
- 36. As soon as possible, and in any event within 1 month from the grant of this licence, the premises shall join the local Pubwatch or other local crime reduction scheme approved by the police, and local radio scheme if available.

- 37. The premises will be a member of the locally approved radio scheme and abide by its policies and procedures.
- 38. The premises is to maintain an incident book to record details of the following:-
 - Any violence or disorder on or immediately outside the premises,
 - Any incident involving controlled drugs (supply / possession or influence on the premises,
 - Any other crime or criminal activity on the premises,
 - Any call for police assistance to the premises,
 - Any ejection from the premises,
 - Any first aid/other care given to a customer.
- 39. An incident book to made available for inspection by a responsible authority on reasonable request.
- 40. In the event that a serious assault is committed on the premises (or appears to have been committed) the management will immediately ensure that:-
 - (a) The police (and, where appropriate, the Ambulance Service) are called without delay;
 - (b) All measures that are reasonably practicable are taken to apprehend any suspects pending the arrival of the police;
 - (c) The crime scene is preserved so as to enable a full forensic investigation to be carried out by the police; and
 - (d) Such other measures are taken (as appropriate) to fully protect the safety of all persons present on the premises.
- 41. Any staff employed at the premises will be provided with training on first appointment and on a regular basis thereafter. Training will include (delete where applicable);
 - Drunk awareness
 - Drugs awareness
 - Age verification training
 - Conflict management training
 - First aid
- 42. A written record will be kept of all training carried out. This record must be kept on the premises and made available for inspection by a responsible authority on reasonable request.
- 43. No entry / re-entry 1 hour before permitted hours.
- 44. No entry / re-entry after (time) (days).
- 45. Any queue (in a designated queuing area) to enter the premises must be supervised at all times by door supervisors.
- 46. Any (designated) queuing area must be within suitable barriers.

- 47. Any outside areas to be demarked by physical barriers or similar with clear signs displayed to instruct patrons that vessels must not be taken outside said area.
- 48. The premises must only operate as a restaurant:
 - in which customers are seated at a table
 - which provide food in the form of substantial table meals that are prepared on the premises and are served and consumed at the table using non-disposable crockery
 - which do not provide any take away service of food or drink for immediate consumption, and where alcohol must not be sold, supplied, or consumed on the premises otherwise than to persons who are taking substantial table meals and provided always that the consumption of alcohol by such persons is ancillary to taking such meals
- 49. The supply of alcohol to customers must be by waiter or waitress service only.
- 50. Alcohol consumed outside the premises building shall only be consumed by patrons seated at tables.

Prevention of Public Nuisance

- 1. Prominent, clear and legible notices must be displayed at all exits requesting that customers respect the needs of local residents and to leave the premises and area quietly.
- 2. The licence holder or his representative shall conduct regular assessments of the noise coming from the premises on every occasion the premises are used for regulated entertainment and shall take steps to reduce the level of noise where it is likely to cause disturbance to local residents. A written record shall be made of these assessments in a log book. This record must be kept on the premises and made available for inspection by a responsible authority on reasonable request.
- 3. All external doors and windows shall be kept closed when regulated entertainment is being provided, except for access and egress and in the event of an emergency.
- 4. The beer garden / outside area is not to be used/occupied after (x) hours daily.
- 5. There will be no external loud speakers.
- 6. The Premises Licence holder / Designated Premises Supervisor will adopt a "cooling down" period where music volume is reduced (insert minutes) before the closing time of the premises.
- 7. At an appropriate time before closing time, announcements shall be made reminding customers to leave quietly.
- 8. Where the premises provide food to the public for consumption off the premises, there shall be provided at or near the exits, (insert number) waste bins to enable the disposal of waste food, food containers, wrappings etc.
- 9. Where the premises provide food for consumption off the premises, the public area immediately surrounding the premises shall be cleared of waste food, food containers, wrapping etc. at the end of trading on each day. Such refuse shall be placed in a container designed for the storage and disposal of refuse and waste foods.
- 10. The Premises Licence holder / Designated Premises Supervisor will ensure that litter arising from people using the premises is cleared away daily and that promotional materials such as flyers do not create litter.
- 11. Outside areas and activity must cease and be cleared at (time).
- 12. The beer garden / outside area(s) is not to be used/occupied after (time) hours daily.
- 13. With the exception of smokers, the outside area shall not be used by customers after (time).
- 14. Drinks shall not be permitted to be consumed in the outside area after (time).

- 15. In relation to the (specified function room) there shall be no admission after (midnight) other than to (1) residents of the hotel and their bona fide guests, or (2) persons attending a pre-booked function.
- 16. Licensable activities at events in the (specified function room) shall only be provided at pre-booked ticketed events.
- 17. Customers shall not enter or leave the premises from / by (insert specific entrances or exits), except in the event of an emergency.
- 18. The licence holder (or his/her nominees) shall ensure that exits are manned at closing time to ensure that patrons leave the area quickly and as quietly as possible.
- 19. The licence holder (or his/her nominees) shall ensure that any queue to enter the premises which forms outside the premises is orderly and supervised by door staff so as to ensure that there is no public nuisance or obstruction to the public highway.
- 20. There shall be no admittance or re-admittance to the premises after (time) except for patrons permitted to temporarily leave the premises to smoke.
- 21. Patrons permitted to temporarily leave and then re-enter the premises, e.g. to smoke, shall be limited to (number) persons at any one time.
- 22. Patrons permitted to temporarily leave and then re-enter the premises, e.g. to smoke, shall not be permitted to take drinks or glass containers with them.
- 23. Patrons permitted to temporarily leave and then re-enter the premises to smoke shall be restricted to a designated smoking area defined as (specify location).
- 24. Clear and legible notices shall be prominently displayed at any area used for smoking requesting patrons to respect the needs of local residents and use the area quietly.
- 25. All outside tables and chairs shall be prohibited from use after (time) each day.
- 26. All tables and chairs shall be removed from the outside area by (time) each day.
- 27. No external seating shall be provided at the premises.
- 28. All external doors and windows shall be kept closed after (time) hours, or at any time when regulated entertainment is being provided, except for the immediate access and egress of persons or in the event of an emergency.
- 29. Staff shall check prior to the commencement of regulated entertainment, and periodically during regulated entertainment that all windows and doors are shut.
- 30. Loudspeakers shall not be located in the entrance lobby or outside the premises building.
- 31. No speakers for amplification of music shall be placed on the outside of the premises or on the outside of any building forming a part of the premises.

- 32. A noise limiter must be fitted to the musical amplification system set at a level determined by and to the satisfaction of an authorised officer of the local authority's Environmental Health Service, so as to ensure that no noise nuisance is caused to local residents or businesses. The operational panel of the noise limiter shall then be secured by key or password to the satisfaction of officers from the Environmental Health Service and access shall only be by persons authorised by the Premises Licence holder. The limiter shall not be altered without prior agreement with the Environmental Health Service. No alteration or modification to any existing sound system(s) should be effected without prior knowledge of an authorised officer of the Environmental Health Service. No additional sound generating equipment for the purposes of providing regulated entertainment shall be used on the premises without being routed through the sound limiter device.
- 33. A sound limiting device shall be fitted to any musical amplification system and set at a level determined by and to the satisfaction of an authorised officer of Warwick District Council's Environmental Health service to ensure that no noise nuisance is caused to local residents. The limiter shall not be altered without prior agreement with the Environmental Health Service.
- 34. No regulated entertainment shall take place in the outdoor areas at any time.
- 35. The provision of live music shall be limited to no more than two performers.
- 36. All outdoor entertainment shall be unamplified.
- 37. Suitable means of ventilation shall be provided and maintained at the premises to enable doors and windows to be closed whilst regulated entertainment is being provided.
- 38. Where the premises provide food to the public for consumption off the premises, there shall be provided at or near the exits, (insert number) waste bins to enable the disposal of waste food, food containers, wrappings etc.
- 39. The Premises Licence holder (or his/her nominees) will ensure that litter arising from people using the premises is cleared away daily and that promotional materials such as flyers do not create litter.
- 40. No waste or recyclable materials, including bottles, shall be moved, removed from or placed in outside areas between (time) hours and (time) hours on the following day.
- 41. During the hours of operation of the premises, the licence holder (or his/her nominees) shall ensure sufficient measures are in place to remove and prevent litter or waste arising or accumulating from customers in the area immediately outside the premises, and that this area shall be swept and or washed, and litter and sweepings collected and stored in accordance with the approved refuse storage arrangements by close of business.

- 42. No collections of waste or recycling materials (including bottles) from the premises shall take place between (time) and (time) on the following day.
- 43. No deliveries to the premises shall take place between (time) and (time) on the following day.

Public Safety

- 1. A designated room, space or location to be provided within the premises to create a safe environment which is monitored by a trained and DBS checked member of staff or volunteer.
- 2. All doors/gates through which persons may have to pass whilst making their way from the premises shall be readily and easily openable from within without the use of a key, code, card etc.
- 3. Once a licence has been granted a Fire Risk Assessment is to be kept on the premises and be available for inspection by an authorised officer.
- 4. The premises should be provided with a means for raising the alarm in the event of fire.
- 5. Staff should be aware of the siting of extinguishers, of their correct method of operation and know which equipment is appropriate for a particular fire. The equipment should be so placed as to be readily available for use. At least one appliance should be placed at or near to the exit from a floor and, where extinguishers are provided for special risks, as far as practical be sited close to the risk for immediate use.
- 6. There shall be maintained on the premises at all times an adequate and appropriate supply of first aid equipment and materials.
- 7. At all times when the public are present, at least one person who holds a current recognised first aid certificate or award shall be present on the premises. Where more than one such person is present, their duties shall be clearly defined.
- 8. The premises licence holder shall develop and operate a procedure for dealing with unwell members of the public including those who appear to be affected by alcohol and drugs. Staff will be appropriately trained in such procedures.
- 9. The maximum number of persons allowed in the premises shall be (insert numbers, areas and occasions).
- 10. A person who is responsible for the management of the premises shall at all times be aware of the number of persons on the premises and shall if required to do so, give that information to an authorised person.
- 11. At all times door supervisors on duty, numbers or persons inside the venue shall be recorded by way of a clicker system or similar, and shall if required to do so, give that information to an authorised person.
- 12. All emergency doors shall be maintained effectively self closing and not held open other than by an approved device.
- 13. The Premises Licence holder / Designated Premises Supervision must develop and operate a policy which ensures the safe evacuation of disabled people in the event of an emergency. All staff shall be made aware of these arrangements.

14. Staff or attendants shall be readily identifiable to members of the public.

Protection of Children from Harm

- 1. Signage to be displayed around the premises in prominent places informing both staff and customers of the 'Challenge 25' policy.
- 2. Challenge 25 policy promoted on business website.
- 3. A notice(s) shall be displayed in and at the entrance to the premises where they can be clearly seen, indicating that there is a "Challenge 25" policy in place at the premises.
- 4. A prompt or reminder to staff, at the point of sale, to consider whether a sale or challenge is to be made.
- 5. All deliveries of alcohol must be made by a person over the age of 18 years.
- 6. Delivery: ID checks by courier at the point of delivery in line with Challenge 25 age verification policy.
- 7. No person under the age of 18 shall be permitted access to the premises when entertainment of an adult nature is taking place.
- 8. No person under the age of (insert age) shall be permitted to remain on the premises after (insert hours).
- 9. Under 18's events will not take place without prior consultation with the Police and Licensing Authority.
- 10. Under 18's events will not take place at the premises.
- 11. No persons under the age of 18 years will be allowed on the premises after (time) unless accompanied by a responsible adult of 18 years or above and with the express permission and knowledge of the DPS or someone acting under their authority.
- 12. The premises will operate a "Challenge 25" proof of age policy.
- 13. The premises is to maintain a refusals book/record to record the details of incidents where a member of staff has refused to sell alcohol to a person suspected of being under the age of 18. The Premises Licence holder / Designated Premises Supervisor or nominated representative shall regularly monitor the book make a record of these checks. The book must be made available to a Police Constable/Authorised Officers of the Licensing Authority on request.
- 14. Any person who is authorised to sell alcohol at the premises will be provided with training on first appointment and on a regular basis thereafter. Training will include information on how to prevent underage sales and any other relevant matters. A written record will be kept of all training provided and this record will be kept on the premises for inspection by any Responsible Authority.

- 15. No persons under the age of 18 years to operate the self-serve pumps at any time.
- The above are generic conditions that applicants may wish to include within their application to speed up any discussions with the relevant responsible authorities. The use of these conditions does not guarantee the granting of a premises licence. Each application will be assessed on its own merits and in the context of its location and potential to impact on the licensing objectives. Additional conditions, or amendments to these model conditions, may be necessary in order to uphold the four licensing objectives. These model conditions are not exhaustive and do not prevent you from volunteering any alternative measures that you believe are more appropriate for your particular premises.
- Applicants will also need to consider whether their premises falls with any designated cumulative impact zones (CIZ) as higher standards will be applied.
- For premises licences relating to outdoor music events, festivals, etc. applicants are encouraged to read the guidance provided by Warwick District Council's Safety Advisory Group (SAG) for information on planning their events. This can be found at: <u>www.warwickdc.gov.uk/safetyadvisorygroup</u>

APPENDIX 2 – LIST OF RESPONSIBLE AUTHORITIES

POLICE:

Chief Officer of Police Warwickshire Police Licensing Team Warwickshire Justice Centre Leamington Spa Newbold Terrace Leamington Spa Warwickshire CV32 4EL

Tel: 01926 684033 Email: <u>southwarksliquorlicensing@warwickshire.police.uk</u>

FIRE AUTHORITY:

County Fire Officer Fire Service Headquarters Warwick Street Leamington Spa CV32 5LH

Tel: 01926 423231 Email: <u>firesafety@warwickshire.gov.uk</u>

HEALTH AND SAFETY:

Warwick District Council Riverside House Milverton Hill Royal Leamington Spa CV32 5HZ

Tel: 01926 456713 Email: <u>hcphealthandsafety@warwickdc.gov.uk</u>

ENVIRONMENTAL HEALTH:

Warwick District Council Riverside House Milverton Hill Royal Leamington Spa CV32 5HZ

Tel: 01926 456714 Email: <u>pollution@warwickdc.gov.uk</u>

COMMUNITY SAFETY:

Licensing Team Warwick District Council Riverside House Milverton Hill Royal Leamington Spa CV32 5HZ

Tel: 01926 456725 Email: <u>communitysafety@warwickdc.gov.uk</u>

PLANNING:

Warwick District Council Riverside House Milverton Hill Royal Leamington Spa CV32 5HZ

Tel: 01926 456522 Email: <u>planning.enforcement@warwickdc.gov.uk</u>

WARWICKSHIRE COUNTY COUNCIL CHILD PROTECTION:

Safeguarding, Quality and Assurance People Directorate B3 Saltisford Office Park Ansell Way Warwick CV34 4UL

Tel: 01926 742379 Email: <u>licenseapplications@warwickshire.gov.uk</u>

WARWICKSHIRE COUNTY COUNCIL TRADING STANDARDS:

Simon Coupe Divisional Trading Standards Officer Old Budbooke Road Warwick CV35 7DP

Tel: 01926 414080 Email: ts@warwickshire.gov.uk

NATIONAL HEALTH SERVICE/PUBLIC HEALTH:

Public Health Department (Licensing) NHSWarwickshire/Warwickshire County Council PO Box 43 Shire Hall Warwick CV34 4SX

Email: phadmin@warwickshire.gov.uk

HOME OFFICE (Immigration Enforcement)

Alcohol Licensing Team Lunar House 40 Wellesley Road Croydon CR9 2BY

Email: <a>alcohol@homeoffice.gov.uk



Revised Guidance issued under section 182 of the Licensing Act 2003

December 2023

Revised Guidance issued under section 182 of the Licensing Act 2003

December 2023



© Crown copyright 2023

This publication is licensed under the terms of the Open Government Licence v3.0 except where otherwise stated. To view this licence, visit <u>nationalarchives.gov.uk/doc/open-government-licence/version/3</u>

Where we have identified any third party copyright information you will need to obtain permission from the copyright holders concerned.

This publication is available at https://www.gov.uk/official-documents

Any enquiries regarding this publication should be sent to us at The Alcohol Team, The Home Office, 5th Floor, Fry Building (North West), 2 Marsham Street, London, SW1P 4DF.

Email: <u>AlcoholTeam2@homeoffice.gov.uk</u>

Contents

1.	Introduction	1
2.	The licensing objectives	6
3.	Licensable activities	15
4.	Personal licences	22
5.	Who needs a premises licence?	39
6.	Club premises certificates	45
7.	Temporary Event Notices (TENs)	48
8.	Applications for premises licences	55
9.	Determining applications	74
10.	Conditions attached to premises licences and club premises certificates	83
11.	Reviews	96
12.	Summary reviews	103
13.	Appeals	111
14.	Statements of licensing policy	114
15.	Licence fees	129
16.	Regulated entertainment	131
17.	Early morning alcohol restriction orders	149

1. Introduction

The Licensing Act 2003

1.1 The Licensing Act 2003 (referred to in this Guidance as the 2003 Act), its explanatory notes and any statutory instruments made under it may be viewed online at <u>www.legislation.gov.uk</u>. The statutory instruments include regulations setting out the content and format of application forms and notices. The Home Office has responsibility for the 2003 Act. However, the Department for Culture, Media and Sport (DCMS) is responsible for regulated entertainment, for which there is provision in Schedule 1 to the 2003 Act (see Chapter 16).

Licensing objectives and aims

- 1.2 The legislation provides a clear focus on the promotion of four statutory objectives which must be addressed when licensing functions are undertaken.
- 1.3 The licensing objectives are:
 - The prevention of crime and disorder;
 - Public safety;
 - The prevention of public nuisance; and
 - The protection of children from harm.
- 1.4 Each objective is of equal importance. There are no other statutory licensing objectives, so that the promotion of the four objectives is a paramount consideration at all times.
- 1.5 However, the legislation also supports a number of other key aims and purposes. These are vitally important and should be principal aims for everyone involved in licensing work. They include:
 - protecting the public and local residents from crime, anti-social behaviour and noise nuisance caused by irresponsible licensed premises;
 - giving the police and licensing authorities the powers they need to effectively manage and police the night-time economy and take action against those premises that are causing problems;
 - recognising the important role which pubs and other licensed premises play in our local communities by minimising the regulatory burden on business, encouraging innovation and supporting responsible premises;
 - providing a regulatory framework for alcohol which reflects the needs of local communities and empowers local authorities to make and enforce decisions about the most appropriate licensing strategies for their local area; and
 - encouraging greater community involvement in licensing decisions and giving local residents the opportunity to have their say regarding licensing decisions that may affect them.

The guidance

1.6 Section 182 of the 2003 Act provides that the Secretary of State must issue and, from time to time, may revise guidance to licensing authorities on the discharge of their functions under the 2003 Act. This revised guidance takes effect as soon as it is published. Where a licence application was made prior to the publication of the revised guidance, it should be processed in accordance with the guidance in effect at the time at which the application was made; the revised guidance does not apply retrospectively. However, all applications received by the licensing authority on or after the date the revised guidance was published should be processed in accordance with the revised guidance.

Purpose

- 1.7 This Guidance is provided to licensing authorities in relation to the carrying out of their functions under the 2003 Act. It also provides information to magistrates' courts hearing appeals against licensing decisions and has been made widely available for the benefit of those who run licensed premises, their legal advisers and the general public. It is a key medium for promoting best practice, ensuring consistent application of licensing powers across England and Wales and for promoting fairness, equal treatment and proportionality.
- 1.8 The police remain key enforcers of licensing law. This Guidance does not bind police officers who, within the parameters of their force orders and the law, remain operationally independent. However, this Guidance is provided to support and assist police officers in interpreting and implementing the 2003 Act in the promotion of the four licensing objectives.

Legal status

- 1.9 Section 4 of the 2003 Act provides that, in carrying out its functions, a licensing authority must 'have regard to' guidance issued by the Secretary of State under section 182. This Guidance is therefore binding on all licensing authorities to that extent. However, this Guidance cannot anticipate every possible scenario or set of circumstances that may arise and, as long as licensing authorities have properly understood this Guidance, they may depart from it if they have good reason to do so and can provide full reasons. Departure from this Guidance could give rise to an appeal or judicial review, and the reasons given will then be a key consideration for the courts when considering the lawfulness and merits of any decision taken.
- 1.10 Nothing in this Guidance should be taken as indicating that any requirement of licensing law or any other law may be overridden (including the obligations placed on any public authorities under human rights legislation). This Guidance does not in any way replace the statutory provisions of the 2003 Act or add to its scope and licensing authorities should note that interpretation of the 2003 Act is a matter for the courts. Licensing authorities and others using this Guidance must take their own professional and legal advice about its implementation.

Licensing policies

- 1.11 Section 5 of the 2003 Act requires a licensing authority to determine and publish a statement of its licensing policy at least once every five years. The policy must be published before it carries out any licensing functions under the 2003 Act.
- 1.12 However, determining and publishing a statement of its policy is a licensing function and as such the authority must have regard to this Guidance when taking this step. A licensing authority may depart from its own policy if the individual circumstances of any case merit such a decision in the interests of the promotion of the licensing objectives. But once again, it is important that it should be able to give full reasons for departing from its published statement of licensing policy. Where revisions to this Guidance are issued by the Secretary of State, there may be a period of time when the licensing policy statement is inconsistent with the Guidance (for example, during any consultation by the licensing authority). In these circumstances, the licensing authority should have regard, and give appropriate weight, to this Guidance and its own existing licensing policy statement.

Licensable activities

- 1.13 For the purposes of the 2003 Act, the following are licensable activities:
 - The sale by retail of alcohol;
 - The supply of alcohol by or on behalf of a club to, or to the order of, a member of the club;
 - The provision of regulated entertainment; and
 - The provision of late night refreshment.

Further explanation of these terms is provided in Chapter 3.

Authorisations or permissions

- 1.14 The 2003 Act provides for four different types of authorisation or permission, as follows:
 - Premises licence to use premises for licensable activities.
 - Club premises certificate to allow a qualifying club to engage in qualifying club activities as set out in Section 1 of the Act.
 - Temporary event notice to carry out licensable activities at a temporary event.
 - Personal licence to sell or authorise the sale of alcohol from premises in respect of which there is a premises licence.

General principles

1.15 If an application for a premises licence or club premises certificate has been made lawfully and there have been no representations from responsible authorities or other persons, the licensing authority must grant the application, subject only to conditions that are consistent with the operating schedule and relevant mandatory conditions. It is recommended that licence applicants contact responsible authorities when preparing their operating schedules.

Licence conditions – general principles

- 1.16 Conditions on a premises licence or club premises certificate are important in setting the parameters within which premises can lawfully operate. The use of wording such as "must", "shall" and "will" is encouraged. Licence conditions:
 - must be appropriate for the promotion of the licensing objectives;
 - must be precise and enforceable;
 - must be unambiguous and clear in what they intend to achieve;
 - should not duplicate other statutory requirements or other duties or responsibilities placed on the employer by other legislation;
 - must be tailored to the individual type, location and characteristics of the premises and events concerned;
 - should not be standardised and may be unlawful when it cannot be demonstrated that they are appropriate for the promotion of the licensing objectives in an individual case;
 - should not replicate offences set out in the 2003 Act or other legislation;
 - should be proportionate, justifiable and be capable of being met;
 - cannot seek to manage the behaviour of customers once they are beyond the direct management of the licence holder and their staff, but may impact on the behaviour of customers in the immediate vicinity of the premises or as they enter or leave; and
 - should be written in a prescriptive format.

Each application on its own merits

1.17 Each application must be considered on its own merits and in accordance with the licensing authority's statement of licensing policy; for example, if the application falls within the scope of a cumulative impact policy. Conditions attached to licences and certificates must be tailored to the individual type, location and characteristics of the premises and events concerned. This is essential to avoid the imposition of disproportionate and overly burdensome conditions on premises where there is no need for such conditions. Standardised conditions should be avoided and indeed may be unlawful where they cannot be shown to be appropriate for the promotion of the licensing objectives in an individual case.

Additional guidance

1.18 From time to time, the Home Office may issue additional supporting guidance to licensing authorities and other persons on the Gov.uk website. This supporting guidance is good practice guidance and should be viewed as indicative and subject to change. Such supporting guidance will broadly reflect but will not be part of the statutory guidance issued by the Secretary of State under section 182 of the 2003 Act. Licensing authorities may wish to refer to, but are under no statutory duty to have regard to such supporting guidance issued by the Home Office.

Other relevant legislation

- 1.19 While licence conditions should not duplicate other statutory provisions, licensing authorities and licensees should be mindful of requirements and responsibilities placed on them by other legislation. Legislation which may be relevant includes:
 - The Gambling Act 2005
 - The Environmental Protection Act 1990
 - The Noise Act 1996
 - The Clean Neighbourhoods and Environmental Act 2005
 - The Regulatory Reform (Fire Safety) Order 2005
 - The Health and Safety at Work etc. Act 1974
 - The Equality Act 2010
 - The Immigration Act 2016
 - Regulators' Code under the Legislative and Regulatory Reform Act 2006

2. The licensing objectives

Crime and disorder

- 2.1 Licensing authorities should look to the police as the main source of advice on crime and disorder. They should also seek to involve the local Community Safety Partnership (CSP).
- 2.2 In the exercise of their functions, licensing authorities should seek to co-operate with the Security Industry Authority ("SIA") as far as possible and consider adding relevant conditions to licences where appropriate. The SIA also plays an important role in preventing crime and disorder by ensuring that door supervisors are properly licensed and, in partnership with police and other agencies, that security companies are not being used as fronts for serious and organised criminal activity. This may include making specific enquiries or visiting premises through intelligence led operations in conjunction with the police, local authorities and other partner agencies. Similarly, the provision of requirements for door supervision may be appropriate to ensure that people who are drunk, drug dealers or people carrying firearms do not enter the premises and ensuring that the police are kept informed.
- 2.3 Conditions should be targeted on deterrence and preventing crime and disorder including the prevention of illegal working in licensed premises (see paragraph 10.10). For example, where there is good reason to suppose that disorder may take place, the presence of closed-circuit television (CCTV) cameras both inside and immediately outside the premises can actively deter disorder, nuisance, anti-social behaviour and crime generally. Some licence holders may wish to have cameras on their premises for the prevention of crime directed against the business itself, its staff, or its customers. But any condition may require a broader approach, and it may be appropriate to ensure that the precise location of cameras is set out on plans to ensure that certain areas are properly covered and there is no subsequent dispute over the terms of the condition.
- 2.4 The inclusion of radio links and ring-round phone systems should be considered an appropriate condition for public houses, bars and nightclubs operating in city and town centre leisure areas with a high density of licensed premises. These systems allow managers of licensed premises to communicate instantly with the police and facilitate a rapid response to any disorder which may be endangering the customers and staff on the premises.
- 2.5 Conditions relating to the management competency of designated premises supervisors should not normally be attached to premises licences. It will normally be the responsibility of the premises licence holder as an employer, and not the licensing authority, to ensure that the managers appointed at the premises are competent and appropriately trained. The designated premises supervisor is the key person who will usually be responsible for the day to day management of the premises by the premises licence holder, including the prevention of disorder. A condition of this kind may only be justified as appropriate in rare circumstances where it can be demonstrated that, in the circumstances associated with particular premises, poor management competency could give rise to issues of crime and disorder and public safety.

- 2.6 The prevention of crime includes the prevention of immigration crime including the prevention of illegal working in licensed premises. Licensing authorities should work with Home Office Immigration Enforcement, as well as the police, in respect of these matters. Licence conditions that are considered appropriate for the prevention of illegal working in licensed premises might include requiring a premises licence holder to undertake right to work checks on all staff employed at the licensed premises or requiring that evidence of a right to work check, either physical or digital (e.g. a copy of any document checked as part of a right to work check or a clear copy of the online right to work check) are retained at the licensed premises.
- 2.7 The objective to crime under the Licensing Act 2003 would include taking measures to prevent incidents of spiking which would usually be prosecuted under section 23 and 24 of the Offences Against the Person Act 1861, and section 61 of the Sexual Offences Act 2003. The following examples are within the range of behaviours that would be considered spiking. This list is not exhaustive:
 - Putting alcohol into someone's drink without their knowledge or permission
 - Putting prescription or illegal drugs into an alcoholic or non-alcoholic drink without their knowledge or permission
 - Injecting another person with prescription or illegal drugs without their knowledge or permission
 - Putting prescription or illegal drugs into another person's food without their knowledge or permission
 - Putting prescription or illegal drugs into another person's cigarette or vape without their knowledge or permission

Public safety

- 2.8 Licence holders have a responsibility to ensure the safety of those using their premises, as a part of their duties under the 2003 Act. This concerns the safety of people using the relevant premises rather than public health which is addressed in other legislation. Physical safety includes the prevention of accidents and injuries and other immediate harms that can result from alcohol consumption such as unconsciousness or alcohol poisoning. Conditions relating to public safety may also promote the crime and disorder objective as noted above. There will of course be occasions when a public safety condition could incidentally benefit a person's health more generally, but it should not be the purpose of the condition as this would be outside the licensing authority's powers (be ultra vires) under the 2003 Act. Conditions should not be imposed on a premises licence or club premises certificate which relate to cleanliness or hygiene.
- 2.9 A number of matters should be considered in relation to public safety. These may include:
 - Fire safety;
 - Ensuring appropriate access for emergency services such as ambulances;
 - Good communication with local authorities and emergency services, for example communications networks with the police and signing up for local incident alerts (see paragraph 2.4 above);

- Ensuring the presence of trained first aiders on the premises and appropriate first aid kits;
- Ensuring the safety of people when leaving the premises (for example, through the provision of information on late-night transportation);
- Ensuring appropriate and frequent waste disposal, particularly of glass bottles;
- Ensuring appropriate limits on the maximum capacity of the premises (see paragraphs 2.12-2.13, and Chapter 10; and
- Considering the use of CCTV in and around the premises (as noted in paragraph 2.3 above, this may also assist with promoting the crime and disorder objective).
- 2.10 The measures that are appropriate to promote public safety will vary between premises and the matters listed above may not apply in all cases. As set out in Chapter 8 (8.38-8.46), applicants should consider when making their application which steps it is appropriate to take to promote the public safety objective and demonstrate how they achieve that.
- 2.11 Counter terrorism and public safety
- 2.12 Licensing committees may wish to give due consideration to appropriate counterterrorism measures and advice when considering licence applications, for example at high profile or large premises or events or where there are factors which may increase attractiveness to attacks, and in particular when in receipt of relevant advice from police counter terrorist staff.
- 2.13 It is particularly relevant to consider conditions around health care provision in this regard. Any additional licensing conditions should be appropriate and proportionate to the venue as noted in para 1.16. We recommend that all licensing authorities pay particular attention to terrorist threats when making specific licence conditions in relation to security or health care provision at appropriate premises and events applying for a licence.
- 2.14 As noted in para 2.8 licensing authorities can recommend conditions where appropriate to do so, such as ensuring appropriate access for emergency services and vehicles such as ambulances; good communication with local authorities and emergency services, for example communications networks with the police and signing up for local incident alerts (see paragraph 2.4); and ensuring the presence of sufficient trained first aiders on the premises and appropriate first aid kits.
- 2.15 However, in some instances, licensing authorities may wish to tailor their approach, depending on the particular nature of the application, and go further when addressing concerns around possible terrorist or other such risks. We have provided an Annex to help inform and support licensing authorities in their decision making with regards to additional licence conditions for premises and events, that may benefit from further security planning and health care provision in the aftermath of a particular type of incident.

Ensuring safe departure of those using the premises

2.16 Licence holders should make provision to ensure that premises users safely leave their premises. Measures that may assist include:
- Providing information on the premises of local taxi companies who can provide safe transportation home; and
- Ensuring adequate lighting outside the premises, particularly on paths leading to and from the premises and in car parks.

Maintenance and repair

2.17 Where there is a requirement in other legislation for premises open to the public or for employers to possess certificates attesting to the safety or satisfactory nature of certain equipment or fixtures on the premises, it would be inappropriate for a licensing condition to require possession of such a certificate. However, it would be permissible to require as a condition of a licence or certificate, if appropriate, checks on this equipment to be conducted at specified intervals and for evidence of these checks to be retained by the premises licence holder or club provided this does not duplicate or gold-plate a requirement in other legislation. Similarly, it would be permissible for licensing authorities, if they receive relevant representations from responsible authorities or any other persons, to attach conditions which require equipment of particular standards to be maintained on the premises. Responsible authorities – such as health and safety authorities – should therefore make their expectations clear in this respect to enable prospective licence holders or clubs to prepare effective operating schedules and club operating schedules.

Safe capacities

- 2.18 "Safe capacities" should only be imposed where appropriate for the promotion of public safety or the prevention of disorder on the relevant premises. For example, if a capacity has been imposed through other legislation, it would be inappropriate to reproduce it in a premises licence. Indeed, it would also be wrong to lay down conditions which conflict with other legal requirements. However, if no safe capacity has been imposed through other legislation, a responsible authority may consider it appropriate for a new capacity to be attached to the premises which would apply at any material time when the licensable activities are taking place and make representations to that effect. For example, in certain circumstances, capacity limits may be appropriate in preventing disorder, as overcrowded venues can increase the risks of crowds becoming frustrated and hostile.
- 2.19 The permitted capacity is a limit on the number of persons who may be on the premises at any time, following a recommendation by the relevant fire and rescue authority under the Regulatory Reform (Fire Safety) Order 2005. For any application for a premises licence or club premises certificate for premises without an existing permitted capacity where the applicant wishes to take advantage of the special provisions set out in section 177 of the 2003 Act¹, the applicant should conduct their own risk assessment as to the appropriate capacity of the premises. They should send their recommendation to the fire and rescue authority which will consider it and decide what the "permitted capacity" of those premises should be.
- 2.20 Public safety may include the safety of performers appearing at any premises, but does not extend to the prevention of injury from participation in a boxing or wrestling

entertainment.

Public nuisance

- 2.21 The 2003 Act enables licensing authorities and responsible authorities, through representations, to consider what constitutes public nuisance and what is appropriate to prevent it in terms of conditions attached to specific premises licences and club premises certificates. It is therefore important that in considering the promotion of this licensing objective, licensing authorities and responsible authorities focus on the effect of the licensable activities at the specific premises on persons living and working (including those carrying on business) in the area around the premises which may be disproportionate and unreasonable. The issues will mainly concern noise nuisance, light pollution, noxious smells and litter.
- 2.22 Public nuisance is given a statutory meaning in many pieces of legislation. It is however not narrowly defined in the 2003 Act and retains its broad common law meaning. It may include in appropriate circumstances the reduction of the living and working amenity and environment of other persons living and working in the area of the licensed premises. Public nuisance may also arise as a result of the adverse effects of artificial light, dust, odour and insects or where its effect is prejudicial to health.
- 2.23 Conditions relating to noise nuisance will usually concern steps appropriate to control the levels of noise emanating from premises. This might be achieved by a simple measure such as ensuring that doors and windows are kept closed after a particular time, or persons are not permitted in garden areas of the premises after a certain time. More sophisticated measures like the installation of acoustic curtains or rubber speaker mounts to mitigate sound escape from the premises may be appropriate. However, conditions in relation to live or recorded music may not be enforceable in circumstances where the entertainment activity itself is not licensable (see chapter 16). Any conditions appropriate to promote the prevention of public nuisance should be tailored to the type, nature and characteristics of the specific premises and its licensable activities. Licensing authorities should avoid inappropriate or disproportionate measures that could deter events that are valuable to the community, such as live music. Noise limiters, for example, are expensive to purchase and install and are likely to be a considerable burden for smaller venues.
- 2.24 As with all conditions, those relating to noise nuisance may not be appropriate in certain circumstances where provisions in other legislation adequately protect those living in the area of the premises. But as stated earlier in this Guidance, the approach of licensing authorities and responsible authorities should be one of prevention and when their powers are engaged, licensing authorities should be aware of the fact that other legislation may not adequately cover concerns raised in relevant representations and additional conditions may be appropriate.
- 2.25 Where applications have given rise to representations, any appropriate conditions should normally focus on the most sensitive periods. For example, the most sensitive period for people being disturbed by unreasonably loud music is at night and into the early morning when residents in adjacent properties may be attempting to go to sleep or are sleeping. This is why there is still a need for a licence for performances of live music between 11 pm and 8 am. In certain circumstances, conditions relating to noise

emanating from the premises may also be appropriate to address any disturbance anticipated as customers enter and leave.

- 2.26 Measures to control light pollution will also require careful thought. Bright lighting outside premises which is considered appropriate to prevent crime and disorder may itself give rise to light pollution for some neighbours. Applicants, licensing authorities and responsible authorities will need to balance these issues.
- 2.27 Beyond the immediate area surrounding the premises, these are matters for the personal responsibility of individuals under the law. An individual who engages in antisocial behaviour is accountable in their own right. However, it would be perfectly reasonable for a licensing authority to impose a condition, following relevant representations, that requires the licence holder or club to place signs at the exits from the building encouraging patrons to be quiet until they leave the area, or that, if they wish to smoke, to do so at designated places on the premises instead of outside, and to respect the rights of people living nearby to a peaceful night.

Protection of children from harm

- 2.28 The protection of children from harm includes the protection of children from moral, psychological and physical harm. This includes not only protecting children from the harms associated directly with alcohol consumption but also wider harms such as exposure to strong language and sexual expletives (for example, in the context of exposure to certain films or adult entertainment). Licensing authorities must also consider the need to protect children from sexual exploitation when undertaking licensing functions.
- 2.29 The Government believes that it is completely unacceptable to sell alcohol to children. Conditions relating to the access of children where alcohol is sold and which are appropriate to protect them from harm should be carefully considered. Moreover, conditions restricting the access of children to premises should be strongly considered in circumstances where:
 - adult entertainment is provided;
 - a member or members of the current management have been convicted for serving alcohol to minors or with a reputation for allowing underage drinking (other than in the context of the exemption in the 2003 Act relating to 16 and 17 year olds consuming beer, wine and cider when accompanied by an adult during a table meal);
 - it is known that unaccompanied children have been allowed access;
 - there is a known association with drug taking or dealing; or
 - in some cases, the premises are used exclusively or primarily for the sale of alcohol for consumption on the premises.
- 2.30 It is also possible that activities, such as adult entertainment, may take place at certain times on premises but not at other times. For example, premises may operate as a café bar during the day providing meals for families but also provide entertainment with a sexual content after 8.00pm. It is not possible to give an exhaustive list of what amounts to entertainment or services of an adult or sexual nature. Applicants, responsible authorities and licensing authorities will need to consider this point carefully. This would broadly include topless bar staff, striptease, lap-, table- or pole-dancing, performances involving feigned violence or horrific incidents, feigned or actual sexual acts or fetishism,

or entertainment involving strong and offensive language.

- 2.31 Applicants must be clear in their operating schedules about the activities and times at which the events would take place to help determine when it is not appropriate for children to enter the premises. Consideration should also be given to the proximity of premises to schools and youth clubs so that applicants take appropriate steps to ensure that advertising relating to their premises, or relating to events at their premises, is not displayed at a time when children are likely to be near the premises.
- 2.32 Licensing authorities and responsible authorities should expect applicants, when preparing an operating schedule or club operating schedule, to set out the steps to be taken to protect children from harm when on the premises.
- 2.33 Conditions, where they are appropriate, should reflect the licensable activities taking place on the premises. In addition to the mandatory condition regarding age verification, other conditions relating to the protection of children from harm can include:
 - restrictions on the hours when children may be present;
 - restrictions or exclusions on the presence of children under certain ages when particular specified activities are taking place;
 - restrictions on the parts of the premises to which children may have access;
 - age restrictions (below 18);
 - restrictions or exclusions when certain activities are taking place;
 - requirements for an accompanying adult (including for example, a combination of requirements which provide that children under a particular age must be accompanied by an adult); and
 - full exclusion of people under 18 from the premises when any licensable activities are taking place.
- 2.34 Please see also Chapter 10 for details about the Licensing Act 2003 (Mandatory Licensing Conditions) Order 2010.
- 2.35 Licensing authorities should give considerable weight to representations about child protection matters. In addition to the responsible authority whose functions relate directly to child protection, the Director of Public Health may also have access to relevant evidence to inform such representations. These representations may include, amongst other things, the use of health data about the harms that alcohol can cause to underage drinkers. Where a responsible authority, or other person, presents evidence to the licensing authority linking specific premises with harms to children (such as ambulance data or emergency department attendances by persons under 18 years old with alcohol- related illnesses or injuries) this evidence should be considered, and the licensing authority should also consider what action is appropriate to ensure this licensing objective is effectively enforced. In relation to applications for the grant of a licence in areas where evidence is presented on high levels of alcohol-related harms in persons aged under 18, it is recommended that the licensing authority considers what conditions may be appropriate to ensure that this objective is promoted effectively.
- 2.36 The 2003 Act provides that, where a premises licence or club premises certificate authorises the exhibition of a film, it must include a condition requiring the admission of children to films to be restricted in accordance with recommendations given either by a body designated under section 4 of the Video Recordings Act 1984 specified in the

licence (the British Board of Film Classification is currently the only body which has been so designated) or by the licensing authority itself. Further details are given in Chapter 10.

2.37 Theatres may present a range of diverse activities and entertainment including, for example, variety shows incorporating adult entertainment. It is appropriate in these cases for a licensing authority to consider restricting the admission of children in such circumstances. Entertainments may also be presented at theatres specifically for children. It will be appropriate to consider whether a condition should be attached to a premises licence or club premises certificate which requires the presence of a sufficient number of adult staff on the premises to ensure the wellbeing of the children during any emergency.

Offences relating to the sale and supply of alcohol to children

2.38 Licensing authorities are expected to maintain close contact with the police, young offenders' teams and trading standards officers (who can carry out test purchases under section 154 of the 2003 Act) about the extent of unlawful sales and consumption of alcohol by minors and to be involved in the development of any strategies to control or prevent these unlawful activities and to pursue prosecutions. Licensing authorities, alongside the police, are prosecuting authorities for the purposes of these offences, except for the offences under section 147A (persistently selling alcohol to children). Where, as a matter of policy, warnings are given to retailers prior to any decision to prosecute in respect of an offence, it is important that each of the enforcement arms should be aware of the warnings each of them has given.

Section	Offence	Prosecuting Authority
Section 145	Unaccompanied children prohibited from	Police and/or Licensing Authority
	certain premises	
Section 146	Sale of alcohol to children	Police, Licensing Authority and/or
		Local Weights and Measures
		Authority
Section 147	Allowing the sale of alcohol to children	Police, Licensing Authority and/or
		Local Weights and Measures
		Authority
Section 147A	Persistently selling alcohol to children	Police and/or Local Weights and
		Measures Authority
Section 149	Purchase of alcohol by or on behalf of	Police and/or Licensing Authority
	children	
Section 150	Consumption of alcohol by children	Police and/or Licensing Authority
Section 151	Delivering alcohol to children	Police and/or Licensing Authority
Section 152	Sending a child to obtain alcohol	Police and/or Licensing Authority
Section 153	Prohibition of unsupervised sales by	Police and/or Licensing Authority
	children	

Table of relevant offences under the 2003 Act

3. Licensable activities

Summary

- 3.1 A premises licence authorises the use of any premises (see Chapter 5) for licensable activities. Licensable activities are defined in section 1 of the 2003 Act, and a fuller description of certain activities is set out in Schedules 1 and 2 to the 2003 Act.
- 3.2 The licensable activities are:
 - the sale by retail of alcohol;
 - the supply of alcohol by or on behalf of a club to, or to the order of, a member of the club;
 - the provision of regulated entertainment; and
 - the provision of late night refreshment.

Wholesale of alcohol

- 3.3 The sale of alcohol to the general public is licensable under the 2003 Act in accordance with the definition of "sale by retail" in section 192 of the 2003 Act. This section makes it clear that, to be excluded from the meaning of "sale by retail", a sale must be:
 - made from premises owned by the person making the sale, or occupied under a lease with security of tenure; and
 - for consumption off the premises.
- 3.4 In addition, to be excluded, the sales must be sales which are made to:
 - a trader for the purpose of his trade;
 - to a club for the purposes of that club;
 - to a holder of a premises licence or a personal licence for the purpose of making sales under a premises licence; or
 - a premises user who has given a temporary event notice, for the purpose of making sales authorised by that notice.
- 3.5 If an employee were buying alcohol as an "agent" for their employer and for the purposes of their employer's trade (i.e. selling alcohol), this could be treated as a sale to a trader. If, however, an employee were buying for the employee's own consumption, this would be a retail sale, and would require a licence.
- 3.6 The same considerations apply in the case of caterers who supply alcohol to their customers. Where a caterer purchases alcohol and then sells this alcohol to its customer, an authorisation will be required at the location where the retail sale of the alcohol is made (likely to be the caterer's own premises). If the customer was proposing to sell the alcohol under an authorisation, it is the customer who would need an authorisation under the 2003 Act. In this case, the exemption under the 2003 Act may apply to the sale made by the caterer.
- 3.7 From 1 April 2017, businesses which sell alcohol (for example, retailers of alcohol and trade buyers) will need to ensure that the UK wholesalers that they buy alcohol from have been approved by HMRC under the Alcohol Wholesaler Registration Scheme (AWRS). They will need to check their wholesalers Unique Registration Number (URN)

against the HMRC online database. This is an ongoing obligation and if a business is found to have bought alcohol from an unapproved wholesaler, they may be liable to a penalty or could even face criminal prosecution and their alcohol stock may be seized. Any trader who buys alcohol from a wholesaler for onward sale to the general public (known as a 'trade buyer') does not need to register unless they sell alcohol to other businesses. Examples of trade buyers would be pubs, clubs, restaurants, cafes, retailers and hotels. However, they will need to check that the wholesaler they purchase alcohol from is registered with HMRC. Further information may be found at: https://www.gov.uk/guidance/the-alcohol-wholesaler-registration-scheme-awrs.

Mobile, remote, internet and other delivery sales

- 3.8 The sale by retail of alcohol is a licensable activity and may only be carried out in accordance with an authorisation under the 2003 Act. Therefore, a person cannot sell alcohol from a vehicle or moveable structure at a series of different locations (e.g. house to house), unless there is a premises licence in respect of the vehicle or moveable structure at each location at which a sale of alcohol is made in, on or from it.
- 3.9 The place where the order for alcohol, or payment for it, takes place may not be the same as the place where the alcohol is appropriated to the contract (i.e. the place where it is identified and specifically set apart for delivery to the purchaser). This position can arise when sales are made online, by telephone, or mail order. Section 190 of the 2003 Act provides that the sale of alcohol is to be treated as taking place where the alcohol is appropriated to the contract. It will be the premises at this location which need to be licensed; for example, a call centre receiving orders for alcohol would not need a licence but the warehouse where the alcohol is stored and specifically selected for, and despatched to, the purchaser would need to be licensed. These licensed premises will, as such, be subject to conditions including the times of day during which alcohol may be sold. The premises licence will also be subject to the mandatory licence conditions.
- 3.10 Persons who run premises providing 'alcohol delivery services' should notify the relevant licensing authority that they are operating such a service in their operating schedule. This ensures that the licensing authority can properly consider what conditions are appropriate. Premises with an existing premises licence, which choose to operate such a service in addition to their existing licensable activities, may consider contacting their licensing authority for its view on whether this form of alcohol sale is already permitted or whether an application to vary the licence will be required.

Regulated entertainment

3.11 Schedule 1 to the 2003 Act sets out what activities are to be treated as the provision of regulated entertainment and those that are not and are therefore exempt from the regulated entertainment aspects of the licensing regime, including incidental music – (see paragraphs 16.1 to 16.3 below). Chapter 16 of this Guidance document sets out the types of entertainment regulated by the 2003 Act.

Late night refreshment

- 3.12 Schedule 2 to the 2003 Act provides a definition of what constitutes the provision of late night refreshment. It involves the supply of 'hot food or hot drink' between the hours of 23.00 and 05.00 to the public for consumption on or off the premises. It includes the supply of hot food or hot drink between those hours on premises to which the public has access. Under Schedule 2, food or drink is considered to be 'hot' if, before it is supplied, it has been heated on the premises or elsewhere for the purpose of enabling it to be consumed at a temperature above the ambient air temperature and at the time of supply it is above that temperature; or after it is supplied, may be heated on the premises for the purpose of enabling it to be consumed at a temperature.
- 3.13 Shops, stores and supermarkets selling only cold food and cold drink, whether it is immediately consumable or not, from 23.00 are not licensable as providing late night refreshment. The 2003 Act affects premises such as night cafés and takeaway food outlets where people may gather to purchase hot food or hot drink at any time from 23.00 and until 05.00. In this case, supply takes place when the hot food or hot drink is given to the customer and not when payment is made. For example, supply takes place when a table meal is served in a restaurant or when a takeaway is handed to a customer over the counter.
- 3.14 Some premises provide hot food or hot drink between 23.00 and 05.00 by means of vending machines. The supply of hot drink by a vending machine is not a licensable activity and is exempt under the 2003 Act provided the public have access to and can operate the machine without any involvement of the staff.
- 3.15 However, this exemption does not apply to hot food. Premises supplying hot food for a charge by vending machine are licensable if the food has been heated on the premises, even though no staff on the premises may have been involved in the transaction.
- 3.16 It is not expected that the provision of late night refreshment as a secondary activity in licensed premises open for other purposes such as public houses, cinemas or nightclubs or casinos should give rise to a need for significant additional conditions.
- 3.17 The supply of hot drink which consists of or contains alcohol is exempt under the 2003 Act as late night refreshment because it is licensed by the provisions relating to the sale or supply of alcohol.
- 3.18 The supply of hot food or hot drink free of charge is not a licensable activity. However, where any charge is made for either admission to the premises or for some other item in order to obtain the hot food or hot drink, this will not be regarded as "free of charge". Supplies by a registered charity or anyone authorised by a registered charity are also exempt.
- 3.19 Supplies made on moving vehicles (for example boats, trains or coaches) are also exempt. However supplies made from a vehicle which is permanently or temporarily parked, such as from a mobile takeaway van, are not exempt (see section 3.34 below for more detail on provisions for 'Vessels, vehicles and moveable structures').
- 3.20 Supplies of hot food or hot drink from 23.00 are exempt from the provisions of the 2003 Act if there is no admission to the public to the premises involved and they are supplies to:

- a member of a recognised club supplied by the club;
- persons staying overnight in a hotel, guest house, lodging house, hostel, a caravan or camping site or any other premises whose main purpose is providing overnight accommodation;
- an employee of a particular employer (for example in a staff canteen);
- a person who is engaged in a particular profession or who follows a particular vocation (for example, a tradesman carrying out work at particular premises);
- a guest of any of the above.

Late night refreshment exemptions based on designated locations, premises types and times

- 3.21 The provision of late night refreshment is regulated primarily because it is often linked to alcohol-fuelled crime and disorder in the night-time economy, such as at fast-food takeaways where late-night drinkers congregate. However, these safeguards may not be needed everywhere or for every type of late night refreshment business. For example, some late-night cafés serving hot drinks after 23.00 may be located nowhere near pubs and nightclubs or areas associated with alcohol-related crime and disorder.
- 3.22 Paragraph 2A of Schedule 2 to the 2003 Act (as inserted by the Deregulation Act 2015) gives licensing authorities powers to exempt premises, in certain circumstances, from the requirement to have a licence to provide late night refreshment. Decisions to exempt supplies of late night refreshment are best made with local knowledge. The powers therefore allow licensing authorities to choose to apply an exemption specifically where they think it will be helpful to businesses and where there are no problems with antisocial behaviour, disorder associated with the night time economy, or illegal working in licensed premises. As well as freeing up the businesses in question from unnecessary costs, this can also provide greater flexibility for licensing authorities to target their resources more effectively.
- 3.23 The powers allow a relevant licensing authority to exempt the supply of late night refreshment if it takes place:
 - on or from premises which are wholly situated in a designated area;
 - on or from premises which are of a designated description; or
 - during a designated period (beginning no earlier than 23.00 and ending no later than 05.00.
- 3.24 When choosing to designate a particular area as exempt, the relevant licensing authority must define the location, which can be of any size.
- 3.25 When choosing to designate particular categories of premises as exempt, a licensing authority can only exempt types of premises set out in the regulations. These are:
 - Motorway service areas;
 - petrol stations;
 - local authority premises (except domestic premises) unless there is an event taking place at which more than 500 people are present;
 - schools (except domestic premises) unless there is an event taking place at which more than 500 people are present;

- hospitals (except domestic premises);
- community premises (church, chapel, village, parish or community hall or other similar building) unless there is an event taking place at which more than 500 people are present;
- licensed premises authorised to sell by retail alcohol for consumption on the premises between the hours of 23.00 and 05.00.
- 3.26 When choosing to exempt the provision of late night refreshment at particular times, the relevant licensing authority must determine the times between 23.00 and 05.00 when the exemption applies. The exemption and any subsequent change to the time will apply to the whole licensing authority area.
- 3.27 A relevant licensing authority may use more than one type of exemption at the same time, for example by changing the times across the licensing authority area during which licensing requirements will apply and also exempting premises by type across the whole licensing authority area. However, it cannot use different forms of exemption in conjunction with one another for example, it would not be permitted to change the times in one geographic area only.
- 3.28 Where a premises is situated in the areas of two or more licensing authorities, any of those authorities may be the relevant licensing authority and it would be advisable for an authority wishing to apply an exemption to discuss it with the other authority concerned. This might apply, for example, where an area or premises type exemption is being applied and the licensing authority is aware that a particular premises such as a motorway service area sits across the boundary of two or more licensing authority areas.
- 3.29 Licensing authorities should consider deregulation where possible. However, they do not have to use the exemptions and can continue to require all late night refreshment providers to be licensed where this is appropriate for the promotion of the licensing objectives.
- 3.30 Existing late night refreshment licences for premises that become exempt from regulation will remain extant unless the holder chooses to surrender it to the licensing authority, but there will be no requirement on the licence holder to pay annual fees and any conditions on the licence will cease to apply for as long as the exemption is in place. In cases where an exemption in relation to late night refreshment provision is applied, other licensing is unaffected. For example if a premises is licensed to sell alcohol and is exempt from requiring a late night refreshment licence, their licence in respect of the sale of alcohol is unaffected. Where a premises benefits from an exemption applied by the licensing authority, any existing conditions on a licence relating solely to the provision of late night refreshment will have no effect during the period of the exemption.
- 3.31 When deciding which exemption to use, if any, the relevant licensing authority should always first consider what the risks are in terms of the promotion of the licensing objectives, including the prevention of illegal working in those premises. The decision to make an exemption is a licensing function that licensing authorities should include within their statement of licensing policy. It would then therefore be subject to the statutory consultation process with other responsible authorities and relevant parties set out in section 5 of the 2003 Act. However, it is for the licensing authority to decide on the detail

and extent of the consultation beyond the statutory minimum; for example, in areas where there are concerns about illegal working in licensed premises the licensing authority should consult Home Office Immigration Enforcement. The licensing authority may decide to only consult on the proposed exemption or, alternatively, it may form part of a wider review of other matters within its statement of licensing policy.

- 3.32 When applying any of the exemptions the relevant licensing authority must publicise the changes and should decide on the most appropriate way to do this, in addition to updating its statement of licensing policy as soon as is practical. There is no requirement for licensing authorities to tell premises individually, however they should publicise the exemption in a way that ensures that those who are likely to be affected may benefit from it. If any fees are paid prior to an exemption coming into effect, licensing authorities should consider whether a refund or partial refund is appropriate. It is for each individual licensing authority to develop its own refund policy and ensure that it is communicated appropriately to all licence holders that are likely to be affected by an exemption.
- 3.33 Licensing authorities can review the exemptions at any time, to change the times, locations or types. However, unlike many other types of licensing decision, the late night refreshment exemptions are not made on a case by case basis and there is no recourse to bring an individual premises back into the licensing regime if there is a problem with that particular premises. In such cases the licensing authority would have to take a decision about the entire exemption and apply it across the whole area. Alternatively, depending on the scale of the problem, other powers could be used such as closure powers under the Anti-social Behaviour, Crime and Policing Act 2014. Environmental health legislation around noise nuisance may also offer a solution.

Late night refreshment from vessels, vehicles and moveable structures

3.34 Under section 189 of the 2003 Act, a vehicle which is not permanently situated in the same place and is or is proposed to be used for one or more licensable activities while parked at a particular place, is to be treated as if it were premises situated at that place. Therefore, a mobile provider of late night refreshment, such as a kebab van, could be treated as exempt if it supplied hot food to the public late at night in an area which had been designated as exempt. If the mobile van drove to and began operating in a non-exempt area, a licence to carry on this activity would be required. Should the licensing authority introduce an exemption, and subsequently wish to revoke it if problems arise, it has the power to do so. Areas which are likely to be considered for exemption by licensing authorities (for example, an area outside a town centre) are unlikely to be areas in which mobile kebab vans would frequently operate. As such, mobile vehicles selling late night refreshment are likely to still require licences in the areas in which they are more commonly found.

Unauthorised activities

3.35 It is a criminal offence under section 136 of the 2003 Act to carry on any of the licensable activities listed at paragraph 3.2 above other than in accordance with a licence or other authorisation under the 2003 Act. The fine for this offence is unlimited. Police and local authorities have powers to take action in relation to premises carrying on unauthorised activities.

4. Personal licences

4.1 This chapter provides advice about the framework for personal licences. It also contains guidance for decision-making on applications by those managing community premises (church and village halls etc.) to remove the usual mandatory conditions that relate to personal licences and the requirement for a designated premises supervisor (DPS). The Deregulation Act 2015 removed the requirement to renew a personal licence with effect from 1 April 2015.

Requirements for a personal licence

- 4.2 The sale and supply of alcohol, because of its impact on the wider community and on crime and anti-social behaviour, carries with it greater responsibility than the provision of regulated entertainment and late night refreshment. This is why sales of alcohol may not be made under a premises licence unless there is a DPS in respect of the premises (who must hold a personal licence); and every sale must be made or authorised by a personal licence holder. The exception is only for those community premises which have successfully applied to remove the DPS requirement (see paragraph 4.84 below).
- 4.3 Any premises at which alcohol is sold or supplied where the requirement for a personal licence holder does apply may employ one or more such licence holders. For example, there may be one owner or senior manager and several junior managers holding a personal licence. However, the requirement that every sale of alcohol must at least be authorised by a personal licence holder does not mean that the licence holder has to be present on the premises or oversee each sale; it is sufficient that such sales are authorised. It should be noted that there is no requirement to have a DPS in relation to a Temporary Event Notice (TEN) or club premises certificate, and sales or supplies of alcohol authorised by a TEN or club premises certificate do not need to be authorised by a personal licence holder.

Who can apply?

- 4.4 In the case of an application for a personal licence under Part 6 of the 2003 Act, the requirements are that:
 - the applicant must be aged 18 or over;
 - the applicant, if subject to immigration control, must have permission to work in a licensable activity (see paragraph 4.8);
 - the applicant possesses a licensing qualification accredited by the Secretary of State (or one which is certified as if it is such a qualification or is considered equivalent) or is a person as prescribed in the Licensing Act 2003 (Personal licences) Regulations 2005²).
 - the applicant must not have forfeited a personal licence within five years of their application;
 - · the applicant has paid the appropriate fee to the licensing authority; and

² Currently persons prescribed in regulations are: a member of the company of the Master, Wardens, Freemen and Commonalty of the Mystery of the Vintners of the City of London; a person operating under a licence granted by the University of Cambridge; or a person operating premises under a licence granted by the Board of the Green Cloth.

- in a case in which the applicant has an unspent conviction for a relevant offence or a foreign offence, the police have not objected to the grant of the application on crime prevention grounds or the licensing authority has considered their objection but determined that the grant of the application will not undermine the crime prevention objective.
- 4.5 Any individual may apply for a personal licence whether or not they are currently employed or have business interests associated with the use of the licence. The issues which arise when the holder of a personal licence becomes associated with particular licensed premises and is specified as the DPS for those premises are dealt with at paragraphs 4.84 to 4.97 below. Licensing authorities may not therefore take these matters into account when considering an application for a personal licence.
- 4.6 Applicants for personal licences who are ordinarily resident in a licensing authority's area are required to make the application to that licensing authority. An applicant who is not ordinarily resident in a licensing authority's area (which may include persons living outside England and Wales), may apply for the grant of a personal licence to any licensing authority in England and Wales.
- 4.7 For applications made after 6 April 2017, applicants who are subject to UK immigration control must be entitled to work in a licensable activity. Section 192A of the Licensing Act 2003 defines 'entitlement to work' for the purposes of the Act

Entitlement to work in the UK

- 4.8 Individuals applying for a personal licence must be entitled to work in the UK. The Immigration Act 2016 amended the Licensing Act 2003 with effect from 6 April 2017 so that an application made on or after that date by someone who is not entitled to work in the UK must be rejected. The purpose of this provision is to help prevent illegal working in the UK.
- 4.9 Licensing Authorities must be satisfied that an applicant has the right to work in the UK. Applicants are required to submit one of the documents listed at Annex A of the Home Office's Employer right to work checks supporting guidance published on gov.uk at: <u>Right to work checks: an employer's guide - GOV.UK (www.gov.uk)</u> to show that they have permission to be in the UK and are permitted to undertake work in a licensable activity. This also applies to individuals who apply for premises licences.
- 4.10 As an alternative to using one of the documents listed, applicants may choose to demonstrate their right to work by allowing the licensing authority to carry out a check with the Home Office online right to work checking service. Where a right to work check has been conducted using the online service, the information is provided in real-time directly from Home Office systems and there is no requirement to carry out a manual document-based check. Applicants should be invited to provide their share code in their application which, along with the applicant's date of birth, will allow the licensing authority to_check their immigration status via the online service available on gov.uk at: <u>View a job applicant's right to work details GOV.UK (www.gov.uk)</u>
- 4.11 Licensing authorities may encourage use of the online checking service and may support applicants in doing so (e.g. by providing access to hardware and the internet). However, licensing authorities are not permitted to mandate online checks, except for those applicants who have been provided with digital evidence of their immigration

status only (known as an eVisa).

- 4.12 For applications made on or after 6 April 2017, where an applicant's immigration permission to live and work in the UK is time-limited, a personal licence may be issued. However, it will become invalid when the immigration permission expires. In the event that the Home Office cuts short or ends a person's immigration permission (referred to as curtailment or revocation), any licence issued in respect of an application made on or after 6 April 2017 will automatically lapse (see paragraph 4.17 below).
- 4.13 A person is disqualified from applying for a personal licence or a premises licence by reason of their immigration status if:
 - The person requires leave to enter or remain in the UK and has not been granted it; or
 - The person's leave to enter or remain in the UK:
 - is invalid,
 - has ceased to have effect (whether by reason of curtailment, revocation, cancellation, passage of time, or otherwise), or
 - is subject to a condition preventing the person from doing work of that kind.
- 4.14 The requirement to demonstrate immigration status is not retrospective. This means that licensing authorities do not need to check the immigration status of those people who already hold a licence which was issued before 6 April 2017.
- 4.15 Applicants may provide photocopies or scanned copies of the documents demonstrating their right to work. These do not need to be endorsed as a copy of the original. Applicants may prefer to provide original copies, but licensing authorities cannot mandate this. The licensing authority must be satisfied that the applicant is entitled to work in the UK. The licensing authority should establish whether or not an applicant has a lawful immigration status in the UK or is prohibited from working because they are in the UK unlawfully or is subject to a condition that prevents them from holding a licence.
- 4.16 To ensure that licensing authorities do not discriminate against anyone, all licence applicants should be treated in the same way during the licence application process. This will ensure a fair, transparent and consistent application process. Assumptions should not be made about a person's right to work in the UK or their immigration status on the basis of their nationality, ethnic origin, accent, the colour of their skin, or the length of time they have been resident in the UK.
- 4.17 If an applicant has restrictions on the length of time they may work in the UK, a premises licence or personal licence may still be issued. However, the licence will cease to have effect when the right to work lapses. Migrants who are subject to UK immigration control may be granted permission to enter or remain in the UK, with a condition permitting employment, on a time-limited basis or on an indefinite basis. When the person's stay is time limited, this will be shown in their immigration documentation or on their profile available from the online checking service where applicable. It is possible for a migrant to apply to extend their permission to remain. If they do so before their previous status expires, they continue to have any right to work that they previously had while their application and any associated administrative review or appeal is outstanding. In such cases, a person's status may be confirmed by the licensing

authority contacting the Home Office Status, Verification, Evidence and Checking (SVEC) Unit at ICESSVECworkflow@homeoffice.gov.uk.

- 4.18 In most cases the licensing authority should be able to make an assessment that the applicant is not disqualified from applying for a premises or personal licence based on any information provided with the application. This will include all cases where the applicant is a British citizen. It is only necessary to contact the Home Office SVEC Unit in the following circumstances to verify that someone has the right to hold a premises or personal licence:
 - licensing authorities are presented with a document (non-digital CoA or an acknowledgement letter or email) confirming receipt of an application to the EU Settlement Scheme (EUSS) on or before 30 June 2021; or
 - licensing authorities are presented with a non-digital CoA confirming receipt of an application to the EUSS on or after 1 July 2021; or
 - licensing authorities have checked a digital CoA, using the online service, confirming receipt of an application to the EUSS on or after 1 July 2021, and been directed to SVEC; or
 - the applicant is unable to provide evidence of their right to work due to an outstanding application for permission to remain in the UK which was made before their previous immigration leave expired, or has an outstanding appeal or administrative review against a Home Office decision that grants them a right to work; or
 - licensing authorities consider that they have not been provided with any acceptable documents, but the applicant presents other information indicating they are a long-term resident of the UK who arrived in the UK before 1988.

In these five circumstances, SVEC will confirm the individual's immigration status. However, licensing authorities will still have to determine whether the applicant should be granted a licence.

- 4.19 Advice and assistance on this process may be obtained from Home Office Local Partnership Managers, or by email <u>ISDLPMSsupportTeam@homeoffice.gov.uk.</u> In most cases, a Local Partnership Manager or local Immigration, Compliance and Enforcement (ICE) team will be the first point of contact for licensing authorities.
- 4.20 A premises or personal licence issued in respect of an application made on or after 6 April 2017 will lapse if the holder's permission to live or work in the UK comes to an end. This could be because their permission to be in the UK has expired or because the Home Office has brought it to an end (for example, the Home Office has curtailed their permission to live and work in the UK). However, licensing authorities are not required to carry out ongoing immigration status checks to see whether a licence holder's permission to be in the UK has expired. Furthermore, licensing authorities are not required withdraw or revoke the licence if a licence holder's immigration permission expires. The migrant will be aware when their time-limited permission has ended, and the Home Office will inform them if their permission to be in the UK is curtailed. If the individual is subsequently granted permission to stay which holds a right to work in the UK and wishes to hold a personal licence once again, they must make an application for a new personal licence.

Entitlement to work in the UK for EEA citizens since 1 July 2021

- 4.21 References to 'EEA citizens' in this guidance means EU, EEA and Swiss citizens unless stated otherwise. The UK has left the European Union (EU) and the Immigration and Social Security Co-ordination (EU Withdrawal) Act 2020 ended free movement on 31 December 2020. There followed a grace period of six-months, during which relevant aspects of free movement were saved to allow eligible EEA citizens and their family members resident in the UK by 31 December 2020 to apply to the EU Settlement Scheme (EUSS). This period ended on 30 June 2021.
- 4.22 Since 1 July 2021, EEA citizens and their family members require immigration status in the UK. They can no longer rely on an EEA passport or national identity card, which only confirms their nationality, to prove their right to work. They are required to provide evidence of lawful immigration status in the UK, in the same way as other foreign nationals.

Irish citizens

4.23 Irish citizens continue to have unrestricted access to work in the UK, they can prove their right to work using their Irish passport or Irish passport card. Alternatively, they can use their Irish birth or adoption certificate, together with an official document giving their permanent National Insurance number and their name. The latter must be issued by a Government Agency or a previous employer.

EEA Citizens granted status under the EU Settlement Scheme (EUSS)

- 4.24 Since 1 July 2021, the majority of EEA citizens granted status under the EUSS prove their right to work via an eVisa.
- 4.25 If an EEA citizen has been granted 'settled status' by the Home Office they will have a continuous right to work, in the same way as someone with indefinite leave to enter/remain status. online right to work service.
- 4.26 If an EEA citizen has been granted 'pre-settled' status by the Home Office, they will have a time-limited right to work and a premises licence or personal licence may still be issued, but the licence will cease to have effect when the right to work lapses.
- 4.27 Individuals will provide licensing authorities with a share code and their date of birth during the licence application process. This will enable licencing authorities to check the individual's immigration status via the online service available on gov.uk: <u>View a job</u> applicant's right to work details GOV.UK (www.gov.uk).

Outstanding UK EUSS applications

4.28 EEA citizens, and their family members, who have made a valid application to the EUSS can continue to live their life in the UK and maintain a right to work until their application is finally determined. This includes pending the outcome of any appeal against a

decision to refuse status. Licensing authorities must provide applicants with every opportunity to prove their right to work and should not discriminate against those applicants with an outstanding, valid application.

EU Settlement Scheme status granted by a Crown Dependency

- 4.29 The Crown Dependencies (the Bailiwick of Jersey, the Bailiwick of Guernsey, and the Isle of Man) each operate their own EUSS, for those eligible to apply. The UK and the Crown Dependencies recognise status granted under each other's Scheme. An individual granted settled or pre-settled status by a Crown Dependency will be considered to have settled or pre-settled status in the UK. The Isle of Man and Guernsey issue a letter to those granted EUSS status. Jersey issues a letter and operate an immigration status checker service for individuals to obtain confirmation of their status at any point.
- 4.30 Since 1 July 2021, when presented with a letter or email confirmation of EUSS leave from a Crown Dependency licensing authorities licensing authorities must contact ICESSVECWorkflow@homeoffice.gov.uk. The SVEC will be able to check the individual's status and verify that they have a right to work. The licencing authority must keep a copy of the Crown Dependency letter or email and retain this with the response from the SVEC.

EEA citizens with valid indefinite leave to enter or remain

- 4.31 EEA citizens with Indefinite Leave to Enter or Remain (ILE/R) are not required to make an application to the EUSS, but can do so if they wish.
- 4.32 Since 1 July 2021, EEA citizens with ILE/R have been required to prove their right to work in the same way as other foreign nationals who do not have a digital status. Licensing authorities can carry out a manual check of a copy or scanned copies of their Home Office documentation such as an endorsement / vignette in a current passport stating, 'indefinite leave to enter or remain' or 'no time limit'. Carrying out either a manual check of the documents or an online check, as set out in this guidance, will satisfy the requirements to demonstrate entitlement to work.
- 4.33 If licensing authorities encounter EEA citizens who believe they have ILE/R but do not have a document to confirm this, please encourage them to:
 - apply to the EU Settlement Scheme to obtain settled or pre-settled status (individuals may still be eligible to make an application to the EUSS after 30 June 2021); or
 - apply to the Windrush Scheme to get proof of their ILE/R status
 - If they are from Malta or Cyprus, they may also be able to apply for British citizenship through the Windrush Scheme.

Applications for either scheme are free of charge.

Points-Based Immigration System

- 4.34 Since 1 January 2021, EEA citizens who come to the UK to live, work or study need to obtain immigration status under the points-based system in the same way as other foreign nationals.
- 4.35 The majority of EEA citizens will be provided with an eVisa, however, this will be dependent upon the immigration route and how they made their application. Some EEA citizens will have a Biometric Residence Permit (BRP). Those with a valid BRP can use this to access the online right to work service.
- 4.36 To prove their right to work from 1 July 2021, individuals will provide a share code and their date of birth. This will enable licencing authorities to check their Home Office immigration status via the online service available on GOV.UK: https://www.gov.uk/view-right-to-work.

EEA Citizens without lawful immigration status after 30 June 2021

- 4.37 If an EEA citizen applies for a licence but has not applied to the EUSS (and has no alternative immigration status in the UK), they will not be able to pass a right to work check and a licence must not be issued. If they believe they are eligible for the EUSS, you could signpost them to make an application.
- 4.38 However, there may be situations in which licensing authorities identify an EEA citizen who holds an existing licence, has not applied to the EUSS and does not hold any other form of leave in the UK. This might be discovered during an inspection of the individual's business, an encounter with a law enforcement organisation or via a licence review brought by a Responsible Authority.
- 4.39 A licence issued in respect of an application made on or after 6 April 2017 will lapse when the holder's permission to be in the UK comes to an end.
- 4.40 In these circumstances, advice and assistance may be obtained from the Home Office Local Partnership Managers, or by email <u>ISDLPMSsupportTeam@homeoffice.gov.uk</u> In most cases, a Local Partnership Manager or local Immigration, Compliance and Enforcement (ICE) team will be the first point of contact for licensing authorities.

Immigration Enforcement 28-day notice

- 4.41 Since 1 July 2021, where Immigration Enforcement encounter EEA citizens, or their family members, who are working without immigration status, they will issue a written, 28-day notice to the individual. This provides those who may have been eligible to apply to the EUSS by 30 June 2021 an opportunity to make a late application. They will need to demonstrate reasonable grounds for missing the deadline.
- 4.42 Failure to make an EUSS application may impact their eligibility to access services in the UK, and they may be required to leave the UK.

Non-EEA Family Members

- 4.43 Non-EEA family members of EEA citizens are required to make an application to the EUSS to continue living in the UK lawfully after 30 June 2021. Whilst they will be granted an eVisa, they may also have a valid Biometric Residence Card (BRC).
- 4.44 Since 1 July 2021, non-EEA family members of EEA citizens can provide licensing authorities with a share code and their date of birth. This will enable licencing authorities to check the individual's Home Office immigration status via the online service available on GOV.UK: https://www.gov.uk/view-right-to-work.
- 4.45 Alternatively, the individual may choose to present their BRC, for a manual check. They can continue to use this to prove their right to work until April 2022 when BRC holders will transition to utilising the online service only for right to work checks. Further information about these changes is available at: <u>Right to work checks: an employer's guide</u> <u>- GOV.UK (www.gov.uk)</u>
- 4.46 Finally, where applicable, joining family members (JFM) may apply to the EUSS. Where a JFM makes a valid application to the EUSS, they will receive a Certificate of Application (CoA), issued by the Home Office. They will be able to use their CoA for the purposes of a right to work check.

Further support available for EU, EEA and Swiss citizens

- 4.47 If licensing authorities have existing or prospective licence applicants who require further advice and support with regard to their immigration status, they can access information on GOV.UK: https://www.gov.uk/government/publications/view-and-prove-your-immigration-status-evisa. This also provides further information on how to prove immigration status, how to update personal details, and support available.
- 4.48 If a licence applicant needs help accessing or using the Home Office online services, they can contact the UKVI Resolution Centre:

Telephone: 0300 790 6268, select option 3

Monday to Friday (excluding bank holidays), 8am to 8pm

Saturday and Sunday, 9:30am to 4:30pm.

Criminal record

- 4.49 Regulations made under the 2003 Act require that, in order to substantiate whether or not an applicant has a conviction for an unspent relevant offence, an applicant for the grant of a personal licence must include a criminal conviction certificate, a criminal record certificate or the results of a subject access search of the Police National Computer by the National Identification Service to the licensing authority.
- 4.50 The requirement for an individual to establish whether or not they have unspent convictions for a relevant offence or foreign offence applies whether or not the individual has been living for a length of time in a foreign jurisdiction. It does not follow that such

individuals will not have recorded offences in this country. All applicants are also required to make a clear statement as to whether or not they have been convicted outside England and Wales of a relevant offence or an equivalent foreign offence. This applies both to applicants ordinarily resident in England and Wales and any person from a foreign jurisdiction. Details of relevant offences as set out in the 2003 Act should be appended to application forms for the information of applicants, together with a clear warning that making any false statement is a criminal offence liable to prosecution.

- 4.51 Licensing authorities are required to notify the police when an applicant is found to have an unspent conviction for a relevant offence defined in the 2003 Act or for a foreign offence. The police have no involvement or locus in such applications until notified by the licensing authority.
- 4.52 Civil penalties received after 6 April 2017 for immigration matters are treated in the same way as relevant offences. Licensing authorities are required to notify the Secretary of State for the Home Department (through Home Office Immigration Enforcement) when an applicant declares that they have been issued with an immigration penalty or convicted of an immigration offence or a foreign offence comparable to an immigration offence. The Home Office may object to an application on grounds that granting the personal licence would be prejudicial to the prevention of illegal working in licensed premises. Civil penalties for immigration matters were added to the Licensing Act with effect from 6 April 2017, and penalties received before that date cannot be taken into account in respect of grant, revocation or suspension of a personal licence.
- 4.53 Where an applicant has an unspent conviction for a relevant or foreign offence, and the police object to the application on crime prevention grounds the applicant is entitled to a hearing before the licensing authority. The applicant is also entitled to a hearing if the Home Office (Immigration Enforcement) object to the application on the grounds of the prevention of illegal working where the applicant has an unspent conviction for a relevant immigration offence or has been required to pay an immigration penalty. If the police or Home Office (Immigration Enforcement) do not issue an objection notice and the application otherwise meets the requirements of the 2003 Act, the licensing authority must grant it. Home Office (Immigration Enforcement) can object only with respect to convictions and civil immigration penalties received on or after 6 April 2017.
- 4.54 A number of relevant offences never become spent. However, where an applicant is able to demonstrate that the offence in question took place so long ago and that the applicant no longer has a propensity to re-offend, a licensing authority may consider that it is appropriate to grant the application on the basis that doing so would not undermine the crime prevention objective.
- 4.55 If an application is refused, the applicant will be entitled to appeal against the decision they make. Similarly, if the application is granted despite a police objection notice or an objection from the Home Office (Immigration Enforcement), the chief officer of police or Home Office are entitled to appeal against the licensing authority's determination. Licensing authorities are therefore expected to record in full the reasons for any decision which they make.

Issuing of personal licences by Welsh licensing authorities

4.56 All application forms in Wales should be bilingual. Proceedings before a court must be capable of being conducted in Welsh at the request of the applicant. There is a panel of Welsh speaking magistrates so this can be arranged if necessary. Licensing authorities in Wales should consider issuing personal licences in a bilingual format.

Licensing qualifications

4.57 Details of licensing qualifications accredited by the Secretary of State will be notified to licensing authorities and the details may be viewed on the GOV.UK website.

Relevant licensing authority

4.58 Personal licences remain valid unless surrendered, suspended, revoked or declared forfeit by the courts. For applications made on or after 6 April 2017, a licence granted to someone subject to immigration control will lapse if the individual ceases to be entitled to work in the UK. The requirement to renew a personal licence was removed from the Licensing Act 2003 by the Deregulation Act 2015. While personal licences issued before the 2015 Act have expiry dates, these licences will remain valid and such dates no longer have an effect. Once granted, the licensing authority which issued the licence remains the "relevant licensing authority" for it and its holder, even though the individual may move out of the area or take employment elsewhere. The personal licence itself will give details of the issuing licensing authority.

Changes in name or address

- 4.59 The holder of the licence is required by the 2003 Act to notify the licensing authority of any changes to a holder's name or address. These changes should be recorded by the licensing authority. The holder is also under a duty to notify any convictions for relevant offences to the licensing authority and the courts are similarly required to inform the licensing authority of such convictions, whether or not they have ordered the suspension or forfeiture of the licence. The holder must also notify the licensing authority of any conviction for a foreign offence. These measures ensure that a single record will be held of the holder's history in terms of licensing matters.
- 4.60 The 2003 Act authorises the provision and receipt of such personal information to such agencies for the purposes of that Act.

Specification of new designated premises supervisors

- 4.61 Every premises licence that authorises the sale of alcohol must specify a DPS. This will normally be the person who has been given day to day responsibility for running the premises by the premises licence holder. The only exception is for community premises which have successfully made an application to remove the usual mandatory conditions set out in the 2003 Act. Guidance on such applications is set out in paragraphs 4.84 to 4.97 of this Guidance.
- 4.62 The Government considers it essential that police officers, fire officers or officers of the licensing authority can identify immediately the DPS so that any problems can be dealt with swiftly. For this reason, the name of the DPS and contact details must be specified on the premises licence and this must be held at the premises and displayed in

summary form. The DPS' personal address should not be included in the summary form in order to protect their privacy.

- 4.63 To specify a DPS, the premises licence holder should normally submit an application to the licensing authority (which may include an application for immediate interim effect) with:
 - a form of consent signed by the individual concerned to show that they consent to taking on this responsible role, and
 - the relevant part (Part A) of the licence.
- 4.64 If they are applying in writing, they must also notify the police of the application. If the application is made electronically via GOV.UK or the licensing authority's own electronic facility, the licensing authority must notify the police no later than the first working day after the application is given.
- 4.65 The premises licence holder must notify the existing DPS (if there is one) of the application on the same day as the application is given to the licensing authority. This requirement applies regardless of whether the application was given by means of an electronic facility, or by some other means.
- 4.66 The general guidance in Chapter 8 on electronic applications applies in respect of new applications.
- 4.67 Only one DPS may be specified in a single premises licence, but a DPS may supervise two or more premises as long as the DPS is able to ensure that the licensing objectives are properly promoted and that each premises complies with the 2003 Act and conditions on the premises licence. The DPS is not required to be present at all times when licensed premises are used for the sale of alcohol.
- 4.68 Where there are frequent changes of DPS, the premises licence holder may submit the form in advance specifying the date when the new individual will be in post and the change will take effect.

Police objections to new designated premises supervisors

- 4.69 The police may object to the designation of a new DPS where, in exceptional circumstances, they believe that the appointment would undermine the crime prevention objective. The police can object where, for example, a DPS is first specified in relation to particular premises and the specification of that DPS in relation to the particular premises gives rise to exceptional concerns. For example, where a personal licence holder has been allowed by the courts to retain their licence despite convictions for selling alcohol to children (a relevant offence) and then transfers into premises known for underage drinking.
- 4.70 Where the police do object, the licensing authority must arrange for a hearing at which the issue can be considered and both parties can put forward their arguments. The 2003 Act provides that the applicant may apply for the individual to take up post as DPS immediately and, in such cases, the issue would be whether the individual should be removed from this post. The licensing authority considering the matter must restrict its consideration to the issue of crime and disorder and give comprehensive reasons for its decision. Either party would be entitled to appeal if their argument is rejected.
- 4.71 The portability of personal licences between premises is an important concept under the 2003 Act. It is expected that police objections would arise in only genuinely exceptional

circumstances. If a licensing authority believes that the police are routinely objecting to the designation of new premises supervisors on grounds which are not exceptional, they should raise the matter with the chief officer of police as a matter of urgency.

Police objections to existing designated premises supervisors

4.72 The 2003 Act also provides for the suspension and forfeiture of personal licences by the courts and licensing authorities following convictions for relevant offences, including breaches of licensing law. The police can at any stage after the appointment of a DPS seek a review of a premises licence on any grounds relating to the licensing objectives if problems arise relating to the performance of a DPS. The portability of personal licences is also important to industry because of the frequency with which some businesses move managers from premises to premises. It is not expected that licensing authorities or the police should seek to use the power of intervention as a routine mechanism for hindering the portability of a licence or use hearings of this kind as a fishing expedition to test out the individual's background and character. It is expected that such hearings should be rare and genuinely exceptional.

Convictions and liaison with the courts

- 4.73 Where a personal licence holder is convicted by a court for a relevant offence, the court is under a duty to notify the relevant licensing authority of the conviction and of any decision to order that the personal licence is suspended or declared forfeit. The sentence of the court has immediate effect despite the fact that an appeal may be lodged against conviction or sentence (although the court may suspend the forfeiture or suspension of the licence pending the outcome of any appeal).
- 4.74 When the licensing authority receives such a notification, it should contact the holder and request the licence so that the necessary action can be taken. The holder must then produce the relevant licence to the authority within 14 days. It is expected that the chief officer of police for the area in which the holder resides would be advised if they do not respond promptly. The licensing authority should record the details of the conviction, endorse them on the licence, together with any period of suspension and then return the licence to the holder. If the licence is declared forfeit, it should be retained by the licensing authority.

Licensing authority powers to revoke or suspend personal licences

4.75 The Policing and Crime Act 2017 gives licensing authorities the power to revoke or suspend personal licences, with effect from 6 April 2017. This is a discretionary power; licensing authorities are not obliged to give consideration to all personal licence holders subject to convictions for relevant offences, foreign offences or civil penalties for immigration matters. When a licensing authority has granted a personal licence and becomes aware that the licence holder has been convicted of a relevant offence or foreign offence or been required to pay an immigration penalty, a licensing authority may revoke the licence or suspend it for a period of up to six months. This applies to convictions received and civil immigration penalties which a person has been required to pay at any time before or after the licence was granted, as long as the conviction was received after 6 April 2017, or the requirement to pay the civil penalty arose after 6 April 2017. Only magistrates' courts can order the forfeiture or suspension of a personal licence for convictions received prior to 6 April 2017. The process which must be undertaken by the licensing authority to suspend or revoke a personal licence is set out at section 132A of the 2003 Act. The decision to revoke or suspend a personal licence

must be made by the licensing committee or sub-committee, but the actions required before making a final decision may be made by a licensing officer.

- 4.76 The licensing authority may not take action if the licence holder has appealed against the conviction or the sentence imposed in relation to the offence, until the appeal is disposed of. Where an appeal is not lodged, the licensing authority may not take action until the time limit for making an appeal has expired.
- 4.77 If a licensing authority is considering revoking or suspending a personal licence, the authority must give notice to the licence holder. This notice must invite the holder to make representations about the conviction, any decision of a court in relation to the licence, or any decision by an appellate court if the licence holder has appealed such a decision. The licence holder may also decide to include any other information, for example, about their personal circumstances. The licence holder must be given 28 days to make their representation, beginning on the day the notice was issued. The licensing authority does not need to hold a hearing to consider the representations. Before deciding whether to revoke or suspend the licence the licensing authority must consider any representations made by the licence holder, any decisions made by the court or appellate court in respect of the personal licence of which the licensing authority is aware, and any other information which the licensing authority considers relevant. The licensing authority may not be aware of whether the court considered whether to revoke or suspend the licence, and there is no obligation on the licensing authority to find this out before making a decision themselves. Where the court has considered the personal licence and decided not to take action, this does not prevent the licensing authority from deciding to take action itself. Licensing authorities have different aims to courts in that they must fulfil their statutory duty to promote the licensing objectives, and therefore it is appropriate for the licensing authority to come to its own decision about the licence.
- 4.78 If the licensing authority, having considered a suspension and revocation and subsequently considered all the information made available to it, proposes not to revoke the licence it must give notice to the chief officer of police in the licensing authority's area, and invite the chief officer to make representations about whether the licence should be suspended or revoked, having regard to the prevention of crime. The chief officer may make representations within the period of 14 days from the day they receive the notice from the licensing authority. Any representations made by the chief officer of police must be taken into account by the licensing authority in deciding whether to suspend or revoke the licence. Convictions may come to light via police in another area, for example if the personal licence holder no longer lives in the area of the licensing authority which issued the licence, or if the offence took place in another police force area. In this instance it would be good practice for the police providing the information to notify the police force in the licensing authority area, because it is the local chief officer who must provide representations if the licensing authority proposes not to revoke the licence. Where the licence holder is convicted of immigration offences or has been required to pay a civil penalty for immigration matters, the licensing authority may consider notifying Home Office Immigration Enforcement to enable them to liaise with the Police.

- 4.79 The licensing authority must notify the licence holder and the chief officer of police of the decision made (even if the police did not make representations). The licence holder may appeal the licensing authority's decision to revoke or suspend their personal licence. A decision to revoke or suspend the licence does not take effect until the end of the period allowed for appealing the decision (21 days); or if the decision is appealed against, until the appeal is disposed of.
- 4.80 If the personal licence holder is a DPS, the licensing authority may notify the premises licence holder once the decision to revoke or suspend the licence has been made if it becomes necessary to do so in order for the licensing authority to be able to carry out their functions. The licensing authority may also notify any person who has declared an interest in the premises under section178 of the 2003 Act if it becomes necessary to do so in order for the licence holder to carry out their functions. The licensing authority to be able to carry out their functions. The licensing authority to be able to carry out their functions. The licensing authority to be able to carry out their functions. The licensing authority may invite the premises licence holder to make representations about the personal licence holder before deciding whether to revoke or suspend the licence; this is not a legal requirement and may not be appropriate in all circumstances.

Relevant offences

- 4.81 Relevant offences are set out in Schedule 4 to the 2003 Act. If a person has been required to pay a civil penalty for immigration matters on or after 6 April 2017, this may be taken into consideration in the same way as a relevant offence. Offences added to the list of relevant offences with effect from 6 April 2017 may only be taken into consideration if the conviction was received on or after 6 April 2017.
- 4.82 Closure Notices
- 4.83 Section 19 of the Criminal Justice and Police Act 2001 (the Act) gives licensing authorities, police and local authorities the power to serve a closure notice where any premises are being used (or have been used within the last 24 hours), for the sale of alcohol for consumption on or in the vicinity of the premises; and the activity was not authorised (premises licence, club premises certificate or temporary event notice) or not in accordance with the conditions of authorisation. The notice informs the person with control of, or responsibility for, the activities carried on at the premises (normally the licence holder or the designated premises supervisor) that if unauthorised alcohol sales continue, an application may be made to a court under section 20 for an order to close the premises under section 21 of the Act. Such an application cannot be made less than seven days or more than six months after the service of the closure notice under section 19.
- 4.84 The person should be warned that it is an offence under section 136 of the Licensing Act 2003 to carry on the sale of alcohol without, or in breach of, an authorisation.

Disapplication of certain mandatory conditions for community premises

4.85 The 2003 Act was amended in 2009 to allow certain community premises which have, or are applying for, a premises licence that authorises alcohol sales to also apply to include the alternative licence condition in sections 25A(2) and 41D(3) ("the alternative licence condition") of that Act in the licence instead of the usual mandatory conditions in sections 19(2) and 19(3). Such an application may only be made if the licence holder is,

or is to be, a committee or board of individuals with responsibility for the management of the premises (the "management committee"). If such an application is successful, the effect of the alternative licence condition will be that all alcohol sales will have to be made or authorised by the management committee. There will be no requirement for a DPS or for alcohol sales to be authorised by a personal licence holder.

- 4.86 Community premises are defined as premises that are or form part of a church hall, chapel hall or other similar building; or a village hall, parish hall or community hall or other similar building.
- 4.87 The process requires the completion of a form which is prescribed in regulations made under the 2003 Act. Where the management committee of a community premises is applying for authorisation for the sale of alcohol for the first time, it should include the form with the new premises licence application or the premises licence variation application. No extra payment is required beyond the existing fee for a new application or a variation.
- 4.88 Where a community premises already has a premises licence to sell alcohol, but wishes to include the alternative licence condition in place of the usual mandatory conditions in sections 19(2) and 19(3) of the 2003 Act, it should submit the form on its own together with the required fee.

Definition of community premises

- 4.89 In most instances, it should be self evident whether a premises is, or forms part of a church hall, chapel hall or other similar building; or a village hall, parish hall, community hall or other similar building.
- 4.90 Licensing authorities may have previously taken a view on how to determine whether a premises meets the definition of community premises for the purpose of the fee exemptions set out in regulation 9(2)(b) of the Licensing Act 2003 (Fees) Regulations 2005 (SI 2005/79). As the criteria are the same, premises that qualify for these fee exemptions for regulated entertainment will also be "community premises" for present purposes.
- 4.91 Where it is not clear whether premises are "community premises", licensing authorities will need to approach the matter on a case-by-case basis. The main consideration in most cases will be how the premises are predominately used. If they are genuinely made available for community benefit most of the time, and accessible by a broad range of persons and sectors of the local community for purposes which include purposes beneficial to the community as a whole, the premises will be likely to meet the definition.
- 4.92 Many community premises such as school and private halls are available for private hire by the general public. This fact alone would not be sufficient for such halls to qualify as "community premises". The statutory test is directed at the nature of the premises themselves, as reflected in their predominant use, and not only at the usefulness of the premises for members of the community for private purposes.
- 4.93 If the general use of the premises is contingent upon membership of a particular organisation or organisations, this would strongly suggest that the premises in question are not a "community premises" within the definition. However, the hire of the premises to individual organisations and users who restrict their activities to their own members

and guests would not necessarily conflict with the status of the premises as a "community premises", provided the premises are generally available for use by the community in the sense described above. It is not the intention that qualifying clubs, which are able to apply for a club premises certificate, should instead seek a premises licence with the removal of the usual mandatory conditions in sections 19(2) and 19(3) of the 2003 Act relating to the supply of alcohol.

Management of the premises

- 4.94 Sections 25A(1) and 41D(1) and (2) of the 2003 Act allow applications by community premises to apply the alternative licence condition rather than the usual mandatory conditions in sections 19(2) and 19(3) of the 2003 Act only where the applicant for the licence is the management committee of the premises in question. In addition, sections 25A(6) and 41D(5) require the licensing authority to be satisfied that the arrangements for the management of the premises by the committee or board are sufficient to ensure the adequate supervision of the supply of alcohol on the premises.
- 4.95 The reference to a "committee or board of individuals" is intended to cover any formally constituted, transparent and accountable management committee or structure. Such a committee should have the capacity to provide sufficient oversight of the premises to minimise any risk to the licensing objectives that could arise from allowing the responsibility for supervising the sale of alcohol to be transferred from a DPS and personal licence holder or holders. This could include management committees, executive committees and boards of trustees.
- 4.96 The application form requires applicants to set out how the premises is managed, its committee structure and how the supervision of alcohol sales is to be ensured in different situations (e.g. when the hall is hired to private parties) and how responsibility for this is to be determined in individual cases and discussed within the committee procedure in the event of any issues arising. The application form requires that the community premises submit copies of any constitution or other management documents with their applications and that they provide the names of their key officers. Where the management arrangements are less clear, licensing authorities may wish to ask for further details to confirm that the management board or committee is properly constituted and accountable before taking a decision on whether to grant the application (subject to the views of the police). Community premises may wish to check with the licensing authority before making an application. The management committee is strongly encouraged to notify the licensing authority if there are key changes in the committee's composition and to submit a copy to the chief officer of police. A failure to do so may form the basis of an application to review the premises licence, or be taken into account as part of the consideration of such an application.
- 4.97 As the premise licence holder, the management committee will collectively be responsible for ensuring compliance with licence conditions and the law (and may remain liable to prosecution for one of the offences in the 2003 Act) although there would not necessarily be any individual member always present at the premises. While overall responsibility will lie with the management committee, where the premises are hired out the hirer may be clearly identified as having responsibility for matters falling within his or her control (e.g. under the contract for hire offered by the licence holder), much in the same way that the event organiser may be responsible for an event held

under a Temporary Event Notice. Where hirers are provided with a written summary of their responsibilities under the 2003 Act in relation to the sale of alcohol, the management committee is likely to be treated as having taken adequate steps to avoid liability to prosecution if a licensing offence is committed.

4.98 As indicated above, sections 25A(6) and 41D(5) of the 2003 Act require the licensing authority to consider whether the arrangements for the management of the premises by the committee are sufficient to ensure adequate supervision of the supply of alcohol on the premises. Where private hire for events which include the sale of alcohol is permitted by the licence, it would be necessary to have an effective hiring agreement. Licensing authorities may wish to consider model hiring agreements that have been made available by organisations such as ACRE and Community Matters. Such model agreements can be revised to cater for the circumstances surrounding each hire arrangement; for example to state that the hirer is aware of the licensing objectives and offences in the 2003 Act and will ensure that it will take all appropriate steps to ensure that no offences are committed during the period of the hire.

Police views

4.99 In exceptional circumstances, the chief officer of police for the area in which the community premises is situated can object to a request for inclusion of the alternative licence condition on the grounds of crime and disorder, and any responsible authority or other person can seek reinstatement of the mandatory conditions through a review of the licence (as provided in section 52A of the 2003 Act). The police will want to consider any history of incidents at an establishment in light of the actual or proposed management arrangements, including the use of appropriate hire agreements. If the chief officer of police issues a notice seeking the refusal of the application to include the alternative licence condition, the licensing authority must hold a hearing in order to reach a decision on whether to grant the application.

Appeals

4.100 Where the chief officer of police has made relevant representations against the inclusion of the alternative licence condition, or given a notice under section 41D(6) which was not withdrawn, the chief officer of police can appeal the decision of the licensing authority to allow the inclusion of the alternative licence condition. Similarly, a community premises can appeal a decision by the licensing authority to refuse to include the alternative licence condition following a hearing triggered by relevant representations or by a notice given under section 41D(6). Following a review of the licence in which the mandatory conditions are reinstated, the licence holder may appeal against the decision. If the alternative licence condition is retained on review, the applicant for the review or any person who made relevant representations may appeal against the decision.

5. Who needs a premises licence?

5.1 A premises licence authorises the use of any premises (which is defined in the 2003 Act as a vehicle, vessel or moveable structure or any place or a part of any premises) for licensable activities defined in section 1 of the 2003 Act.

Relevant parts of Act

5.2 In determining whether any premises should be licensed, the following parts of the 2003 Act are relevant:

Relevant part of Act	Description
Section 1	Outlines the licensable activities
Part 3	Provisions relating to premises licences
Part 4	Provisions for qualifying clubs
Section 173	Activities in certain locations which are not licensable
Section 174	Premises that may be exempted on grounds of national security
Section 175	Exemption for incidental non-commercial lottery (e.g. a minor raffle or tombola)
Section 176	Prohibits the sale of alcohol at motorway service areas; and restricts the circumstances in which alcohol may be sold at garages
Section 189	Special provision in relation to the licensing of vessels, vehicles and moveable structures
Section 190	Where the place where a contract for the sale of alcohol is made is different from the place where the alcohol is appropriated to the contract, the sale of alcohol is to be treated as taking place where the alcohol is appropriated to the contract
Section 191	Defines "alcohol" for the purposes of the 2003 Act
Section 192	Defines the meaning of "sale by retail"
Section 193	Defines among other things "premises", "vehicle", "vessel" and "wine"
Schedules 1 and 2	Provision of regulated entertainment and provision of late night refreshment

- 5.3 Section 191 provides the meaning of "alcohol" for the purposes of the 2003 Act. It should be noted that a wide variety of foodstuffs contain alcohol but generally in a highly diluted form when measured against the volume of the product. For the purposes of the 2003 Act, the sale or supply of alcohol which is of a strength not exceeding 0.5 per cent ABV (alcohol by volume) at the time of the sale or supply in question is not a licensable activity. However, where the foodstuff contains alcohol at greater strengths, for example, as with some alcoholic jellies, the sale would be a licensable activity.
- 5.4 The definition of alcohol was amended by the Policing and Crime Act 2017 to include alcohol "in any state". This is to make it clear that products such as powdered and vaporised alcohol fall within the definition provided by the 2003 Act.

Premises licensed for gambling

5.5 Gambling is the subject of separate legislation and licensing authorities should not duplicate any conditions imposed by this legislation when granting, varying or reviewing licences that authorise licensable activities under the 2003 Act. When making a licence application, the applicant may, in detailing the steps to be taken in promoting the licensing objectives, refer to the statutory conditions in respect of their gambling licence (where relevant). In addition, any conditions which are attached to premises licences should not prevent the holder from complying with the requirements of gambling legislation or supporting regulations. Further information about the Gambling Act 2005 can be found on the GOV.UK website.

Designated sports grounds, designated sports events and major outdoor sports stadia

- 5.6 Outdoor sports stadia are regulated by separate legislation and sports events taking place at outdoor stadia do not fall within the definition of the provision of regulated entertainment under the 2003 Act, with the exception of boxing or wrestling entertainment (see 16.49-16.51). Licensing authorities should therefore limit their consideration of applications for premises licences to activities that are licensable under the 2003 Act.
- 5.7 Major stadia will often have several bars and restaurants, including bars generally open to all spectators as well as bars and restaurants to which members of the public do not have free access. Alcohol may also be supplied in private boxes and viewing areas. A premises licence may make separate arrangements for public and private areas or for restaurant areas on the same premises. It may also designate areas where alcohol may not be consumed at all or at particular times.
- 5.8 Licensing authorities should be aware that paragraphs 98 and 99(c) of Schedule 6 to the 2003 Act and the repeals of section 2(1A) and section 5A of the Sporting Events (Control of Alcohol etc.) Act 1985 have not been commenced because their effect would have been different from that which Parliament had intended.

Sporting events at stadia with retractable roofs

5.9 A sporting event at a stadium or sports ground with a roof that opens and closes does not fall within the definition of an "indoor sporting event" under the 2003 Act. As a result, indoor sporting events taking place in these stadia are not 'regulated entertainment' and are not licensable under the 2003 Act.

Vessels

- 5.10 The 2003 Act applies to vessels (including ships and boats) as if they were premises. A vessel which is not permanently moored or berthed is treated as if it were premises situated in a place where it is usually moored or berthed. The relevant licensing authority for considering an application for a premises licence for a vessel is therefore the licensing authority for the area in which it is usually moored or berthed.
- 5.11 However, an activity is not a licensable activity if it takes place aboard a vessel engaged on an international journey. An "international journey" means a journey from a place in the United Kingdom to an immediate destination outside the United Kingdom or a

journey from outside the United Kingdom to an immediate destination in the United Kingdom.

- 5.12 If a vessel is not permanently moored and carries more than 12 passengers it is a passenger ship and will be subject to safety regulation by the Maritime and Coastguard Agency (MCA).
- 5.13 When a licensing authority receives an application for a premises licence in relation to a vessel, it should consider the promotion of the licensing objectives, but should not focus on matters relating to safe navigation or operation of the vessel, the general safety of passengers, or emergency provision; all of which are subject to regulations which must be met before the vessel is issued with its Passenger Certificate and Safety Management Certificate.
- 5.14 If the MCA is satisfied that the vessel complies with Merchant Shipping standards for a passenger ship, the premises should normally be accepted as meeting the public safety objective. In relation to other public safety aspects of the application, representations made by the MCA on behalf of the Secretary of State should be given particular weight.
- 5.15 If a vessel, which is not permanently moored and carries no more than 12 passengers, goes to sea, it will be subject to the Code of Practice for the Safety of Small Commercial Sailing Vessels. This code sets the standards for construction, safety equipment and manning for these vessels and MCA will be able to confirm that it has a valid safety certificate.
- 5.16 If a vessel carries no more than 12 passengers and does not go to sea, it may be regulated or licensed by the competent harbour authority, navigation authority or local authority. The recommended standards for these vessels are set out in the (non-statutory) Inland Waters Small Passenger Boat Code, which provides best practice guidance on the standards for construction, safety equipment and manning. Some authorities may use their own local rules. MCA has no direct responsibility for these vessels and will not normally comment on a premises licence application.

International airports and ports

- 5.17 Under the 2003 Act, the Secretary of State may 'designate' a port, hoverport or airport with a substantial amount of international traffic so that an activity carried on there is not licensable. The Secretary of State may also preserve existing designations made under earlier legislation.
- 5.18 Areas at designated ports which are "airside" or "wharfside" are included in the exemption in the 2003 Act from the licensing regime. The non-travelling public does not have access to these areas and they are subject to stringent bye-laws. The exemption allows refreshments to be provided to travellers at all times of the day and night. Other parts of designated ports, hoverports and airports are subject to the normal licensing controls.

Vehicles

- 5.19 Under the 2003 Act, alcohol may not be sold on a moving vehicle and the vehicle may not be licensed for that purpose. However, licensing authorities may consider applications for the sale of alcohol from a parked or stationary vehicle. For example, mobile bars could sell alcohol at special events as long as they were parked. Any permission granted would relate solely to the place where the vehicle is parked and where sales are to take place.
- 5.20 The provision of any entertainment on premises consisting of or forming part of any vehicle while it is in motion and not permanently or temporarily parked is not regulated entertainment for the purposes of the 2003 Act.

Trains and aircraft

5.21 Railway vehicles and aircraft engaged on journeys are exempted from the requirement to have an authorisation to carry on licensable activities (although a magistrates' court can make an order to prohibit the sale of alcohol on a railway vehicle if this is appropriate to prevent disorder). Licensing authorities should note that some defunct aircraft and railway carriages remain in a fixed position and are used as restaurants and bars. These premises are subject to the provisions of the 2003 Act.

Garages and motorway service areas

- 5.22 Section 176 of the 2003 Act prohibits the sale or supply of alcohol at motorway service areas (MSAs) and from premises which are used primarily as a garage, or are part of premises used primarily as a garage. Premises are used primarily as a garage if they are used for one or more of the following:
 - the retailing of petrol;
 - the retailing of derv (diesel);
 - the sale of motor vehicles; and
 - the maintenance of motor vehicles.
- 5.23 It is for the licensing authority to decide, based on the licensing objectives, whether it is appropriate for that premises to be granted a licence, taking into account the documents and information listed in section 17(3) and (4) which must accompany the application.
- 5.24 If a licence is granted in respect of a premises and the primary use of that premises subsequently changes (for example, the primary use becomes that of a garage rather than a shop) it would no longer be legal to sell alcohol on that premises. If a relevant representation is made, the licensing authority must decide whether or not the premises are used primarily as a garage. The licensing authority may ask the licence holder to provide further information to help establish what the primary use of the premises is.

Large scale time-limited events requiring premises licences

5.25 Licensing authorities should note that a premises licence may be sought for a short, discrete period. The 2003 Act provides that a temporary event notice is subject to various limitations (see Chapter 7 of this Guidance). The temporary provision of licensable activities that fall outside these limits will require the authority of a premises licence if the premises are currently unlicensed for the activities involved.

5.26 The procedures for applying for and granting such a licence are identical to those for an unlimited duration premises licence except that it should be stated on the application that the applicant's intention is that the period of the licence should be limited. Licensing authorities should clearly specify on such a licence when it comes into force and when it ceases to have effect. If the sale of alcohol is involved, a personal licence holder must be specified as the designated premises supervisor.
6. Club premises certificates

6.1 This Chapter covers the administration of the processes for issuing, varying, and reviewing club premises certificates and other associated procedures.

General

- 6.2 Clubs are organisations where members have joined together for particular social, sporting or political purposes. They may then combine to buy alcohol in bulk as members of the organisation to supply in the club.
- 6.3 Technically the club only sells alcohol by retail at such premises to guests. Where members purchase alcohol, there is no sale (as the member owns part of the alcohol stock) and the money passing across the bar is merely a mechanism to preserve equity between members where one may consume more than another.
- 6.4 Only 'qualifying' clubs may hold club premises certificates. In order to be a qualifying club, a club must have at least 25 members and meet the qualifying conditions set out in paragraph 6.9. The grant of a club premises certificate means that a qualifying club is entitled to certain benefits. These include:
 - the authority to supply alcohol to members and sell it to guests on the premises to which the certificate relates without the need for any member or employee to hold a personal licence;
 - the authority to provide late night refreshment to members of the club without requiring additional authorisation;
 - more limited rights of entry for the police and authorised persons because the premises are considered private and not generally open to the public; and
 - exemption from orders of the magistrates' court for the closure of all licensed premises in an area when disorder is happening or expected.
- 6.5 Qualifying clubs should not be confused with proprietary clubs, which are clubs run commercially by individuals, partnerships or businesses for profit. These require a premises licence and are not qualifying clubs.
- 6.6 A qualifying club will be permitted under the terms of a club premises certificate to sell and supply alcohol to its members and their guests only. Instant membership is not permitted and members must wait at least two days between their application and their admission to the club. A qualifying club may choose to apply for a premises licence if it decides that it wishes to offer its facilities commercially for use by the general public, including the sale of alcohol to them. However, an individual on behalf of a club may give temporary event notices. See Chapter 7.
- 6.7 The 2003 Act does not prevent visitors to a qualifying club being supplied with alcohol as long as they are 'guests' of any member of the club , and nothing in the 2003 Act prevents the admission of such people as guests without prior notice. The 2003 Act does not define "guest" and whether or not somebody is a genuine guest would in all cases be a question of fact.
- 6.8 There is no mandatory requirement under the 2003 Act for guests to be signed in by a member of the club. However, a point may be reached where a club is providing

commercial services to the general public in a way that is contrary to its qualifying club status. It is at this point that the club would no longer be conducted in "good faith" and would no longer meet "general condition 3" for qualifying clubs in section 62 of the 2003 Act. Under the 2003 Act, the licensing authority must decide when a club has ceased to operate in "good faith" and give the club a notice withdrawing the club premises certificate. The club is entitled to appeal against such a decision to a magistrates' court. Unless the appeal is successful, the club would need to apply for a premises licence to authorise licensable activities taking place there.

Qualifying conditions

6.9 Section 62 of the 2003 Act sets out five general conditions which a relevant club must meet to be a qualifying club. Section 63 also sets out specified matters for licensing authorities to enable them to determine whether a club is established and conducted in good faith – the third qualifying condition. Section 64 sets out additional conditions which only need to be met by clubs intending to supply alcohol to members and guests. Section 90 of the 2003 Act gives powers to the licensing authority to issue a notice to a club withdrawing its certificate where it appears that it has ceased to meet the qualifying conditions. There is a right of appeal against such a decision.

Associate members and guests

6.10 As well as their own members and guests, qualifying clubs are also able to admit associate members and their guests (i.e. members and guests from another 'recognised club' as defined by section 193 of the 2003 Act) to the club premises when qualifying club activities are being carried on without compromising the use of their club premises certificate.

Applications for the grant or variation of club premises certificates

- 6.11 The arrangements for applying for or seeking to vary club premises certificates are extremely similar to those for a premises licence. Clubs may also use the minor variation process to make small changes to their certificates as long as these could have no adverse impact on the licensing objectives. Licensing authorities should refer to Chapter 8 of this Guidance on the handling of such applications. Licensing authorities do not have to satisfy themselves that applicants for club premises certificates are entitled to work in the UK before issuing a club premises certificate. Consequently, Home Office Immigration Enforcement is not a responsible authority in relation to club premises certificates.
- 6.12 In addition to a plan of the premises and a club operating schedule, clubs must also include the rules of the club with their application (as well as making a declaration to the licensing authority in accordance with regulations made under the 2003 Act). On notifying any alteration to these rules to the licensing authority, the club is required to pay a fee set down in regulations. Licensing authorities cannot require any changes to the rules to be made as a condition of receiving a certificate unless relevant representations have been made. However, if a licensing authority is satisfied that the rules of a club indicate that it does not meet the qualifying conditions in the 2003 Act, a club premises certificate should not be granted.

Steps needed to promote the licensing objectives

6.13 Club operating schedules prepared by clubs, must include the steps it intends to take to promote the licensing objectives. These will be translated into conditions included in the certificate, unless the conditions have been modified by the licensing authority following consideration of relevant representations. Guidance on these conditions is given in Chapter 10 of this Guidance.

7. Temporary Event Notices (TENs)

7.1 This Chapter covers the arrangements in Part 5 of the 2003 Act for the temporary carrying on of licensable activities which are not authorised by a premises licence or club premises certificate.

General

- 7.2 The system of permitted temporary activities is intended as a light touch process, and as such, the carrying on of licensable activities does not have to be authorised by the licensing authority on an application. Instead, a person wishing to hold an event at which such activities are proposed to be carried on (the "premises user") gives notice to the licensing authority of the event (a "temporary event notice" or "TEN").
- 7.3 The TEN must be given to the licensing authority in the form prescribed in regulations made under the 2003 Act. The form requires the user to describe key aspects of the proposed event, including the general nature of the premises and the event, the licensable activities intended to be carried on at the proposed event, including whether they will include any relevant entertainment as defined in Schedule 3 to the Local Government (Miscellaneous Provisions) Act 1982 (i.e. displays of nudity designed to sexually stimulate any member of the audience including, but not limited to, lap dancing and pole dancing). Under the 2003 Act 'premises' can mean any place. Events authorised under a TEN will therefore not always be in a building with a formal address and can take place, for example, in public parks and plots within larger areas of land. In all cases, the premises user should provide a clear description of the area in which they propose to carry on licensable activities, including whether the premises are, for example, an open field or a beer tent. Local authorities and "relevant persons" (see below) may wish to make enquiries where appropriate to satisfy themselves of the exact location for where a TEN is being given.
- 7.4 Unless it is sent electronically, it must be sent to the relevant licensing authority, to the police and "local authority exercising environmental health functions" ("EHA") at least ten working days before the event. A premises user may also give a limited number of "late TENs" to the licensing authority less than 10 working days before the event to which they relate, although certain restrictions apply (see paragraphs 7.12-7.14). "Working day" under the 2003 Act means any day other than a Saturday, Sunday, Christmas Day, Good Friday or Bank Holiday. For limited purposes in relation to a TEN, the 2003 Act defines a "day" as a period of 24 hours beginning at midnight.
- 7.5 If a TEN is sent electronically via GOV.UK or the licensing authority's own facility, the licensing authority must notify the police and EHA as soon as possible and no later than the first working day after the TEN is given.
- 7.6 The police or EHA ("relevant persons" for the purposes of TENs) may intervene to prevent such an event taking place by sending an objection to the licensing authority, which the licensing authority must consider on the basis of the statutory licensing objectives and decide whether the event should go ahead. A relevant person may also intervene by agreeing a modification of the proposed arrangements directly with the TENs user (see paragraph 7.36). If a relevant person sends an objection, this may

result in the licensing authority imposing conditions on a TEN but only where the venue at which the event is to be held has an existing premises licence or club premises certificate. When giving a TEN, the premises user should consider the promotion of the four licensing objectives. The licensing authority may only otherwise intervene if the statutory permitted limits on TENs would be exceeded (see paragraphs 7.15-7.22).

7.7 A TEN does not relieve the premises user from any requirements under planning law for appropriate planning permission where it is required.

Standard and late temporary event notices

7.8 There are two types of TEN: a standard TEN and a late TEN. These are subject to different processes: a standard notice is given no later than ten working days before the event to which it relates; and a late notice is given not before nine and not later than five working days before the event.

Standard temporary event notices

- 7.9 "Ten working days" (and other periods of days which apply to other requirements in relation to TENs) exclude the day the notice is received and the first day of the event.
- 7.10 The police and EHA have a period of three working days from when they are given the notice to object to it on the basis of any of the four licensing objectives.
- 7.11 Although ten clear working days is the minimum possible notice that may be given, licensing authorities should publicise their preferences in terms of advance notice and encourage premises users to provide the earliest possible notice of events planned by them. Licensing authorities should also consider publicising a preferred maximum time in advance of an event by when TENs should ideally be given to them.

Late temporary event notices

- 7.12 Late TENs are intended to assist premises users who are required for reasons outside their control to, for example, change the venue for an event at short notice. However, late TENs may, of course, be given in any circumstances providing the limits specified at paragraph 7.15 are not exceeded.
- 7.13 Late TENs can be given up to five clear working days but no earlier than nine clear working days before the event is due to take place and, unless given electronically to the licensing authority, must also be sent by the premises user to the police and EHA. A late TEN given less than five days before the event to which it relates will be returned as void and the activities to which it relates will not be authorised.
- 7.14 A key difference between standard and late TENs is the process following an objection notice from the police or EHA. Where an objection notice is received in relation to a standard TEN the licensing authority must hold a hearing to consider the objection, unless all parties agree that a hearing is unnecessary. If the police, EHA or both give an objection to a late TEN, the notice will not be valid and the event will not go ahead as there is no scope for a hearing or the application of any existing licence conditions.

Limitations

- 7.15 A number of limitations are imposed on the use of TENs by the 2003 Act:
 - the number of times a premises user may give a TEN is 50 times in a calendar year

for a personal licence holder and five times in a calendar year for other people;

- the number of times a premises user may give a late TEN is limited to 10 times in a calendar year for a personal licence holder and twice for other people. Late TENs count towards the total number of permitted TENs (i.e. the limit of five TENs a year for non-personal licence holders and 50 TENs for personal licence holders). A notice that is given less than ten working days before the event to which it relates, when the premises user has already given the permitted number of late TENs in that calendar year, will be returned as void and the activities described in it will not be authorised.
- the number of times a TEN may be given for any particular premises is 15 times in a calendar year;
- the maximum duration of an event authorised by a TEN is 168 hours (seven days);
- the maximum total duration of the events authorised by TENs in relation to individual premises is 21 days in a calendar year;
- the maximum number of people attending at any one time is 499 (including any staff and any other persons); and
- the minimum period between events authorised under separate TENs in relation to the same premises (not including withdrawn TENs) by the same premises user is 24 hours.
- 7.16 Any associate, relative or business partner of the premises user is considered to be the same premises user in relation to these restrictions. The 2003 Act defines an associate, in relation to the premises user, as being:
 - the spouse or civil partner of that person;
 - a child, parent, grandchild, grandparent, brother or sister of that person;
 - an agent or employee of that person; or
 - the spouse or civil partner of a person listed in either of the two preceding bullet points.
- 7.17 A person living with another person as their husband or wife, is treated for these purposes as their spouse. 'Civil partner' has its meaning in the Civil Partnership Act 2004.
- 7.18 A TEN that is given may be subsequently withdrawn by the TEN user by giving the licensing authority a notice to that effect no later than 24 hours before the beginning of the event period specified in the TEN. Otherwise, the TEN will be included within the limits of TENs allowed in a given calendar year, even if the event does not go ahead.
- 7.19 Once these limits have been reached, the licensing authority should issue a counter notice (permitted limits) if any more are given. Proposed activities that exceed these limits will require a premises licence or club premises certificate.
- 7.20 TENs may be given in respect of premises which already have a premises licence or club premises certificate to cover licensable activities not permitted by the existing authorisation.
- 7.21 In determining whether the maximum total duration of the periods covered by TENs at any individual premises has exceeded 21 days, an event beginning before midnight and continuing into the next day would count as two days towards the 21 day limitation.

7.22 There is nothing in the 2003 Act to prevent notification of multiple events at the same time, provided the first event is at least ten working days away (or five working days away in the case of a late TEN). For example, an individual personal licence holder wishing to exhibit and sell beer at a series of farmers' markets may wish to give several notices simultaneously. However, this would only be possible where the limits are not exceeded in the case of each notice. Where the events are due to take place in different licensing authority (and police) areas, the respective licensing authorities and relevant persons would each need to be notified accordingly.

Who can give a temporary event notice?

Personal licence holders

7.23 A personal licence holder can give a TEN at any premises on up to 50 occasions in a calendar year. This limit is inclusive of any late TENs (subject to a maximum of 10) given in the same year. The use of each TEN must of course observe the limits described above, including the limit of 15 TENs in respect of each premises in a calendar year.

Non-personal licence holders

7.24 The 2003 Act provides that any individual aged 18 or over may give a TEN to authorise the carrying on of all licensable activities under the Licensing Act 2003, whether or not that individual holds a personal licence. Such an individual will not, therefore, have met the requirements that apply to a personal licence holder under Part 6 of the 2003 Act. Where alcohol is not intended to be sold, this should not matter. However, many events will involve a combination of licensable activities and the 2003 Act limits the number of notices that may be given by any non-personal licence holder to five occasions in a calendar year (inclusive of any late TENs – subject to a maximum of 2 - in the same year). In every other respect, the Guidance and information set out in the paragraphs above applies.

Role of the licensing authority

- 7.25 The licensing authority must check that the limitations set down in Part 5 of the 2003 Act are being observed and intervene if they are not (see paragraph 7.15). For example, a TEN would be void unless there is a minimum of 24 hours between events notified by the same premises user, or an associate or someone who is in business with the relevant premises user in respect of the same premises. This is to prevent evasion of the seven day (or 168 hour) limit on such events and the need to obtain a full premises licence or club premises certificate for more major or permanent events. In addition, for these purposes, a TEN is treated as being from the same premises user if it is given by an associate.
- 7.26 Where the application is not within the statutory parameters described earlier, the licensing authority will issue a counter notice to the premises user.
- 7.27 Where the TEN is in order, the relevant fee paid and the event falls within the prescribed limits, the licensing authority will record the notice in its register and send an acknowledgement to the premises user (which may be given electronically). The licensing authority must do so, no later than the end of the first working day following the day on which it was received (or by the end of the second working day if it was received

on a non-working day), unless an objection notice is received beforehand from the police or EHA on the basis of any of the four licensing objectives (see paragraphs below).

- 7.28 If the licensing authority receives an objection notice from the police or EHA that is not withdrawn, it must (in the case of a standard TEN only) hold a hearing to consider the objection unless all parties agree that this is unnecessary. The licensing committee may decide to allow the licensable activities to go ahead as stated in the notice. If the notice is in connection with licensable activities at licensed premises, the licensing authority may also impose one or more of the existing licence or certificate conditions on the TEN (insofar as such conditions are not inconsistent with the event) if it considers that this is appropriate for the promotion of the licensing objectives. If the authority decides to impose conditions, it must give notice to the premises user which includes a statement of conditions (a "notice (statement of conditions)") and provide a copy to each relevant party. Alternatively, it can decide that the event would undermine the licensing objectives and should not take place. In this case, the licensing authority must give a counter notice.
- 7.29 Premises users are not required to be on the premises during the event authorised by the TEN, but they will remain liable to prosecution for certain offences that may be committed at the premises during the period covered by it. These include, for example, the offences of the sale of alcohol to a person who is drunk; persistently selling alcohol to children and allowing disorderly conduct on licensed premises.
- 7.30 In the case of an event authorised by a TEN, failure to adhere to the requirements of the 2003 Act, such as the limitation of no more than 499 being present at any one time, would mean that the event was unauthorised. In such circumstances, the premises user would be liable to prosecution.
- 7.31 Section 8 of the 2003 Act requires licensing authorities to keep a register containing certain matters, including a record of TENs received. Under Schedule 3 of the 2003 Act, the licensing authority must also keep a record of such matters including any notice of withdrawal of a TEN, any counter notice to a TEN given following an objection by a relevant person and any TEN received following modification. If requested to do so, a licensing authority must supply a person with a copy of the information contained in any entry in its register. Each licensing authority must also provide facilities for making the information contained in the entries in its register available for inspection by any person during office hours and without payment. Licensing authorities may wish to consider bringing TENs to the attention of local councillors and residents by making their register available online or including relevant details of a TEN when it is received, along with notice of licence applications on the authority's website. There is no requirement to record all the personal information given on a TEN.

Police and environmental health intervention

- 7.32 The system of permitted temporary activities gives police and EHAs the opportunity to consider whether they should object to a TEN on the basis of any of the licensing objectives.
- 7.33 If the police or EHA believe that allowing the premises to be used in accordance with the TEN will undermine the licensing objectives, they must give the premises user and

the licensing authority an objection notice. The objection notice must be given within the period of three working days following the day on which they received the TEN.

- 7.34 Where a standard TEN was given, the licensing authority must consider the objection at a hearing before a counter notice can be issued. At the hearing, the police, EHA and the premises user may make representations to the licensing authority. Following the hearing, the licensing authority may decide to impose conditions which already apply to an existing premises licence or club premises certificate at the venue, or issue a counter notice to prevent the event going ahead. As noted above, there is no scope for hearings (or appeals) in respect of late TENs and if objections are raised by the police or EHA in relation to a late TEN, the notice will be invalid and the event will not go ahead.
- 7.35 Such cases might arise because of concerns about the scale, location, timing of the event or concerns about public nuisance even where the statutory limits on numbers are being observed. The premises user who signs the form is legally responsible for ensuring that the numbers present do not exceed the permitted limit at any one time. In cases where there is reason to doubt that the numbers will remain within the permitted limit the premises user should make clear what the nature of the event(s) is and how they will ensure that the permitted persons limit will not be exceeded. For example, where notices are being given for TENs simultaneously on adjacent plots of land it may be appropriate for door staff to be employed with counters. In each case it is important that licensing authorities and relevant persons can consider whether they believe that the premises user intends to exceed the 499 person limit, or will be unable to control or know whether the limit will be exceeded. Where the planned activities are likely to breach the statutory limits or undermine the licensing objectives, it is likely to be appropriate for the police or EHA to raise objections.
- 7.36 However, in most cases, where for example, alcohol is supplied away from licensed premises at a temporary bar under the control of a personal licence holder, (such as at weddings with a cash bar or small social or sporting events) this should not usually give rise to the use of these powers.

Modification

7.37 As noted above, the police or EHA (as "relevant persons") may contact the premises user to discuss their objections and try to come to an agreement which will allow the proposed licensable activities to proceed. The TEN can be modified (for example, by changing the details of the parts of the premises that are to be used for the event, the description of the nature of the intended activities or their duration). The other relevant person has to agree for the modification to be made. There is no scope under the 2003 Act for the modification of a late TEN.

Applying conditions to a TEN

- 7.38 The 2003 Act provides that only the licensing authority can impose conditions to a TEN from the existing conditions on the premises licence or club premises certificate at the venue. The licensing authority can only do so:
 - if the police or the EHA have objected to the TEN;
 - if that objection has not been withdrawn;
 - if there is a licence or certificate in relation to at least a part of the premises in respect

of which the TEN is given;

- and if the licensing authority considers it appropriate for the promotion of the licensing objectives to impose one or more conditions.
- 7.39 This decision is one for the licensing authority alone, regardless of the premises user's views or willingness to accept conditions. The conditions must be notified to the premises user on the form prescribed by regulations.

Duty of premises users to keep and produce TENs

7.40 Where a TEN is not prominently displayed at the premises, the police and licensing officers have the right under sections 109(5) and (6) of the 2003 Act to request the premises user (or relevant nominated person who has the TEN in their custody) to produce the TEN for examination. If the police do not intervene when a TEN is given, they will still be able to rely on their powers of closure under the Anti-social Behaviour, Crime and Policing Act 2014³.

8. Applications for premises licences

Relevant licensing authority

- 8.1 Premises licences are issued by the licensing authority in which the premises are situated or, in the case of premises straddling an area boundary, the licensing authority where the greater part of the premises is situated. Where the premises is located equally in two or more areas, the applicant may choose but, in these rare cases, it is important that each of the licensing authorities involved maintain close contact.
- 8.2 Section 13 of the 2003 Act defines the parties holding important roles in the context of applications, inspection, monitoring and reviews of premises licences.

Authorised persons

- 8.3 The first group –"authorised persons"– are bodies empowered by the 2003 Act to carry out inspection and enforcement roles. The police and immigration officers are not included because they are separately empowered by the 2003 Act to carry out their duties.
- 8.4 For all premises, the authorised persons include:
 - officers of the licensing authority;
 - fire inspectors;
 - inspectors with responsibility in the licensing authority's area for the enforcement of the Health and Safety at Work etc Act 1974;
 - · officers of the local authority exercising environmental health functions
- 8.5 Local authority officers will most commonly have responsibility for the enforcement of health and safety legislation, but the Health and Safety Executive is responsible for certain premises. In relation to vessels, authorised persons also include an inspector or a surveyor of ships appointed under section 256 of the Merchant Shipping Act 1995. These would normally be officers acting on behalf of the Maritime and Coastguard Agency. The Secretary of State may prescribe other authorised persons by means of regulations, but has not currently prescribed any additional bodies. If any are prescribed, details will be made available on the GOV.UK website.
- 8.6 Where an immigration officer has reason to believe that any premises are being used for a licensable activity, the officer may enter the premises with a view to seeing whether an offence under any of the Immigration Acts is being committed in connection with the licensable activity.

Responsible authorities

- 8.7 The second group –"responsible authorities"– are public bodies that must be fully notified of applications and that are entitled to make representations to the licensing authority in relation to the application for the grant, variation or review of a premises licence. These representations must still be considered 'relevant' by the licensing authority and relate to one or more of the licensing objectives. For all premises, responsible authorities include:
 - · the relevant licensing authority and any other licensing authority in whose area part of

the premises is situated;

- the chief officer of police;
- the local fire and rescue authority;
- the relevant enforcing authority under the Health and Safety at Work etc Act 1974;
- the local authority with responsibility for environmental health;
- the local planning authority;
- a body that represents those who are responsible for, or interested in, matters relating to the protection of children from harm;
- each local authority's Director of Public Health (DPH) in England⁴ and Local Health Boards (in Wales);
- the local weights and measures authority (trading standards); and
- Home Office Immigration Enforcement (on behalf of the Secretary of State).
- 8.8 The licensing authority should indicate in its statement of licensing policy which body it recognises to be competent to advise it on the protection of children from harm. This may be the local authority social services department, the Local Safeguarding Children Board or another competent body. This is important as applications for premises licences have to be copied to the responsible authorities in order for them to make any representations they think are relevant.
- 8.9 In relation to a vessel, responsible authorities also include navigation authorities within the meaning of section 221(1) of the Water Resources Act 1991 that have statutory functions in relation to the waters where the vessel is usually moored or berthed, or any waters where it is proposed to be navigated when being used for licensable activities; the Environment Agency; the Canal and River Trust; and the Secretary of State (who in practice acts through the Maritime and Coastguard Agency (MCA)). In practice, the Environment Agency and the Canal and River Trust only have responsibility in relation to vessels on waters for which they are the navigation statutory authority.
- 8.10 The MCA is the lead responsible authority for public safety, including fire safety, affecting passenger ships (those carrying more than 12 passengers) wherever they operate and small commercial vessels (carrying no more than 12 passengers) which go to sea. The safety regime for passenger ships is enforced under the Merchant Shipping Acts by the MCA which operates certification schemes for these vessels. Fire and rescue authorities, the Health and Safety Executive and local authority health and safety inspectors should normally be able to make "nil" returns in relation to such vessels and rely on the MCA to make any appropriate representations in respect of this licensing objective.
- 8.11 Merchant Shipping legislation does not, however, apply to permanently moored vessels. So, for example, restaurant ships moored on the Thames Embankment, with permanent shore connections should be considered by the other responsible authorities concerned with public safety, including fire safety. Vessels carrying no more than 12 passengers which do not go to sea are not subject to MCA survey and certification, but may be licensed by the local port or navigation authority.

⁴ This change was made as a result of the commencement of measures in the Health and Social Care Act 2012 which amended the 2003 Act and further provision in the NHS Bodies and Local Authorities (Partnership Arrangements, Care Trusts, Public Health and Local Healthwatch) Regulations 2012.

8.12 The Secretary of State may prescribe other responsible authorities by means of regulations. Any such regulations are published on the Government's legislation website: <u>www.legislation.gov.uk</u>.

Other persons

- 8.13 As well as responsible authorities, any other person can play a role in a number of licensing processes under the 2003 Act. This includes any individual, body or business entitled to make representations to licensing authorities in relation to applications for the grant, variation, minor variation or review of premises licences and club premises certificates, regardless of their geographic proximity to the premises. In addition, these persons may themselves seek a review of a premises licence. Any representations made by these persons must be 'relevant', in that the representation relates to one or more of the licensing objectives. It must also not be considered by the licensing authority to be frivolous or vexatious. In the case of applications for reviews, there is an additional requirement that the grounds for the review should not be considered by the licensing authority to be repetitious. Chapter 9 of this guidance (paragraphs 9.4 to 9.10) provides more detail on the definition of relevant, frivolous and vexatious representations.
- 8.14 While any of these persons may act in their own right, they may also request that a representative makes the representation to the licensing authority on their behalf. A representative may include a legal representative, a friend, a Member of Parliament, a Member of the Welsh Government, or a local ward or parish councillor who can all act in such a capacity.

Who can apply for a premises licence?

- 8.15 Any person (if an individual aged 18 or over) who is carrying on or who proposes to carry on a business which involves the use of premises (any place including one in the open air) for licensable activities may apply for a premises licence either on a permanent basis or for a time-limited period.
- 8.16 "A person" in this context includes, for example, a business or a partnership. Licensing authorities should not require the nomination of an individual to hold the licence or determine the identity of the most appropriate person to hold the licence.
- 8.17 In considering joint applications (which is likely to be a rare occurrence), it must be stressed that under section 16(1)(a) of the 2003 Act each applicant must be carrying on a business which involves the use of the premises for licensable activities. In the case of public houses, this would be easier for a tenant to demonstrate than for a pub owning company that is not itself carrying on licensable activities. Where licences are to be held by businesses, it is desirable that this should be a single business to avoid any lack of clarity in accountability.
- 8.18 A public house may be owned, or a tenancy held, jointly by a husband and wife, civil partners or other partnerships of a similar nature, and both may be actively involved in carrying on the licensable activities. In these cases, it is entirely possible for the husband and wife or the partners to apply jointly as applicant for the premises licence, even if they are not formally partners in business terms. This is unlikely to lead to the same issues of clouded accountability that could arise where two separate businesses

apply jointly for the licence. If the application is granted, the premises licence would identify the holder as comprising both names and any subsequent applications, for example for a variation of the licence, would need to be made jointly.

8.19 A wide range of other individuals and bodies set out in section 16 of the 2003 Act may apply for premises licences. They include, for example, Government Departments, local authorities, hospitals, schools, charities or police forces. In addition to the bodies listed in section 16, the Secretary of State may prescribe by regulations other bodies that may apply and any such regulations are published on the Government's legislation website. There is nothing in the 2003 Act which prevents an application being made for a premises licence at premises where a premises licence is already held.

Application forms

8.20 The Provision of Services Regulations 2009 require local authorities to ensure that all procedures relating to access to, or the exercise of, a service activity may be easily completed, at a distance and by electronic means. Electronic application facilities for premises licences may be found either on GOV.UK or the licensing authority's own website. It remains acceptable to make an application in writing.

Electronic applications

8.21 Applicants may apply using the licence application forms available on GOV.UK, or will be re-directed from GOV.UK to the licensing authority's own electronic facility if one is available. Applicants may also apply directly to the licensing authority's facility without going through GOV.UK.

Electronic applications using forms on gov.uk

- 8.22 GOV.UK will send a notification to the licensing authority when a completed application form is available for it to download from GOV.UK. This is the day that the application is taken to be 'given' to the licensing authority, even if it is downloaded at a later stage, and the application must be advertised from the day after that day (as for a written application). The licensing authority must acknowledge the application as quickly as possible, specifying the statutory time period and giving details of the appeal procedure.
- 8.23 The period of 28 consecutive days during which the application must be advertised on a notice outside the premises is, effectively, the statutory timescale by which the application must be determined (unless representations are made). This will be published on GOV.UK and must also be published on the licensing authority's own electronic facility if one exists. If no representations are made during this period, the licensing authority must notify the applicant as quickly as possible that the licence has been granted. The licensing authority must send the licence to the applicant as soon as possible after this, but the applicant may start the licensed activity as soon as they have been notified that the application is granted (subject to compliance with the conditions of the licence). The licence may be supplied in electronic or written format as long as the applicant is aware which document constitutes 'the licence'. If representations are made, the guidance in Chapter 9 applies.

Requirement to copy application to responsible authorities

8.24 The licensing authority must copy electronic applications, made via GOV.UK or its own facility, to responsible authorities no later than the first working day after the application

is given. However, if an applicant submits any part of their application in writing, the applicant will remain responsible for copying it to responsible authorities.

Applications via the local authority electronic application facility

8.25 Where applications are made on the licensing authority's own electronic facility, the application will be taken to be 'given' when the applicant has submitted a complete application form and paid the fee. The application is given at the point at which it becomes accessible to the authority by means of the facility. The licensing authority must acknowledge the application as quickly as possible, specifying the statutory time period and giving details of the appeal procedure.

'Holding' and 'deferring' electronic applications

- 8.26 The Government recommends (as for written applications) that electronic applications should not be returned if they contain obvious and minor errors such as typing mistakes, or small errors that can be rectified with information already in the authority's possession. However, if this is not the case and required information is missing or incorrect, the licensing authority may 'hold' the application until the applicant has supplied all the required information. This effectively resets the 28 day period for determining an application and may be done any number of times until the application form is complete. Licensing authorities must ensure that they notify the applicant as quickly as possible of any missing (or incorrect) information, and explain how this will affect the statutory timescale and advertising requirements.
- 8.27 If an application has been given at the weekend, the notice advertising the application (where applicable) may already be displayed outside the premises by the time that the licensing authority downloads the application. It is therefore recommended that, if a licensing authority holds an application, it should inform the applicant that the original (or if necessary, amended) notice must be displayed until the end of the revised period. The licensing authority should also advise the applicant that they should not advertise the application in a local newspaper until they have received confirmation from the licensing authority that the application includes all the required information. To ensure clarity for applicants, the Government recommends that licensing authorities include similar advice on their electronic application facilities (where these exist) to ensure that applicants do not incur any unnecessary costs.
- 8.28 If an applicant persistently fails to supply the required information, the licensing authority may refuse the application and the applicant must submit a new application.
- 8.29 Licensing authorities may also 'defer' electronic applications once if the application is particularly complicated, for example if representations are received and a hearing is required. This allows the licensing authority to extend the statutory time period for the determination of the application by such time as is necessary, including, if required, arranging and holding a hearing. Licensing authorities must ensure that applicants are informed as quickly as possible of a decision to defer, and the reasons for the deferral, before the original 28 days has expired.

Written applications

8.30 A written application for a premises licence must be made in the prescribed form to the relevant licensing authority and be copied to each of the appropriate responsible authorities. For example, it would not be appropriate to send an application for premises

which was not a vessel to the Maritime and Coastguard Agency. The application must be accompanied by:

- the required fee (details of fees may be viewed on the GOV.UK website);
- an operating schedule (see below);
- a plan of the premises in a prescribed form; and
- if the application involves the supply of alcohol, a form of consent from the individual who is to be specified in the licence as the designated premises supervisor (DPS).
- 8.31 If the application is being made by an individual it should be accompanied by acceptable evidence of entitlement to work in the UK (this includes where the application is submitted electronically), as set out in the application form (see paragraph 4.9)
- 8.32 If the application is being made in respect of a community premises, it may be accompanied by the form of application to apply the alternative licence condition.
- 8.33 Guidance on completing premises licence, club premises certificate and minor variation forms can be found on the GOV.UK website. The Licensing Act 2003 (Premises licences and club premises certificates) Regulations 2005 contain provision about the prescribed form of applications, operating schedules and plans and are published on the legislation.gov.uk website.

Plans

8.34 Plans, for written and electronic applications, will not be required to be submitted in any particular scale, but they must be in a format which is "clear and legible in all material respects", i.e. they must be accessible and provides sufficient detail for the licensing authority to be able to determine the application, including the relative size of any features relevant to the application. There is no requirement for plans to be professionally drawn as long as they clearly show all the prescribed information.

Beer gardens or other outdoor spaces

- 8.35 Applicants will want to consider whether they might want to use a garden or other outdoor space as a location from which alcohol will be consumed. The sale of alcohol is to be treated as taking place where the alcohol is appropriated to the contract. In scenarios where drink orders are taken by a member of staff in the garden or outdoor space and the member of staff then collects the drinks from the licensed premises and returns to deliver them to the customer this would be treated as an off-sale and any conditions that relate to off-sales would apply.
- 8.36 In such cases it will not be necessary to include the garden or other outdoor space on the plan as part of the area covered by the premises licence. However, it will be necessary for the applicant to include the garden or other outdoor space on the plan as part of the area covered by the premises licence if the intention is to provide a service whereby drinks are available for sale and consumption directly from that area (i.e. the provision of on-sales). This would apply in the case of an outdoor bar or a service whereby a member of staff who is in the garden or outdoor space carries with them drinks that are available for sale (without the need for the staff member to return to the licensed premises to collect them).
- 8.37 If the beer garden or other outdoor area is to be used for the consumption of off-sales only, there is no requirement to show it on the plan of the premises, but the prescribed

application form requires the applicant to provide a description of where the place is and its proximity to the premises.

Entitlement to work in the UK

- 8.38 Individuals applying for a premises licence for the sale of alcohol or late night refreshment must be entitled to work in the UK. Licensing authorities must be satisfied that an individual who applies for a premises licence is entitled to work in the UK. This includes applications made by more than one individual applicant. An application made by an individual without the entitlement to work in the UK must be rejected. This applies to applications which include the sale of alcohol and the provisions of late night refreshment, but does not include applications which apply to regulated entertainment only. For example, a person applying for a licence for a music venue who does not intend to sell alcohol or late night refreshment is not prohibited from applying for a licence on grounds of immigration status. However, they will commit a criminal offence if they work illegally.
- 8.39 An applicant may demonstrate their right to work either by submitting documentation, or by an online right to work check. The documents which may be relied on in support of an application demonstrating an entitlement to work in the UK are the same as for personal licence applicants see paragraph 4.9. Alternatively, as for personal licences, applicants may demonstrate their right to work digitally by providing their share code and date of birth to enable the licensing authority to carry out a check with the Home Office online right to work checking service (available on GOV.UK: https://www.gov.uk/view-right-to-work) see paragraph 4.10. Where there is sufficient evidence that the applicant is not resident in the UK there is no requirement that the applicant has an entitlement to work in the UK. See paragraphs 4.21 to 4.48 in relation to entitlement to work in the UK for EEA citizens from 1 July 2021.
- 8.40 Where an applicant's permission to work in the UK is time-limited the licensing authority may issue a premises licence for an indefinite period, but the licence will become invalid when the immigration permission expires. The individual's entitlement to work in the UK may be extended or made permanent by the Home Office, and granting the licence for an indefinite period prevents the licensee from having to re-apply for a new licence. In the event that the Home Office cuts short or ends a person's immigration permission (referred to as a curtailment or revocation), any licence issued on or after 6 April 2017 which authorises the sale of alcohol or provision of late night refreshment will automatically lapse. As with personal licences, the licensing authority is under no duty to carry out on going immigration checks to see whether a licence holder's permission to be in the UK has been brought to an end. For further details on entitlement to work see paragraphs 4.8 to 4.48.

Steps to promote the licensing objectives

8.41 In completing an operating schedule, applicants are expected to have regard to the statement of licensing policy for their area. They must also be aware of the expectations of the licensing authority and the responsible authorities as to the steps that are appropriate for the promotion of the licensing objectives, and to demonstrate knowledge of their local area when describing the steps they propose to take to promote the licensing objectives. Licensing authorities and responsible authorities are expected to

publish information about what is meant by the promotion of the licensing objectives and to ensure that applicants can readily access advice about these matters. However, applicants are also expected to undertake their own enquiries about the area in which the premises are situated to inform the content of the application.

- 8.42 Applicants are, in particular, expected to obtain sufficient information to enable them to demonstrate, when setting out the steps they propose to take to promote the licensing objectives, that they understand:
 - the layout of the local area and physical environment including crime and disorder hotspots, proximity to residential premises and proximity to areas where children may congregate;
 - any risk posed to the local area by the applicants' proposed licensable activities; and
 - any local initiatives (for example, local crime reduction initiatives or voluntary schemes including local taxi-marshalling schemes, street pastors and other schemes) which may help to mitigate potential risks.
- 8.43 Applicants are expected to include positive proposals in their application on how they will manage any potential risks. Where specific policies apply in the area (for example, a cumulative impact assessment), applicants are also expected to demonstrate an understanding of how the policy impacts on their application; any measures they will take to mitigate the impact; and why they consider the application should be an exception to the policy.
- 8.44 It is expected that enquiries about the locality will assist applicants when determining the steps that are appropriate for the promotion of the licensing objectives. For example, premises with close proximity to residential premises should consider what effect this will have on their smoking, noise management and dispersal policies to ensure the promotion of the public nuisance objective. Applicants must consider all factors which may be relevant to the promotion of the licensing objectives, and where there are no known concerns, acknowledge this in their application.
- 8.45 The majority of information which applicants will require should be available in the licensing policy statement in the area. Other publicly available sources which may be of use to applicants include:
 - the Crime Mapping website;
 - Neighbourhood Statistics websites;
 - websites or publications by local responsible authorities;
 - websites or publications by local voluntary schemes and initiatives; and
 - on-line mapping tools.
- 8.46 While applicants are not required to seek the views of responsible authorities before formally submitting their application, they may find them to be a useful source of expert advice on local issues that should be taken into consideration when making an application. Licensing authorities may wish to encourage co-operation between applicants, responsible authorities and, where relevant, local residents and businesses before applications are submitted in order to minimise the scope for disputes to arise.
- 8.47 Applicants are expected to provide licensing authorities with sufficient information in this section to determine the extent to which their proposed steps are appropriate to

promote the licensing objectives in the local area. Applications must not be based on providing a set of standard conditions to promote the licensing objectives and applicants are expected to make it clear why the steps they are proposing are appropriate for the premises.

- 8.48 All parties are expected to work together in partnership to ensure that the licensing objectives are promoted collectively. Where there are no disputes, the steps that applicants propose to take to promote the licensing objectives, as set out in the operating schedule, will very often translate directly into conditions that will be attached to premises licences with the minimum of fuss.
- 8.49 For some premises, it is possible that no measures will be appropriate to promote one or more of the licensing objectives, for example, because they are adequately covered by other existing legislation. It is however important that all operating schedules should be precise and clear about the consideration given to the licensing objectives and any measures that are proposed to promote them.

Variations

Introduction

8.50 Where a premises licence holder wishes to amend the licence, the 2003 Act in most cases permits an application to vary to be made rather than requiring an application for a new premises licence. The process to be followed will depend on the nature of the variation and its potential impact on the licensing objectives. Applications to vary can be made electronically via GOV.UK or by means of the licensing authority's own electronic facility following the procedures set out above.

Simplified processes

- 8.51 There are simplified processes for making applications, or notifying changes, in the following cases:
 - a change of the name or address of someone named in the licence (section 33);
 - an application to vary the licence to specify a new individual as the designated premises supervisor (DPS) (section 37);
 - a request to be removed as the designated premises supervisor (section 41);
 - an application by a licence holder in relation to community premises authorised to sell alcohol to remove the usual mandatory conditions set out in sections 19(2) and 19(3) of the 2003 Act concerning the supervision of alcohol sales by a personal licence holder and the need for a DPS who holds a personal licence (sections 25A and 41D); and
 - an application for minor variation of a premises licence (sections 41A to 41C) or club premises certificate (sections 86A to 86C).
- 8.52 If an application to specify a new DPS or to remove the mandatory conditions concerning the supervision of alcohol sales is made electronically via GOV.UK or the licensing authority's own electronic facility, the authority must notify the police no later than the first working day after the application is given.
- 8.53 Where a simplified process requires the applicant (if they are not also the personal licence holder) to copy the application to the licence holder for information, this will apply regardless of whether the application is made in writing or electronically.

Otherwise the general guidance set out above (paragraphs 8.21 to 8.29) on electronic applications applies.

Minor variations process

- 8.54 Variations to premises licences or club premises certificates that could not impact adversely on the licensing objectives are subject to a simplified 'minor variations' process. Under this process, the applicant is not required to advertise the variation in a newspaper or circular, or copy it to responsible authorities. However, they must display it on a white notice (to distinguish it from the blue notice used for full variations and new applications). The notice must comply with the requirements set out in regulation 26A of the Licensing Act 2003 (Premises licences and club premises certificates) Regulations 2005. In accordance with those regulations, the notice must be displayed for a period of ten working days starting on the working day after the minor variation application was given to the licensing authority.
- 8.55 On receipt of an application for a minor variation, the licensing authority must consider whether the variation could impact adversely on the licensing objectives. It is recommended that decisions on minor variations should be delegated to licensing officers.
- 8.56 In considering the application, the licensing authority must consult relevant responsible authorities (whether the application is made in writing or electronically) if there is any doubt about the impact of the variation on the licensing objectives and they need specialist advice, and take their views into account in reaching a decision. The application is unlikely to be relevant to all responsible authorities.
- 8.57 The licensing authority must also consider any relevant representations received from other persons within the time limit referred to below. As stated earlier in this Guidance, representations are only relevant if they clearly relate to the likely effect of the grant of the variation on the promotion of at least one of the licensing objectives; representations must be confined to the subject matter of the variation. In the case of minor variations, there is no right to a hearing (as for a full variation or new application), but licensing authorities must take any representations into account in arriving at a decision.
- 8.58 Other persons have ten working days from the 'initial day', that is to say, the day after the application is received by the licensing authority, to submit representations. The licensing authority must therefore wait until this period has elapsed before determining the application, but must do so at the latest within 15 working days, beginning on the first working day after the authority received the application, with effect either that the minor variation is granted or the application is refused.
- 8.59 If the licensing authority fails to respond to the applicant within 15 working days (see section 193 of the 2003 Act for the definition of working day), the application will be treated as refused and the authority must return the fee to the applicant forthwith. However, the licensing authority and the applicant may agree instead that the undetermined application should be treated as a new application and that the fee originally submitted will be treated as a fee for the new application.
- 8.60 Where an application is refused and is then re-submitted through the full variation process, the full 28 day notification period will apply from the date the new application is received and applicants should advertise the application and copy it to all responsible authorities (in accordance with the regulations applicable to full variations).

8.61 Minor variations will generally fall into four categories: minor changes to the structure or layout of premises; small adjustments to licensing hours; the removal of out of date, irrelevant or unenforceable conditions or addition of volunteered conditions; and the addition of certain licensable activities. In all cases the overall test is whether the proposed variation could impact adversely on any of the four licensing objectives.

Changes to structure/layout

- 8.62 Many small variations to layout will have no adverse impact on the licensing objectives. However, changes to layout should be referred to the full variation process if they could potentially have an adverse impact on the promotion of the licensing objectives, for example by:
 - increasing the capacity for drinking on the premises;
 - affecting access between the public part of the premises and the rest of the premises or the street or public way, for instance, block emergency exits or routes to emergency exits; or
 - impeding the effective operation of a noise reduction measure such as an acoustic lobby.

Licensable Activities

- 8.63 Licensing authorities will also need to consider the combined effect of a series of applications for successive small layout changes (for example, as part of a rolling refurbishment of premises) which in themselves may not be significant, but which cumulatively may impact adversely on the licensing objectives. This emphasises the importance of having an up-to-date copy of the premises plan available.
- 8.64 An application to remove a licensable activity should normally be approved as a minor variation. Variations to add the sale by retail or supply of alcohol to a licence are excluded from the minor variations process and must be treated as full variations in all cases.
- 8.65 For other licensable activities, licensing authorities will need to consider each application on a case by case basis and in light of any licence conditions put forward by the applicant.

Licensing hours

- 8.66 Variations to the following are excluded from the minor variations process and must be treated as full variations in all cases:
 - to extend licensing hours for the sale or supply of alcohol for consumption on or off the premises between the hours of 23.00 and 07.00; or
 - to increase the amount of time on any day during which alcohol may be sold or supplied for consumption on or off the premises.
- 8.67 Applications to reduce licensing hours for the sale or supply of alcohol or, in some cases, to move (without increasing) the licensed hours between 07.00 and 23.00 will normally be processed as minor variations.
- 8.68 Applications to vary the time during which other licensable activities take place should be considered on a case-by-case basis with reference to the likely impact on the licensing objectives.

Licensing conditions

a) Imposed conditions

8.69 Licensing authorities cannot impose their own conditions on the licence through the minor variations process. If the licensing officer considers that the proposed variation would impact adversely on the licensing objectives unless conditions are imposed, they should refuse it.

b) Volunteered conditions

- 8.70 Applicants may volunteer conditions as part of the minor variation process. These conditions may arise from their own risk assessment of the variation, or from informal discussions with responsible authorities or the licensing authority.
- 8.71 For instance, there may be circumstances when the licence holder and a responsible authority such as the police or environmental health authority, agree that a new condition should be added to the licence (for example, that a nightclub adds the provision of door staff to its licence). Such a change would not normally impact adversely on the licensing objectives and could be expected to promote them by preventing crime and disorder or public nuisance. In these circumstances, the minor variation process may provide a less costly and onerous means of amending the licence than a review, with no risk to the licensing objectives. However, this route should only be used where the agreed variations are minor and the licence holder and the responsible authority have come to a genuine agreement. The licensing authority should be alive to any attempts to pressure licence or certificate holders into agreeing to new conditions where there is no evidence of a problem at the premises and, if there is any doubt, should discuss this with the relevant parties.

c) Amending or removing existing conditions

- 8.72 However, there may be some circumstances when the minor variation process is appropriate. Premises may change over time and the circumstances that originally led to the condition being attached or volunteered may no longer apply. For example, there may be no need for door supervision if a bar has been converted into a restaurant. Equally some embedded conditions may no longer apply.
- 8.73 Changes in legislation may invalidate certain conditions. Although the conditions do not have to be removed from the licence, licence holders and licensing authorities may agree that this is desirable to clarify the licence holder's legal obligations. There may also be cases where it is appropriate to revise the wording of a condition that is unclear or unenforceable. This would be acceptable as a minor variation as long as the purpose of the condition and its intended effect remain unchanged. Such a change could be expected to promote the licensing objectives by making it easier for the licence holder to understand and comply with the condition and easier for the licensing authority to enforce it.

Full variations process

- 8.74 Any other changes to the licence or certificate require an application to vary under sections 34 or 84 of the 2003 Act.
- 8.75 Licensing authorities may wish to consider whether there is any likely impact on the promotion of the licensing objectives in deciding whether there is a need for an

application to vary in relation to features which are not required to be shown on the plan under section 17 of the 2003 Act, but have nevertheless been included, for example, moveable furniture (altering the position of tables and chairs) or beer gardens (installation of a smoking shelter that will not affect the use of exits or escape routes).

- 8.76 However, it should be noted that a section 34 application cannot be used to vary a licence so as to:
 - extend a time limited licence; vary substantially the premises to which the licence relates;
 - transfer the licence from one holder to another; or
 - transfer the licence from one premises to another.
- 8.77 If an applicant wishes to make these types of changes to the premises licence, the applicant should make a new premises licence application under section 17 of the 2003 Act; or, to transfer the licence to another holder, an application under section 42 of the 2003 Act.

Relaxation of opening hours for local, national and international occasions

- 8.78 It should normally be possible for applicants for premises licences and club premises certificates to anticipate special occasions which occur regularly each year such as bank holidays and St. George's or St. Patrick's Day and to include appropriate opening hours in their operating schedules. Similarly, temporary event notices should be sufficient to cover other events which take place at premises that do not have a premises licence or club certificate.
- 8.79 However, exceptional events of local, national or international significance may arise which could not have been anticipated when the application was first made. In these circumstances, the Secretary of State may make a licensing hours order to allow premises to open for specified, generally extended, hours on these special occasions. This avoids the need for large numbers of applications to vary premises licences and club premises certificates. Typical events might include a one-off local festival or a Royal Jubilee.

Advertising applications

8.80 The requirements governing the advertisement of applications for the grant, variation or review of premises licences and club premises certificates are contained in Regulations 25 and 26 of the Licensing Act 2003 (Premises licences and club premises certificates) Regulations 2005 which are published on the Government's legislation website.

Applicants are required to:

- publish a notice in a local newspaper or, if there is none, in a local newsletter, circular or similar document circulating in the area in which the premises are situated; and
- display a brief summary of the application on an A4 (or larger) size notice, on pale blue paper in a prominent position (or positions) immediately on or outside the premises for at least 28 consecutive days (starting on the day after the day on which the application was given to the relevant licensing authority). The notice must be printed legibly in black ink or typed in black in size 16 font or larger.

- ensure that the above notices contain the name of the applicant, postal addresses of the premises (or if there is no postal address a description of the premises sufficient to enable the location to be identified), relevant licensing authority and the date by which any representations in relation to the application need to be made to the licensing authority. They should also contain a statement of the relevant licensable activities or relevant qualifying club activities that it is proposed will be carried on at the premises, or in the case of an application to vary a premises licence or a club premises certificate the notices shall briefly describe the proposed variation.
- 8.81 It is the responsibility of the applicant for putting the notice up, however licensing authorities should consider where the signs should be placed and advise the applicant where appropriate, to ensure people will see them, in particular if an application is likely to be of interest to the public. As prescribed in regulations, licensing authorities must also place a notice on their website outlining key details of the application as set out in regulations, including:
 - the name of the applicant or club;
 - the postal address of the premises or club premises;
 - the postal address and, where applicable, the internet address where the relevant licensing authority's register is kept and where and when the record of the application may be inspected;
 - the date by which representations from responsible authorities or other persons should be received and how these representations should be made; and
 - that it is an offence knowingly or recklessly to make a false statement in connection with an application and the maximum fine for which a person is liable on summary conviction for the offence.
- 8.82 The summary of the application should set out matters such as the proposed licensable activities and the proposed hours of opening and should be clearly displayed for the period during which representations may be made, together with information about where the details of the application may be viewed.
- 8.83 Licensing authorities in Wales should consider encouraging applicants to provide details in the alternative language (Welsh or English) to that of the main advertisement itself where the application may be viewed. Therefore, if an applicant publishes a notice in English they should be encouraged to provide a statement in Welsh as to where the application may be viewed, and vice versa. This would allow the reader of the notice to make enquiries to the licensing authority and find out the nature of the application.
- 8.84 Licensing authorities in Wales are also required to publish key information from licence applications in Welsh on their websites.
- 8.85 In the case of applications for premises licences involving internet or mail order sales, notices should be conspicuously displayed at the place where the alcohol is appropriated to the contract.
- 8.86 A vessel which is not permanently moored or berthed is treated as if it were a premises situated in a place where it is usually moored or berthed. The newspaper advertisement notice for such a vessel would need to be in relation to this place (where it is usually moored or berthed) and there is no provision requiring such advertising in other areas, for instance, if the vessel journeys through other licensing authority areas.

8.87 Arrangements should be put in place by the licensing authority for other parties to view a record of the application in the licensing register as described in Schedule 3 to the 2003 Act. Charges made for copies of the register should not exceed the cost of preparing such copies. Licensing authorities may wish to conduct random and unannounced visits to premises to confirm that notices have been clearly displayed and include relevant and accurate information.

Applications to change the designated premises supervisors

8.88 Chapter 4 covers designated premises supervisors and applications to vary a premises licence covering sales of alcohol by specifying a new designated premises supervisor. Chapter 4 covers applications by community premises to disapply the usual mandatory conditions in sections 19(2) and 19(3) of the 2003 Act concerning the authorisation of alcohol sales by a personal licence holder and the need for a designated premises supervisor who holds a personal licence.

Provisional statements

- 8.89 Where premises are being or are about to be constructed, extended or otherwise altered for the purpose of being used for one or more licensable activities, investors may be unwilling to commit funds unless they have some assurance that a premises licence covering the desired licensable activities would be granted for the premises when the building work is completed.
- 8.90 The 2003 Act does not define the words "otherwise altered", but the alteration must relate to the purpose of being used for one or more licensable activities.
- 8.91 Any person falling within section 16 of the 2003 Act can apply for a premises licence before new premises are constructed, extended or changed. This would be possible where clear plans of the proposed structure exist and the applicant is in a position to complete an operating schedule including details of:
 - the activities to take place there;
 - the time at which such activities will take place;
 - the proposed hours of opening;
 - where the applicant wishes the licence to have effect for a limited period, that period;
 - the steps to be taken to promote the licensing objectives; and
 - where the sale of alcohol is involved, whether supplies are proposed to be for consumption on or off the premises (or both) and the name of the designated premises supervisor the applicant wishes to specify.
- 8.92 In such cases, the licensing authority would include in the licence the date upon which it would come into effect. A provisional statement will normally only be required when the information described above is not available.
- 8.93 The 2003 Act therefore provides for a person, if an individual aged 18 or over, who has an interest in the premises to apply for a "provisional statement". This will not be time limited, but the longer the delay before an application for a premises licence is made, the more likely it is that there will be material changes and that the licensing authority will accept representations. "Person" in this context includes a business.
- 8.94 When a hearing is held, the licensing authority must decide whether, if the premises

were constructed or altered in the way proposed in the schedule of works and if a premises licence was sought for those premises, it would consider it appropriate for the promotion of the licensing objectives to:

- attach conditions to the licence;
- rule out any of the licensable activities applied for;
- · refuse to specify the person nominated as premises supervisor; or
- reject the application.

It will then issue the applicant with a provisional statement setting out the details of that decision together with its reasons.

- 8.95 The licensing authority must copy the provisional statement to each person who made relevant representations, and the chief officer of police for the area in which the premises is situated. The licensing authority should give full and comprehensive reasons for its decision. This is important in anticipation of an appeal by any aggrieved party.
- 8.96 When a person applies for a premises licence in respect of premises (or part of the premises or premises which are substantially the same) for which a provisional statement has been made, representations by responsible authorities and other persons will be excluded in certain circumstances. These are where:
 - the application for a licence is in the same form as the licence described in the provisional statement;
 - the work in the schedule of works has been satisfactorily completed;
 - given the information provided in the application for a provisional statement, the responsible authority or other person could have made the same, or substantially the same, representations about the application then but failed to do so without reasonable excuse; and
 - there has been no material change in the circumstances relating either to the premises or to the area in the proximity of those premises since the provisional statement was made.
- 8.97 Any decision of the licensing authority on an application for a provisional statement will not relieve an applicant of the need to apply for planning permission, building control approval of the building work, or in some cases both planning permission and building control.
- 8.98 A provisional statement may not be sought or given for a vessel, a vehicle or a moveable structure (see section 189 of the 2003 Act).

Transfers of premises licences

8.99 The 2003 Act provides for any person who may apply for a premises licence, which includes a business, to apply for a premises licence to be transferred to them. Where the application is made in writing, the applicant must give notice of the application to the chief officer of police in all cases, and the Home Office (Immigration Enforcement) if the licence authorises the sale of alcohol or provision of late night refreshment. Where it is made electronically via GOV.UK or the licensing authority's electronic facility, the licensing authority must notify the police and the Home Office (Immigration Enforcement) no later than the first working day after the application is given. However, the responsibility to notify the DPS remains with the applicant. Otherwise the general

guidance on electronic applications set out in paragraphs 8.21 to 8.29 applies.

- 8.100 In the vast majority of cases, it is expected that a transfer will be a very simple administrative process. Section 43 of the 2003 Act provides a mechanism which allows the transfer to come into immediate interim effect as soon as the licensing authority receives it, until it is formally determined or withdrawn. This is to ensure that there should be no interruption to normal business at the premises. If the police or the Home Office (Immigration Enforcement) raise no objection about the application, the licensing authority must transfer the licence in accordance with the application, amend the licence accordingly and return it to the new holder.
- 8.101 In exceptional circumstances where the chief officer of police believes the transfer may undermine the crime prevention objective, the police may object to the transfer. The Home Office (Immigration Enforcement) may object if it considers that granting the transfer would be prejudicial to the prevention of illegal working in licensed premises. Such objections are expected to be rare and arise because the police or the Home Office (Immigration Enforcement) have evidence that the business or individuals seeking to hold the licence, or businesses or individuals linked to such persons, are involved in crime (or disorder) or employing illegal workers.
- 8.102 Such objections (and therefore such hearings) should only arise in truly exceptional circumstances. If the licensing authority believes that the police or the Home Office (Immigration Enforcement) are using this mechanism to vet transfer applicants routinely and to seek hearings as a fishing expedition to inquire into applicants' backgrounds, it is expected that it would raise the matter immediately with the chief officer of police or the Home Office (Immigration Enforcement).

Interim authorities

- 8.103 The 2003 Act provides special arrangements for the continuation of permissions under a premises licence when the holder of a licence dies suddenly, becomes bankrupt, mentally incapable or ceases to be entitled to work in the UK. In the normal course of events, the licence would lapse in such circumstances. However, there may also be some time before, for example, the deceased person's estate can be dealt with or an administrative receiver appointed. This could have a damaging effect on those with interests in the premises, such as an owner, lessor or employees working at the premises in question; and could bring unnecessary disruption to customers' plans. The 2003 Act therefore provides for the licence to be capable of being reinstated in a discrete period of time in certain circumstances.
- 8.104 These circumstances arise only where a premises licence has lapsed owing to the death, incapacity or insolvency of the holder or where the holder ceases to be entitled to work in the UK. In such circumstances, an "interim authority" notice may be given to the licensing authority within 28 consecutive days beginning the day after the licence lapsed. Where applications are made in writing, the applicant must give notice of the application to the chief officer of police in all cases, and the Home Office (Immigration Enforcement) if the licence authorises the sale of alcohol or provision of late night refreshment. If an application is made electronically via GOV.UK or the licensing authority's electronic facility, the licensing authority must notify the police and the Home Office (Immigration Enforcement) no later than the first working day after the notice is

given.

- 8.105 An interim notice may only be given either by a person with a prescribed interest in the premises as set out in the regulations made under the 2003 Act (which may be viewed on <u>www.legislation.gov.uk</u>, the Government's legislation website); or by a person connected to the former holder of the licence (normally a personal representative of the former holder; or a person with power of attorney; or where someone has become insolvent, that person's insolvency practitioner). The person giving the interim authority notice must be entitled to work in the UK.
- 8.106 The effect of giving the notice is to reinstate the premises licence as if the person giving the notice is the holder of the licence and thereby allow licensable activities to continue to take place pending a formal application for transfer. The maximum period for which an interim authority notice may have effect is three months.
- 8.107 The interim authority notice ceases to have effect unless, by the end of the initial period of 28 consecutive days, a copy of the notice has been given to the chief officer of police and the Home Office (Immigration Enforcement). Within two working days of receiving the copy, and if satisfied that in the exceptional circumstances of the case failure to cancel the interim authority would undermine the crime prevention objective, the police may give a notice to that effect to the licensing authority. Similarly, the Home Office (Immigration Enforcement) may give a notice to the licensing authority if satisfied that the exceptional circumstances of the case are such that failure to cancel the interim authority would undermine the prevention of illegal working in licensed premises. In such circumstances, the licensing authority must hold a hearing to consider the objection notice and cancel the interim authority notice if it decides that it is appropriate to do so for the promotion of the crime prevention objective.
- 8.108 Licensing authorities should be alert to the need to consider the objection quickly. Under section 50 of the 2003 Act, where the premises licence lapses (because of death, incapacity or insolvency of the holder or because the holder is no longer entitled to work in the UK) or by its surrender, but no interim authority notice has effect, a person who may apply for the grant of a premises licence under section 16(1) may apply within 28 consecutive days of the lapse for the transfer of the licence to them with immediate effect pending the determination of the application. This will result in the licence being reinstated from the point at which the transfer application was received by the licensing authority. Where the application is made in writing, the person applying for the transfer must copy their application to the chief officer of police and the Home Office (Immigration Enforcement). If the application is made electronically the licensing authority must copy the application to the police and the Home Office (Immigration Enforcement).

Right of freeholders etc to be notified of licensing matters

8.109 A person (which will include a business or company) with a property interest in any premises situated in the licensing authority's area may give notice of their interest to the authority using a prescribed form and on payment of the relevant fee. The application may be made in writing or electronically via GOV.UK or the licensing authority's own facility, in which case the guidance at paragraphs 8.21 to 8.29 applies. Details of fees and forms are available on the GOV.UK website. It is entirely at the discretion of such persons whether they choose to register or not. It is not a legal requirement. Those who

may take advantage of this arrangement include the freeholder or leaseholder, a legal mortgagee in respect of the premises, a person in occupation of the premises or any other person prescribed by the Secretary of State.

8.110 The notice will have effect for 12 months but a new notice can be given every year. While the notice has effect, if any change relating to the premises concerned has been made to the licensing register (which the licensing authority has a duty to keep under section 8 of the 2003 Act), the licensing authority must notify the person who registered an interest of the matter to which the change relates. The person will also be notified of their right under section 8 to request a copy of the information contained in any entry in the register. In cases relating to interim authority notices (see above), it is important that such communications are dealt with promptly.

9. Determining applications

General

9.1 When a licensing authority receives an application for a new premises licence or an application to vary an existing premises licence, it must determine whether the application has been made in accordance with section 17 of the 2003 Act, and in accordance with regulations made under sections 17(3) to (6), 34, 42, 54 and 55 of the 2003 Act. It must similarly determine applications for the grant of club premises certificates made in accordance with section 71 of the 2003 Act, and in accordance with regulations 71(4) to (7), 84, 91 and 92 of the 2003 Act. This means that the licensing authority must consider among other things whether the application has been properly advertised in accordance with those regulations.

Where no representations are made

9.2 A hearing is not required where an application has been properly made and no responsible authority or other person has made a relevant representation or where representations are made and subsequently withdrawn. In these cases, the licensing authority must grant the application in the terms sought, subject only to conditions which are consistent with the operating schedule and relevant mandatory conditions under the 2003 Act. This should be undertaken as a simple administrative process by the licensing authority's officials who may replicate some of the proposals contained in the operating schedule to promote the licensing objectives in the form of clear and enforceable licence conditions. Licensing authorities should not hold hearings for uncontested applications, for example in situations where representations have been made and conditions have subsequently been agreed.

Where representations are made

9.3 Where a representation concerning the licensing objectives is made by a responsible authority about a proposed operating schedule and it is relevant (see paragraphs 9.4 to 9.10 below), the licensing authority's discretion will be engaged. It will also be engaged if another person makes relevant representations to the licensing authority, which are also not frivolous or vexatious (see paragraphs 9.4 to 9.10 below). Relevant representations can be made in opposition to, or in support of, an application and can be made by any individual, body or business that has grounds to do so.

Relevant, vexatious and frivolous representations

9.4 A representation is "relevant" if it relates to the likely effect of the grant of the licence on the promotion of at least one of the licensing objectives. For example, a representation from a local businessperson about the commercial damage caused by competition from new licensed premises would not be relevant. On the other hand, a representation by a businessperson that nuisance caused by new premises would deter customers from entering the local area, and the steps proposed by the applicant to prevent that nuisance were inadequate, would be relevant. In other words, representations should relate to the impact of licensable activities carried on from premises on the objectives. For representations in relation to variations to be relevant, they should be confined to

the subject matter of the variation. There is no requirement for a responsible authority or other person to produce a recorded history of problems at premises to support their representations, and in fact this would not be possible for new premises.

- 9.5 It is for the licensing authority to determine whether a representation (other than a representation from a responsible authority) is frivolous or vexatious on the basis of what might ordinarily be considered to be vexatious or frivolous. A representation may be considered to be vexatious if it appears to be intended to cause aggravation or annoyance, whether to a competitor or other person, without reasonable cause or justification. Vexatious circumstances may arise because of disputes between rival businesses and local knowledge will therefore be invaluable in considering such matters. Licensing authorities can consider the main effect of the representation, and whether any inconvenience or expense caused by it could reasonably be considered to be proportionate.
- 9.6 Frivolous representations would be essentially categorised by a lack of seriousness. Frivolous representations would concern issues which, at most, are minor and in relation to which no remedial steps would be warranted or proportionate.
- 9.7 Any person who is aggrieved by a rejection of their representations on either of these grounds may lodge a complaint through the local authority's corporate complaints procedure. A person may also challenge the authority's decision by way of judicial review.
- 9.8 Licensing authorities should not take decisions about whether representations are frivolous, vexatious or relevant to the licensing objectives on the basis of any political judgement. This may be difficult for councillors who receive complaints from residents within their own wards. If consideration is not to be delegated, contrary to the recommendation in this Guidance, an assessment should be prepared by officials for consideration by the sub- committee before any decision is taken that necessitates a hearing. Any councillor who considers that their own interests are such that they are unable to consider the matter independently should disqualify themselves.
- 9.9 It is recommended that, in borderline cases, the benefit of the doubt about any aspect of a representation should be given to the person making that representation. The subsequent hearing would then provide an opportunity for the person or body making the representation to amplify and clarify it.
- 9.10 Licensing authorities should consider providing advice on their websites about how any person can make representations to them.

The role of responsible authorities

9.11 Responsible authorities under the 2003 Act are automatically notified of all new applications. While all responsible authorities may make representations regarding applications for licences and club premises certificates and full variation applications, it is the responsibility of each responsible authority to determine when they have appropriate grounds to do so.

9.12 Each responsible authority will be an expert in their respective field, and in some cases it is likely that a particular responsible authority will be the licensing authority's main source of advice in relation to a particular licensing objective. For example, the police have a key role in managing the night-time economy and should have good working relationships with those operating in their local area⁵. The police should usually therefore be the licensing authority's main source of advice on matters relating to the promotion of the crime and disorder licensing objective. However, any responsible authority under the 2003 Act may make representations with regard to any of the licensing objectives if they have evidence to support such representations. Licensing authorities must therefore consider all relevant representations from responsible authorities carefully, even where the reason for a particular responsible authority's interest or expertise in the promotion of a particular objective may not be immediately apparent. However, it remains incumbent on all responsible authorities to ensure that their representations can withstand the scrutiny to which they would be subject at a hearing.

Licensing authorities acting as responsible authorities

- 9.13 Licensing authorities are included in the list of responsible authorities. A similar framework exists in the Gambling Act 2005. The 2003 Act does not require responsible authorities to make representations about applications for the grant of premises licences or to take any other steps in respect of different licensing processes. It is, therefore, for the licensing authority to determine when it considers it appropriate to act in its capacity as a responsible authority; the licensing authority should make this decision in accordance with its duties under section 4 of the 2003 Act.
- 9.14 Licensing authorities are not expected to act as responsible authorities on behalf of other parties (for example, local residents, local councillors or community groups) although there are occasions where the authority may decide to do so. Such parties can make relevant representations to the licensing authority in their own right, and it is reasonable for the licensing authority to expect them to make representations themselves where they are reasonably able to do so. However, if these parties have failed to take action and the licensing authority is aware of relevant grounds to make a representation, it may choose to act in its capacity as responsible authority.
- 9.15 It is also reasonable for licensing authorities to expect that other responsible authorities should intervene where the basis for the intervention falls within the remit of that other responsible authority. For example, the police should make representations where the representations are based on concerns about crime and disorder. Likewise, it is reasonable to expect the local authority exercising environmental health functions to make representations where there are concerns about noise nuisance. Each responsible authority has equal standing under the 2003 Act and may act independently without waiting for representations from any other responsible authority.
- 9.16 The 2003 Act enables licensing authorities to act as responsible authorities as a means of early intervention; they may do so where they consider it appropriate without having

⁵ Police and Crime Commissioners are expected to have a central role working in partnership with local authorities, enforcement bodies and other local partners to decide on what action is needed to tackle alcohol- related crime and disorder in their areas. However, the Chief Officer of Police remains the named responsible authority under the 2003 Act.

to wait for representations from other responsible authorities. For example, the licensing authority may (in a case where it has published a cumulative impact assessment) consider that granting a new licence application will add to the cumulative impact of licensed premises in its area and therefore decide to make representations to that effect, without waiting for any other person to do so.

- 9.17 In cases where a licensing authority is also acting as responsible authority in relation to the same process, it is important to achieve a separation of responsibilities within the authority to ensure procedural fairness and eliminate conflicts of interest. In such cases licensing determinations will be made by the licensing committee or sub committee comprising elected members of the authority (although they are advised by a licensing officer). Therefore, a separation is achieved by allocating distinct functions (i.e. those of licensing authority and responsible authority) to different officials within the authority.
- 9.18 In these cases, licensing authorities should allocate the different responsibilities to different licensing officers or other officers within the local authority to ensure a proper separation of responsibilities. The officer advising the licensing committee (i.e. the authority acting in its capacity as the licensing authority) must be a different person from the officer who is acting for the responsible authority. The officer acting for the responsible authority should not be involved in the licensing decision process and should not discuss the merits of the case with those involved in making the determination by the licensing authority. For example, discussion should not take place between the officer acting as responsible authority and the officer handling the licence application regarding the merits of the case. Communication between these officers in relation to the case should remain professional and consistent with communication with other responsible authorities. Representations, subject to limited exceptions, must be made in writing. It is for the licensing authority to determine how the separate roles are divided to ensure an appropriate separation of responsibilities. This approach may not be appropriate for all licensing authorities and many authorities may already have processes in place to effectively achieve the same outcome.
- 9.19 Smaller licensing authorities, where such a separation of responsibilities is more difficult, may wish to involve officials from outside the licensing department to ensure a separation of responsibilities. However, these officials should still be officials employed by the authority.

Health bodies acting as responsible authorities

- 9.20 Where a local authority's Director of Public Health in England (DPH)⁶ or Local Health Board (LHB) (in Wales) exercises its functions as a responsible authority, it should have sufficient knowledge of the licensing policy and health issues to ensure it is able to fulfil those functions. If the authority wishes to make representations, the DPH or LHB will need to decide how best to gather and coordinate evidence from other bodies which exercise health functions in the area, such as emergency departments and ambulance services.
- 9.21 Health bodies may hold information which other responsible authorities do not, but

⁶ This change was made as a result of the commencement of measures in the Health and Social Care Act 2012 which amended the 2003 Act and further provision in the NHS Bodies and Local Authorities (Partnership Arrangements, Care Trusts, Public Health and Local Healthwatch) Regulations 2012.

which would assist a licensing authority in exercising its functions. This information may be used by the health body to make representations in its own right or to support representations by other responsible authorities, such as the police. Such representations can potentially be made on the grounds of all four licensing objectives. Perhaps the most obvious example is where drunkenness leads to accidents and injuries from violence, resulting in attendances at emergency departments and the use of ambulance services. Some of these incidents will be reported to the police, but many will not. Such information will often be relevant to the public safety and crime and disorder objectives.

- 9.22 However, health bodies are encouraged to make representations in respect of any of the four licensing objectives without necessarily seeking views from other responsible authorities where they have appropriate evidence to do so. There is also potential for health bodies to participate in the licensing process in relation to the protection of children from harm. This objective not only concerns the physical safety of children, but also their moral and psychological well being.
- 9.23 Evidence relating to under 18s alcohol-related emergency department attendance, hospital admissions and underage sales of alcohol, could potentially have implications for both the protection of children from harm and the crime and disorder objectives. Health bodies can provide evidence to lead or support representations in relation to this objective. In relation to proxy purchases, data collected by health bodies could be used to inform other responsible authorities, including the police and licensing authorities, about a prevalence of proxy purchasing in a particular area. For example, the police could use this data to tackle instances of 'shoulder tapping' (where under 18s approach adults to buy alcohol on their behalf) and to suggest measures which retailers might be able to take to ensure, as far as possible, that they are not knowingly selling alcohol to an adult who is buying on behalf of a person aged under 18. Although less obvious, health bodies may also have a role to play in the prevention of public nuisance where its effect is prejudicial to health and where they hold relevant data.
- 9.24 DPHs and LHBs will need to consider how to collect anonymised information about incidents that relate to specific premises or premises in a particular area (for example, an area which is the subject of a cumulative impact assessment). Many areas have already developed procedures for local information sharing to tackle violence, which could provide useful evidence to support representations. The College of Emergency Medicine has issued guidelines for information sharing to reduce community violence which recommends that data about assault victims should be collected upon admission to emergency departments, including the date, time and location of the assault i.e. the name of the pub, club or street where the incident occurred. Sometimes, it may be possible to link ambulance callouts or attendances at emergency departments to irresponsible practices at specific premises, such as serving alcohol to people who are intoxicated or targeting promotions involving unlimited or unspecified quantities of alcohol at particular groups.

Home Office Immigration Enforcement acting as a responsible authority

9.25 The Immigration Act 2016 made the Secretary of State a responsible authority in respect of premises licensed to sell alcohol or late night refreshment with effect from 6 April 2017. In effect this conveys the role of responsible authority to Home Office Immigration Enforcement who exercises the powers on the Secretary of State's behalf. When Immigration Enforcement exercises its powers as a responsible authority it will do so in respect of the prevention of crime and disorder licensing objective because it is concerned with the prevention of illegal working or immigration offences more broadly.

Disclosure of personal details of persons making representations

- 9.26 Where a notice of a hearing is given to an applicant, the licensing authority is required under the Licensing Act 2003 (Hearings) Regulations 2005 to provide the applicant with copies of the relevant representations that have been made.
- 9.27 In exceptional circumstances, persons making representations to the licensing authority may be reluctant to do so because of fears of intimidation or violence if their personal details, such as name and address, are divulged to the applicant.
- 9.28 Where licensing authorities consider that the person has a genuine and well-founded fear of intimidation and may be deterred from making a representation on this basis, they may wish to consider alternative approaches.
- 9.29 For instance, they could advise the persons to provide the relevant responsible authority with details of how they consider that the licensing objectives are being undermined so that the responsible authority can make representations if appropriate and justified.
- 9.30 The licensing authority may also decide to withhold some or all of the person's personal details from the applicant, giving only minimal details (such as street name or general location within a street). However, withholding such details should only be considered where the circumstances justify such action.

Hearings

9.31 The Licensing Act 2003 (Hearings) Regulations 2005 governing hearings may be found on the www.legislation.gov.uk website. If the licensing authority decides that representations are relevant, it must hold a hearing to consider them. The need for a hearing can only be avoided with the agreement of the licensing authority, where the applicant and all of the persons who made relevant representations have given notice to the authority that they consider a hearing to be unnecessary. Where this is the case and the authority agrees that a hearing is unnecessary, it must forthwith give notice to the parties that the hearing has been dispensed with. Notwithstanding those regulatory provisions, in cases where the licensing authority believes that a hearing is still necessary, it is recommended that the authority should, as soon as possible, provide the parties with reasons in writing for the need to hold the hearing. In cases where only 'positive' representations are received, without gualifications, the licensing authority should consider whether a hearing is required. To this end, it may wish to notify the persons who made representations and give them the opportunity to withdraw those representations. This would need to be done in sufficient time before the hearing to ensure that parties were not put to unnecessary inconvenience.

- 9.32 Responsible authorities should try to conclude any discussions with the applicant in good time before the hearing. The 2005 Hearings Regulations permit licensing authorities to extend a time limit provided for by those Regulations for a specified period where it considers this to be necessary in the public interest. For example, if the application is amended at the last moment, the licensing committee should consider giving other persons time to address the revised application before the hearing commences. Where the authority has extended a time limit it must forthwith give a notice to the parties involved stating the period of the extension and the reasons for it.
- 9.33 The 2005 Hearings Regulations require that representations must be withdrawn 24 hours before the first day of any hearing. If they are withdrawn after this time, the hearing must proceed and the representations may be withdrawn orally at that hearing. However, where discussions between an applicant and those making representations are taking place and it is likely that all parties are on the point of reaching agreement, the licensing authority may wish to use the power given within the hearings regulations to extend time limits, if it considers this to be in the public interest.
- 9.34 Applicants should be encouraged to contact responsible authorities and others, such as local residents, who may be affected by the application before formulating their applications so that the mediation process may begin before the statutory time limits come into effect after submission of an application. The hearing process must meet the requirements of regulations made under the 2003 Act. Where matters arise which are not covered by the regulations, licensing authorities may make arrangements as they see fit as long as they are lawful.
- 9.35 There is no requirement in the 2003 Act for responsible authorities that have made representations to attend, but it is generally good practice and assists committees in reaching more informed decisions. Where several responsible authorities within a local authority have made representations on an application, a single local authority officer may represent them at the hearing if the responsible authorities and the licensing authority agree. This local authority officer representing other responsible authorities may be a licensing officer, but only if this licensing officer is acting as a responsible authority on behalf of the licensing authority and has had no role in the licensing determination process. This is to ensure that the responsible authorities are represented by an independent officer separate from the licensing determination process.
- 9.36 As noted in paragraphs 9.13 to 9.19 above, where the licensing officer is acting as a responsible authority the relevant steps should be followed to ensure that this individual has no role in the decision making process regarding the licensing determination.
- 9.37 As a matter of practice, licensing authorities should seek to focus the hearing on the steps considered appropriate to promote the particular licensing objective or objectives that have given rise to the specific representation and avoid straying into undisputed areas. A responsible authority or other person may choose to rely on their written representation. They may not add further representations to those disclosed to the applicant prior to the hearing, but they may expand on their existing representation and should be allowed sufficient time to do so, within reasonable and practicable limits.
- 9.38 In determining the application with a view to promoting the licensing objectives in the overall interests of the local community, the licensing authority must give appropriate weight to:
- the steps that are appropriate to promote the licensing objectives;
- the representations (including supporting information) presented by all the parties;
- this Guidance;
- its own statement of licensing policy.
- 9.39 The licensing authority should give its decision within five working days of the conclusion of the hearing (or immediately in certain specified cases) and provide reasons to support it. This will be important if there is an appeal by any of the parties. Notification of a decision must be accompanied by information on the right of the party to appeal. After considering all the relevant issues, the licensing authority may grant the application subject to the relevant mandatory conditions and any conditions that are consistent with the operating schedule. Any additional conditions imposed must be appropriate for the promotion of the licensing objectives; there is no power for the licensing authority to attach a condition that is merely aspirational. For example, conditions may not be attached which relate solely to the health of customers rather than their direct physical safety. Any conditions added to the licence must be those imposed at the hearing or those agreed when a hearing has not been necessary.
- 9.40 Alternatively, the licensing authority may refuse the application on the grounds that this is appropriate for the promotion of the licensing objectives. It may also refuse to specify a designated premises supervisor and/or only allow certain requested licensable activities. In the interests of transparency, the licensing authority should publish hearings procedures in full on its website to ensure that those involved have the most current information.
- 9.41 In the context of variations or minor variations, which may involve structural alteration to or change of use of a building, the decision of the licensing authority will not exempt an applicant from the need to apply for building control approval, planning permission or both of these where appropriate.

Determining actions that are appropriate for the promotion of the licensing objectives

- 9.42 Licensing authorities are best placed to determine what actions are appropriate for the promotion of the licensing objectives in their areas. All licensing determinations should be considered on a case-by-case basis. They should take into account any representations or objections that have been received from responsible authorities or other persons, and representations made by the applicant or premises user as the case may be.
- 9.43 The authority's determination should be evidence-based, justified as being appropriate for the promotion of the licensing objectives and proportionate to what it is intended to achieve.
- 9.44 Determination of whether an action or step is appropriate for the promotion of the licensing objectives requires an assessment of what action or step would be suitable to achieve that end. While this does not therefore require a licensing authority to decide that no lesser step will achieve the aim, the authority should aim to consider the potential burden that any condition would impose on the premises licence holder (such as the financial burden due to restrictions on licensable activities) as well as the potential benefit in terms of the promotion of the licensing objectives. However, it is

imperative that the authority ensures that the factors which form the basis of its determination are limited to consideration of the promotion of the objectives and nothing outside those parameters. As with the consideration of licence variations, the licensing authority should consider wider issues such as other conditions already in place to mitigate potential negative impact on the promotion of the licensing objectives and the track record of the business. Further advice on determining what is appropriate when imposing conditions on a licence or certificate is provided in Chapter 10. The licensing authority is expected to come to its determination based on an assessment of the evidence on both the risks and benefits either for or against making the determination.

Considering cases where licensing and planning applications are made simultaneously

9.45 Where businesses have indicated, when applying for a licence under the 2003 Act, that they have also applied for planning permission or that they intend to do so, licensing committees and officers should consider discussion with their planning counterparts prior to determination with the aim of agreeing mutually acceptable operating hours and scheme designs.

10. Conditions attached to premises licences and club premises certificates

General

- 10.1 This chapter provides further guidance in relation to conditions attached to premises licences and club premises certificates. General principles on licence conditions are set out in Chapter 1 (see paragraph 1.16).
- 10.2 Conditions include any limitations or restrictions attached to a licence or certificate and essentially are the steps or actions that the holder of the premises licence or the club premises certificate will be required to take or refrain from taking in relation to the carrying on of licensable activities at the premises in question. Failure to comply with any condition attached to a licence or certificate is a criminal offence, which on conviction is punishable by an unlimited fine or up to six months' imprisonment. The courts have made clear that it is particularly important that conditions which are imprecise or difficult for a licence holder to observe should be avoided.
- 10.3 There are three types of condition that may be attached to a licence or certificate: proposed, imposed and mandatory. Each of these categories is described in more detail below.

Proposed conditions

- 10.4 The conditions that are appropriate for the promotion of the licensing objectives should emerge initially from the risk assessment carried out by a prospective licence or certificate holder, which they should carry out before making their application for a premises licence or club premises certificate. This would be translated into the steps recorded in the operating schedule or club operating schedule, which must also set out the proposed hours during which licensable activities will be conducted and any other hours during which the premises will be open to the public.
- 10.5 It is not acceptable for licensing authorities to simply replicate the wording from an applicant's operating schedule. A condition should be interpreted in accordance with the applicant's intention and be appropriate and proportionate for the promotion of the licensing objectives.

Consistency with steps described in operating schedule

- 10.6 The 2003 Act provides that where an operating schedule or club operating schedule has been submitted with an application and there have been no relevant representations made by responsible authorities or any other person, the licence or certificate must be granted subject only to such conditions as are consistent with the schedule accompanying the application and any mandatory conditions required under the 2003 Act.
- 10.7 Consistency means that the effect of the condition should be substantially the same as that intended by the terms of the operating schedule. If conditions are broken, this may lead to a criminal prosecution or an application for a review and it is extremely important therefore that they should be expressed on the licence or certificate in unequivocal and unambiguous terms. The duty imposed by conditions on the licence holder or club must

be clear to the licence holder, club, enforcement officers and the courts.

Imposed conditions

- 10.8 The licensing authority may not impose any conditions unless its discretion has been exercised following receipt of relevant representations and it is satisfied as a result of a hearing (unless all parties agree a hearing is not necessary) that it is appropriate to impose conditions to promote one or more of the four licensing objectives. In order to promote the crime prevention licensing objective conditions may be included that are aimed at preventing illegal working in licensed premises. This provision also applies to minor variations.
- 10.9 It is possible that in some cases no additional conditions will be appropriate to promote the licensing objectives.

Proportionality

The 2003 Act requires that licensing conditions should be tailored to the size, type, 10.10 location and characteristics and activities taking place at the premises concerned. Conditions should be determined on a case-by-case basis and standardised conditions which ignore these individual aspects should be avoided. For example, conditions should not be used to implement a general policy in a given area such as the use of CCTV, polycarbonate drinking vessels or identity scanners where they would not be appropriate to the specific premises. Conditions that are considered appropriate for the prevention of illegal working in premises licensed to sell alcohol or late night refreshment might include requiring a premises licence holder to undertake right to work checks on all staff employed at the licensed premises or requiring that evidence of a right to work check, either physical or digital (e.g. copy of any document checked or a clear copy of the online right to work check) is retained at the licensed premises. Licensing authorities may also wish to consider placing additional conditions on licences to safeguard patrons against spiking, if deemed appropriate and proportionate for a specific venue where there is evidence to justify such action (a definition of spiking can be found in para 2.7). Licensing authorities and other responsible authorities should be alive to the indirect costs that can arise because of conditions. These could be a deterrent to holding events that are valuable to the community or for the funding of good and important causes. Licensing authorities should therefore ensure that any conditions they impose are only those which are appropriate for the promotion of the licensing objectives.

Naming, packing and promotion in retail premises

- 10.11 The Government acknowledges that the irresponsible naming, packing or promotion of alcoholic drinks may contribute to alcohol related harms. Where there is direct evidence of specific incidents of irresponsible naming, packing or promotion of alcoholic drinks linked to the undermining of one of the licensing objectives, licensing authorities should, in the exercise of their licensing functions (in particular, in relation to an application for the grant, variation or review of a premises licence), consider whether it is appropriate to impose conditions on licences that require the licence holder to comply with the Portman Group's Retailer Alert Bulletins. This condition should be considered on a case by case basis and in the context of the promotion of the licensing objectives.
- 10.12 The Portman Group operates, on behalf of the alcohol industry, a Code of Practice on

the Naming, Packaging and Promotion of Alcoholic Drinks. The Code seeks to ensure that drinks are packaged and promoted in a socially responsible manner and only to those who are 18 years old or older. Complaints about products under the Code are considered by an Independent Complaints Panel and the Panel's decisions are published on the Portman Group's website, in the trade press and in an annual report. If a product's packaging or point-of-sale advertising is found to be in breach of the Code, the Portman Group may issue a Retailer Alert Bulletin to notify retailers of the decision and ask them not to replenish stocks of any such product or to display such point-ofsale material, until there has been compliance with the decision.

Hours of trading

- 10.13 The Government acknowledges that different licensing strategies may be appropriate for the promotion of the licensing objectives in different areas. The 2003 Act gives the licensing authority power to make decisions about the hours during which premises can conduct licensable activities as part of the implementation of its licensing policy statement. Licensing authorities are best placed to make decisions about appropriate opening hours in their areas based on their local knowledge and in consultation with responsible authorities. However, licensing authorities must always consider each application and must not impose predetermined licensed opening hours, without giving individual consideration to the merits of each application.
- 10.14 Where there are objections to an application to extend the hours during which licensable activities are to be carried on and the licensing authority determines that this would undermine the licensing objectives, it may reject the application or grant it with appropriate conditions and/or different hours from those requested.
- 10.15 Shops, stores and supermarkets should normally be free to provide sales of alcohol for consumption off the premises at any times when the retail outlet is open for shopping unless there are good reasons, based on the licensing objectives, for restricting those hours.

The performance of plays

10.16 The 2003 Act provides that other than for the purposes of public safety, conditions must not be attached to premises licences or club premises certificates authorising the performance of a play⁷ which attempt to censor or modify the content of plays in any way. Any such condition would be ultra vires the 2003 Act.

Censorship

10.17 In general, other than in the context of film classification for film exhibitions, licensing authorities should not use their powers under the 2003 Act to seek to impose conditions which censor the content of any form of regulated entertainment. This is not a proper function of licensing law and cannot be properly related to the licensing objectives. The content of regulated entertainment is a matter which is addressed by existing laws governing indecency and obscenity. Where the concern is about protecting children, their access should be restricted where appropriate. But no other limitation should normally be imposed.

⁷ See chapter 16 for when a performance of a play is licensable.

Major festivals and carnivals

- 10.18 Licensing authorities should publicise the need for the organisers of major festivals and carnivals to approach them at the earliest opportunity to discuss arrangements for licensing activities falling under the 2003 Act. For some events, the organisers may seek a single premises licence to cover a wide range of activities at varied locations within the premises. This would involve the preparation of a substantial operating schedule, and licensing authorities should offer advice and assistance about its preparation.
- 10.19 For other events, applications for many connected premises licences may be made which in combination will represent a single festival. It is important that licensing authorities should publicise the need for proper co-ordination of such arrangements and will need to ensure that responsible authorities are aware of the connected nature of the individual applications.
- 10.20 Local authorities should bear in mind their ability to seek premises licences from the licensing authority for land or buildings under public ownership within the community in their own name.⁸ This could include, for example, village greens, market squares, promenades, community halls, local authority owned art centres and similar public areas where festivals and carnivals might take place.⁹ Performers and entertainers would then have no need to obtain a licence or give a temporary event notice themselves to enable them to give performances in these places, although they would need the permission of the local authority to put on the event.

Fixed prices

- 10.21 Licensing authorities should not attach standardised blanket conditions relating to fixed prices for alcoholic drinks to premises licences or club licences or club premises certificates in an area. This may be unlawful under current law. However, it is important to note that the mandatory conditions made under sections 19A and 73B of the 2003 Act prohibit a number of types of drinks promotions including where they give rise to a significant risk to any one of the four licensing objectives; the mandatory conditions also prohibit the sale of alcohol below the permitted price, as defined in paragraph 10.56.
- 10.22 Where licensing authorities are asked by the police, other responsible authorities or other persons to impose restrictions on promotions in addition to those restricted by the mandatory conditions, they should consider each application on its individual merits, tailoring any conditions carefully to cover only irresponsible promotions in the particular and individual circumstances of any premises where these are appropriate for the promotion of the licensing objectives. In addition, when considering any relevant representations which demonstrate a clear causal link between sales promotions or price discounting and levels of crime and disorder on or near the premises, it would be appropriate for the licensing authority to consider the imposition of a new condition prohibiting irresponsible sales promotions or the discounting of prices of alcoholic beverages at those premises. However, before pursuing any form of restrictions at all,

⁸ In some circumstances, no licence is required for any entertainment provided by or on behalf of a local authority, see paragraphs 16.15-16.19

⁹ The register of public spaces: <u>https://www.gov.uk/government/publications/licensed-spaces-register</u>

licensing authorities should take their own legal advice.

Large capacity venues used exclusively or primarily for the "vertical" consumption of alcohol (HVVDs)

- 10.23 Large capacity "vertical drinking" premises, sometimes called High Volume Vertical Drinking establishments (HVVDs), are premises with exceptionally high capacities, which are used primarily or exclusively for the sale and consumption of alcohol, and have little or no seating for patrons. Previous research has demonstrated that the environment within such establishments can have a significant bearing on the likelihood of crime and disorder.
- 10.24 Where appropriate, conditions can be attached to premises licences for the promotion of the prevention of crime and disorder at such premises that require the premises to observe:
 - a prescribed capacity;
 - an appropriate ratio of tables and chairs to customers based on the capacity; and
 - a requirement that security staff holding the appropriate SIA licence or exemption are present to control entry for the purpose of compliance with the capacity limit and to deny entry to individuals who appear drunk or disorderly or both.

Mandatory conditions in relation to the supply of alcohol

10.25 The 2003 Act provides for the following mandatory conditions to be included in every licence and/or club premises certificate in the circumstances specified.

Designated premises supervisor

- 10.26 The 2003 Act provides that, where a premises licence authorises the supply of alcohol, it must include a condition that no supply of alcohol may be made at a time when no designated premises supervisor has been specified in the licence or at a time when the designated premises supervisor does not hold a personal licence or the personal licence has been suspended.
- 10.27 The main purpose of the 'designated premises supervisor' as defined in the 2003 Act is to ensure that there is always one specified individual among these personal licence holders who can be readily identified for the premises where a premises licence is in force. That person will normally have been given day to day responsibility for running the premises by the premises licence holder. The requirements set out in relation to the designated premises supervisor and authorisation of alcohol sales by a personal licence holder do not apply to community premises in respect of which a successful application has been made to disapply the usual mandatory conditions in sections 19(2) and 19(3) of the 2003 Act (see Chapter 4 of this Guidance).
- 10.28 The 2003 Act does not require a designated premises supervisor or any other personal licence holder to be present on the premises at all times when alcohol is sold. However, the designated premises supervisor and the premises licence holder remain responsible for the premises at all times including compliance with the terms of the 2003 Act and conditions attached to the premises licence to promote the licensing objectives.

Authorisation by personal licence holders

- 10.29 In addition, every premises licence that authorises the sale of alcohol must require that every supply of alcohol under the premises licence must be made or authorised by a person who holds a personal licence. This in most instances will be the designated premises supervisor who must hold a valid personal licence. Any premises at which alcohol is sold or supplied may employ one or more personal licence holders. This does not mean that the condition requires the presence of the designated premises supervisor or any other personal licence holder on the premises at all times.
- 10.30 Similarly, the fact that every supply of alcohol must be made under the authority of a personal licence holder does not mean that only personal licence holders can make sales or that they must be personally present at every transaction. A personal licence holder may authorise members of staff to make sales of alcohol but may be absent at times from the premises when a transaction takes place. However, the responsible personal licence holder may not be able to escape responsibility for the actions of anyone authorised to make sales.
- 10.31 "Authorisation" does not imply direct supervision by a personal licence holder of each sale of alcohol. The question arises as to how sales can be authorised. Ultimately, whether an authorisation has been given is a question of fact that would have to be decided by the courts on the evidence before it in the course of a criminal prosecution.
- 10.32 The following factors should be relevant in considering whether or not an authorisation has been given:
 - the person(s) authorised to sell alcohol at any particular premises should be clearly identified;
 - the authorisation should have specified the acts which may be carried out by the person who is authorised to supply alcohol;
 - there should be an overt act of authorisation, for example, a specific written statement given to the individual who is authorised to supply alcohol; and
 - there should be in place sensible arrangements for the personal licence holder to monitor the activity that they have authorised on a reasonably regular basis.
- 10.33 It is strongly recommended that personal licence holders give specific written authorisations to individuals whom they are authorising to retail alcohol. A single written authorisation would be sufficient to cover multiple sales over an unlimited period. This would assist personal licence holders in demonstrating due diligence should issues arise with enforcement authorities; and would protect employees if they themselves are challenged in respect of their authority to sell alcohol.
- 10.34 Written authorisation is not a requirement of the 2003 Act and its absence alone could not give rise to enforcement action.
- 10.35 It must be remembered that while the designated premises supervisor or a personal licence holder may authorise other individuals to sell alcohol in their absence, they are responsible for any sales that may be made. Similarly, the premises licence holder remains responsible for ensuring that licensing law and licence conditions are observed at the premises.

Arrangements for the mandatory licence conditions

- 10.36 The mandatory conditions made under sections 19A and 73B of the 2003 Act (the conditions governing irresponsible promotions, dispensing alcohol directly into the mouth, provision of free tap water, age verification, small measures and the prohibition on sales of alcohol below the permitted price) do not have to be physically included in the licence or certificate but nonetheless will apply to every licence and certificate authorising the sale and supply of alcohol for consumption on the premises. The mandatory conditions set out in section 19 of the 2003 Act (the requirement for a DPS and for all sales to be made or authorised by a personal licence holder) do, however, have to be physically included in the licence. The mandatory licence conditions do not apply to activities (including the supply of alcohol) authorised by a temporary event notice.
- 10.37 Whereas the initial mandatory conditions in section 19 of the 2003 Act are set out in Annex 1 of the licence, the additional mandatory conditions made under section 19A of the 2003 Act are treated as if they were included in existing licences and certificates on the date that those conditions came into force.
- 10.38 Following their commencement, the mandatory conditions overrode any pre-existing conditions already included in a licence or certificate insofar as the mandatory conditions were identical to, or inconsistent with or more onerous than, any pre-existing conditions. It is not necessary to record on the face of existing licences and certificates the impact that the introduction of the mandatory conditions has had on pre-existing conditions.

Irresponsible promotions

10.39 Under this condition, the "responsible person" (defined in the 2003 Act as the holder of a premises licence, designated premises supervisor, a person aged 18 or over who is authorised to allow the sale or supply of alcohol by an under 18 or a member or officer of a club present on the club premises who can oversee the supply of alcohol) should be able to demonstrate that they have ensured that staff do not carry out, arrange or participate in any irresponsible promotions. An irresponsible promotion is one that fits one of the descriptions below (or is substantially similar), is carried on for the purpose of encouraging the sale or supply of alcohol for consumption on the premises. The aim of the condition is to prohibit or restrict promotions which encourage people to drink more than they might ordinarily do and in a manner which undermines the licensing objectives.

Drinking games

10.40 Drinking games which require or encourage individuals to drink a quantity of alcohol within a time limit, or drink as much alcohol as possible within a time limit or otherwise, are prohibited. For example, this may include organised 'drink downing' competitions. This would not prevent the responsible person from requiring all drinks to be consumed or abandoned at, or before, the closing time of the premises. Nor does it necessarily prohibit 'happy hours' as long as these are not designed to encourage individuals to drink excessively or rapidly.

Large quantities of alcohol for free or a fixed price

10.41 Irresponsible promotions can include the provision of unlimited or unspecified quantities of alcohol free or for a fixed or discounted price, where there is a significant risk that such a promotion would undermine one or more of the licensing objectives. This includes alcohol provided to the public or to a group defined by a particular characteristic, for example, a promotion which offers women free drinks before a certain time or "all you can drink for £10". Promotions can be designed with a particular group in mind (for example, over 65s). A common sense approach is encouraged, which may include specifying the quantity of alcohol included in it or not targeting a group which could become more vulnerable or present a greater risk of crime and disorder as a result of excessive alcohol consumption.

Prizes and rewards

10.42 The sale, supply or provision of free or discounted alcohol or any other item as a prize to encourage or reward the purchase and consumption of alcohol can be within the definition of an irresponsible promotion, where there is a significant risk that such a promotion would undermine one or more of the licensing objectives. This may include promotions under which free or discounted alcohol is offered as a part of the sale of alcohol, for example, "Buy one and get two free" and "Buy one cocktail and get a second cocktail for 25p". This includes promotions which involve the provision of free or discounted alcohol within the same 24 hour period.

Posters and flyers

10.43 Irresponsible promotions can also include the sale or supply of alcohol in association with promotional materials on display in or around the premises, which can either be reasonably considered to condone, encourage or glamorise anti social behaviour or refer to the effects of drunkenness in any favourable manner.

Dispensing alcohol directly into the mouth

10.44 The responsible person (see paragraph 10.39) must ensure that no alcohol is dispensed directly into the mouth of a customer. For example, this may include drinking games such as the 'dentist's chair' where a drink is poured continuously into the mouth of another individual and may also prevent a premises from allowing another body to promote its products by employing someone to dispense alcohol directly into customers' mouths. An exception to this condition would be when an individual is unable to drink without assistance due to a disability.

Free potable water

10.45 The responsible person (see paragraph 10.39) must ensure that free potable water is provided on request to customers where it is reasonably available on the premises. What is meant by reasonably available is a question of fact; for example, it would not be reasonable to expect free tap water to be available in premises for which the water supply had temporarily been lost because of a broken mains water supply. However, it may be reasonable to expect bottled water to be provided in such circumstances.

Age verification

- 10.46 The premises licence holder or club premises certificate holder must ensure that an age verification policy applies to the premises in relation to the sale or supply of alcohol. This must as a minimum require individuals who appear to the responsible person (see paragraph 10.39) to be under the age of 18 years of age to produce on request, before being served alcohol, identification bearing their photograph, date of birth, and either a holographic mark or ultraviolet feature. The Home Office encourages licensed premises to accept cards bearing the Proof of Age Standards Scheme (PASS) hologram as their preferred proof of age, while acknowledging that many other forms of identification meet the requirements of the mandatory condition.
- 10.47 The premises licence holder or club premises certificate holder must ensure that staff (in particular, staff who are involved in the supply of alcohol) are made aware of the existence and content of the age verification policy which applies to the premises.
- 10.48 The designated premises supervisor (where there is one) must ensure that the supply of alcohol at the premises is carried on in accordance with the age verification policy. This means that the DPS has personal responsibility for ensuring that staff are not only aware of, but are also applying, the age verification policy.
- 10.49 It is acceptable, and indeed encouraged, for premises to have an age verification policy which requires individuals who appear to the responsible person to be under an age greater than 18 to produce such identification on request. For example, if premises have a policy that requires any individual that appears to be under the age of 21 to produce identification that meets the criteria listed above, this is perfectly acceptable under the mandatory code.
- 10.50 Licence holders should consider carefully what steps they are required to take to comply with the age verification requirements under the 2003 Act in relation to sales of alcohol made remotely. These include sales made online, by telephone and mail order sales, and alcohol delivery services. Each of these sales must comply with the requirements of the 2003 Act. The mandatory condition requires that age verification takes place before a person is served alcohol. Where alcohol is sold remotely (for example, online) or through a telephone transaction, the sale is made at this point but the alcohol is not actually served until it is delivered to the customer. Age verification measures (for example, online age verification) should be used to ensure that alcohol is not sold to any person under the age of 18. However, licence holders should also consider carefully what steps are appropriate to ensure that age verification takes place before the alcohol is served (i.e. physically delivered) to the customer to be satisfied that the customer is aged 18 or over. It is, therefore, the responsibility of the person serving or delivering the alcohol to ensure that age verification has taken place and that photo ID has been checked if the person appears to be less than 18 years of age.

Smaller measures

- 10.51 The responsible person (see paragraph 10.39) shall ensure that the following drinks, if sold or supplied on the premises, are available in the following measures:
 - Beer or cider: 1/2 pint
 - Gin, rum, vodka or whisky: 25ml or 35ml
 - Still wine in a glass: 125ml

- 10.52 As well as making the drinks available in the above measures, the responsible person must also make customers aware of the availability of these measures by displaying them on printed materials available to customers on the premises. This can include making their availability clear on menus and price lists, and ensuring that these are displayed in a prominent and conspicuous place in the relevant premises (for example, at the bar). Moreover, staff must make customers aware of the availability of small measures when customers do not request that they be sold alcohol in a particular measure.
- 10.53 This condition does not apply if the drinks in question are sold or supplied having been made up in advance ready for sale or supply in a securely closed container. For example, if beer is only available in pre-sealed bottles the requirement to make it available in 1/2 pints does not apply.
- 10.54 The premises licence holder or club premises certificate holder must ensure that staff are made aware of the application of this condition.

Ban on sales of alcohol below the permitted price

- 10.55 The relevant person (the holder of the premises licence, the designated premises supervisor (if any) in respect of such a licence, the personal licence holder who makes or authorises a supply of alcohol under such a licence, or any member or officer of a club present on the premises in a capacity which enables the member or officer to prevent the supply in question) shall ensure that no alcohol is sold or supplied for consumption on or off the premises for a price which is less than the permitted price.
- 10.56 The permitted price is defined as the aggregate of the duty chargeable in relation to the alcohol on the date of its sale or supply and the amount of that duty multiplied by a percentage which represents the rate of VAT chargeable in relation to the alcohol on the date of its sale or supply. Detailed guidance on how to make this calculation and a calculator to determine permitted prices for each product are available on the GOV.UK website.
- 10.57 Where there is a change to the rate of duty or VAT applying to alcohol (for instance, following a Budget), the relevant person should ensure that the permitted price reflects the new rates within fourteen days of the introduction of the new rate.
- 10.58 It is still permitted to sell alcohol using promotions (as long as they are compatible with any other licensing condition that may be in force), and the relevant person should ensure that the price of the alcohol is not less than the permitted price. Detailed guidance on the use of promotions is given in the guidance document available on the GOV.UK website.

Exhibition of films

- 10.59 The 2003 Act provides that where a premises licence or club premises certificate authorises the exhibition of a film, it must include a condition requiring the admission of children to films to be restricted in accordance with recommendations given either by a body designated under section 4 of the Video Recordings Act 1984 specified in the licence (currently only the British Board of Film Classification (BBFC)) or by the licensing authority itself.
- 10.60 The effect of paragraph 5 of Schedule 1 to the 2003 Act is to exempt adverts from the

definition of regulated entertainment, but not exempt them from the definition of exhibition of a film. Since the above mandatory condition applies to 'any film', it is therefore applicable to the exhibition of adverts.

Door supervision

- 10.61 Under section 21 of the 2003 Act, when a condition is included in a premises licence that at specified times an individual must be present at the premises to carry out a security activity (as defined in section 21(3)(a) by reference to the Private Security Industry Act 2001 ("the 2001 Act")), the licence must include a condition requiring that individual to be licensed by the Security Industry Authority ("the SIA") under the 2001 Act, or be entitled to carry out that activity by virtue of section 4 of the 2001 Act.
- 10.62 A premises licence need not require a person to hold a licence granted by the SIA if that person benefits from an exemption under section 4 of the 2001 Act. For example, certain employees benefit from an exemption when carrying out conduct in connection with a certified sports ground (section 4(6) to (12)). Furthermore, in certain circumstances persons benefit from an exemption where they operate under the SIA's Approved Contractor Scheme (section 15).
- 10.63 Conditions under section 21 of the 2003 Act should only relate to individuals carrying out security activities defined by section 21(3)(a) of the 2003 Act. Therefore, they should only relate to an activity to which paragraph 2(1)(a) of Schedule 2 to the 2001 Act applies (certain manned guarding activities) and which is licensable conduct within the meaning of section 3(2) of that Act. The requirement does not relate to individuals performing non-security related activities, and section 21 should not be used in relation to any such activities.
- 10.64 Section 21 of the 2003 Act continues to ensure that a premises licence need not impose such a requirement in relation to those licensed premises which the 2001 Act treats as unlicensed premises. Those are:
 - premises in respect of which there is in force a premises licence authorising a performance of a play or an exhibition of a film;
 - casinos or bingo halls licensed under the Gambling Act 2005;
 - premises where a club certificate is in force when activities are being carried on under the authority of that certificate.

See paragraph 8(3) of Schedule 2 to the 2001 Act for full details.

- 10.65 It should be noted, however, that the 2001 Act will require contractors and a small number of employees (those managing/supervising and those supplied under contract) to be licensed as manned guards (rather than door supervisors) when undertaking licensable conduct on premises to which paragraph 8(3) of Schedule 2 to the 2001 Act applies.
- 10.66 It is therefore important that if a licensing authority intends that individuals must be present to carry out security activities (as defined by section 21(3)(a) of the 2003 Act) this should be explicit, as should the mandatory condition for those individuals to hold an SIA licence or be entitled to carry out that activity by virtue of section 4 of the 2001 Act. On the other hand, where a licensing authority intends that individuals must be present to carry out other activities (for example, activities related to safety or steward activities to organise, advise and direct members of the public), no mandatory condition

should be imposed under section 21 of the 2003 Act. In all cases it is important when determining whether or not a condition is to be imposed under section 21 of the 2003 Act to consider whether the activities of any individual working in licensed premises fall within the definition of security activities in section 21(3)(a) of the 2003 Act. (Regardless of whether a condition is imposed under section 21 of the 2003 Act, under the 2001 Act the appropriate SIA licence must be held by any individual performing an activity for which they are licensable under that Act.)

11. Reviews

The review process

- 11.1 The proceedings set out in the 2003 Act for reviewing premises licences and club premises certificates represent a key protection for the community where problems associated with the licensing objectives occur after the grant or variation of a premises licence or club premises certificate.
- 11.2 At any stage, following the grant of a premises licence or club premises certificate, a responsible authority, or any other person, may ask the licensing authority to review the licence or certificate because of a matter arising at the premises in connection with any of the four licensing objectives.
- 11.3 An application for review may be made electronically, provided that the licensing authority agrees and the applicant submits a subsequent hard copy of the application, if the licensing authority requires one. The licensing authority may also agree in advance that the application need not be given in hard copy. However, these applications are outside the formal electronic application process and may not be submitted via GOV.UK or the licensing authority's electronic facility. The applicant must give notice of the review application to the responsible authorities and holder of the licence or certificate. The licensing authority is required to advertise a review application.
- 11.4 In addition, the licensing authority must review a licence if the premises to which it relates was made the subject of a closure order by the police based on nuisance or disorder and the magistrates' court has sent the authority the relevant notice of its determination, or if the police have made an application for summary review on the basis that premises are associated with serious crime and/or disorder.
- 11.5 Any responsible authority under the 2003 Act may apply for a review of a premises licence or club premises certificate. Therefore, the relevant licensing authority may apply for a review if it is concerned about licensed activities at premises and wants to intervene early without waiting for representations from other persons. However, it is not expected that licensing authorities should normally act as responsible authorities in applying for reviews on behalf of other persons, such as local residents or community groups. These individuals or groups are entitled to apply for a review for a licence or certificate in their own right if they have grounds to do so. It is also reasonable for licensing authorities to expect other responsible authority. For example, the police should take appropriate steps where the basis for the review is concern about crime and disorder or the sexual exploitation of children. Likewise, where there are concerns about noise nuisance, it is reasonable to expect the local authority exercising environmental health functions for the area in which the premises are situated to make the application for review.
- 11.6 Where the relevant licensing authority does act as a responsible authority and applies for a review, it is important that a separation of responsibilities is still achieved in this process to ensure procedural fairness and eliminate conflicts of interest. As outlined previously in Chapter 9 of this Guidance, the distinct functions of acting as licensing authority and responsible authority should be exercised by different officials to ensure a separation of responsibilities. Further information on how licensing authorities should

achieve this separation of responsibilities can be found in Chapter 9, paragraphs 9.13 to 9.19 of this Guidance.

- 11.7 In every case, any application for a review must relate to particular premises in respect of which there is a premises licence or club premises certificate and must be relevant to the promotion of one or more of the licensing objectives. Following the grant or variation of a licence or certificate, a complaint regarding a general issue in the local area relating to the licensing objectives, such as a general (crime and disorder) situation in a town centre, should generally not be regarded as relevant unless it can be positively tied or linked by a causal connection to particular premises, which would allow for a proper review of the licence or certificate. For instance, a geographic cluster of complaints, including along transport routes related to an individual public house and its closing time, could give grounds for a review of an existing licence as well as direct incidents of crime and disorder around a particular public house.
- 11.8 Where a licensing authority receives a geographic cluster of complaints, consideration may be given as whether these issues are the result of the cumulative impact of licensed premises within the area concerned. In such circumstances, the licensing authority may be asked to consider whether it would be appropriate to publish a cumulative impact assessment. Further guidance on cumulative impact assessments can be found in Chapter 14 of this Guidance.
- 11.9 Responsible authorities and other persons may make representations in respect of an application to review a premises licence or club premises certificate. They must be relevant (i.e., relate to one or more of the licensing objectives) and, in the case of other persons, must not be frivolous or vexatious. Representations must be made in writing and may be amplified at the subsequent hearing or may stand in their own right. Additional representations which do not amount to an amplification of the original representation may not be made at the hearing. Representations may be made electronically, provided the licensing authority agrees and the applicant submits a subsequent hard copy, unless the licensing authority waives this requirement.
- 11.10 Where authorised persons and responsible authorities have concerns about problems identified at premises, it is good practice for them to give licence holders early warning of their concerns and the need for improvement, and where possible they should advise the licence or certificate holder of the steps they need to take to address those concerns. A failure by the holder to respond to such warnings is expected to lead to a decision to apply for a review. Co-operation at a local level in promoting the licensing objectives should be encouraged and reviews should not be used to undermine this co-operation.
- 11.11 If the application for a review has been made by a person other than a responsible authority (for example, a local resident, residents' association, local business or trade association), before taking action the licensing authority must first consider whether the complaint being made is relevant, frivolous, vexatious or repetitious. Further guidance on determining whether a representation is frivolous or vexatious can be found in Chapter 9 of this Guidance (paragraphs 9.4 to 9.10).

Repetitious grounds of review

11.12 A repetitious ground is one that is identical or substantially similar to:

- a ground for review specified in an earlier application for review made in relation to the same premises licence or certificate which has already been determined; or
- representations considered by the licensing authority when the premises licence or certificate was granted; or
- representations which would have been made when the application for the premises licence was first made and which were excluded then by reason of the prior issue of a provisional statement; and, in addition to the above grounds, a reasonable interval has not elapsed since that earlier review or grant.
- 11.13 Licensing authorities are expected to be aware of the need to prevent attempts to review licences merely as a further means of challenging the grant of the licence following the failure of representations to persuade the licensing authority on an earlier occasion. It is for licensing authorities themselves to judge what should be regarded as a reasonable interval in these circumstances. However, it is recommended that more than one review originating from a person other than a responsible authority in relation to a particular premises should not be permitted within a 12 month period on similar grounds save in compelling circumstances or where it arises following a closure order or illegal working compliance order.
- 11.14 The exclusion of a complaint on the grounds that it is repetitious does not apply to responsible authorities which may make more than one application for a review of a licence or certificate within a 12 month period.
- 11.15 When a licensing authority receives an application for a review from a responsible authority or any other person, or in accordance with the closure procedures described in Part 8 of the 2003 Act (for example, closure orders), it must arrange a hearing. The arrangements for the hearing must follow the provisions set out in regulations. These regulations are published on the Government's legislation website (<u>www.legislation.gov.uk</u>). It is particularly important that the premises licence holder is made fully aware of any representations made in respect of the premises, any evidence supporting the representations and that the holder or the holder's legal representative has therefore been able to prepare a response.

Powers of a licensing authority on the determination of a review

- 11.16 The 2003 Act provides a range of powers for the licensing authority which it may exercise on determining a review where it considers them appropriate for the promotion of the licensing objectives.
- 11.17 The licensing authority may decide that the review does not require it to take any further steps appropriate to promoting the licensing objectives. In addition, there is nothing to prevent a licensing authority issuing an informal warning to the licence holder and/or to recommend improvement within a particular period of time. It is expected that licensing authorities will regard such informal warnings as an important mechanism for ensuring that the licensing objectives are effectively promoted and that warnings should be issued in writing to the licence holder.
- 11.18 However, where responsible authorities such as the police or environmental health officers have already issued warnings requiring improvement either orally or in writing that have failed as part of their own stepped approach to address concerns, licensing authorities should not merely repeat that approach and should take this into account

when considering what further action is appropriate. Similarly, licensing authorities may take into account any civil immigration penalties which a licence holder has been required to pay for employing an illegal worker.

- 11.19 Where the licensing authority considers that action under its statutory powers is appropriate, it may take any of the following steps:
 - modify the conditions of the premises licence (which includes adding new conditions or any alteration or omission of an existing condition), for example, by reducing the hours of opening or by requiring door supervisors at particular times;
 - exclude a licensable activity from the scope of the licence, for example, to exclude the performance of live music or playing of recorded music (where it is not within the incidental live and recorded music exemption)¹⁰;
 - remove the designated premises supervisor, for example, because they consider that the problems are the result of poor management;
 - suspend the licence for a period not exceeding three months;
 - revoke the licence.
- 11.20 In deciding which of these powers to invoke, it is expected that licensing authorities should so far as possible seek to establish the cause or causes of the concerns that the representations identify. The remedial action taken should generally be directed at these causes and should always be no more than an appropriate and proportionate response to address the causes of concern that instigated the review.
- 11.21 For example, licensing authorities should be alive to the possibility that the removal and replacement of the designated premises supervisor may be sufficient to remedy a problem where the cause of the identified problem directly relates to poor management decisions made by that individual.
- 11.22 Equally, it may emerge that poor management is a direct reflection of poor company practice or policy and the mere removal of the designated premises supervisor may be an inadequate response to the problems presented. Indeed, where subsequent review hearings are generated, it should be rare merely to remove a succession of designated premises supervisors as this would be a clear indication of deeper problems that impact upon the licensing objectives.
- 11.23 Licensing authorities should also note that modifications of conditions and exclusions of licensable activities may be imposed either permanently or for a temporary period of up to three months. Temporary changes or suspension of the licence for up to three months could impact on the business holding the licence financially and would only be expected to be pursued as an appropriate means of promoting the licensing objectives or preventing illegal working. So, for instance, a licence could be suspended for a weekend as a means of deterring the holder from allowing the problems that gave rise to the review to happen again. However, it will always be important that any detrimental financial impact that may result from a licensing authority's decision is appropriate and proportionate to the promotion of the licensing objectives and for the prevention of illegal working in licensed premises. But where premises are found to be trading irresponsibly, the licensing authority should not hesitate, where appropriate to do so, to take tough

action to tackle the problems at the premises and, where other measures are deemed insufficient, to revoke the licence.

Reviews arising in connection with crime

- 11.24 A number of reviews may arise in connection with crime that is not directly connected with licensable activities. For example, reviews may arise because of drugs problems at the premises, money laundering by criminal gangs, the sale of contraband or stolen goods, the sale of firearms, or the sexual exploitation of children. Licensing authorities do not have the power to judge the criminality or otherwise of any issue. This is a matter for the courts. The licensing authority's role when determining such a review is not therefore to establish the guilt or innocence of any individual but to ensure the promotion of the crime prevention objective.
- 11.25 Reviews are part of the regulatory process introduced by the 2003 Act and they are not part of criminal law and procedure. There is, therefore, no reason why representations giving rise to a review of a premises licence need be delayed pending the outcome of any criminal proceedings. Some reviews will arise after the conviction in the criminal courts of certain individuals, but not all. In any case, it is for the licensing authority to determine whether the problems associated with the alleged crimes are taking place on the premises and affecting the promotion of the licensing objectives. Where a review follows a conviction, it would also not be for the licensing authority to attempt to go beyond any finding by the courts, which should be treated as a matter of undisputed evidence before them.
- 11.26 Where the licensing authority is conducting a review on the grounds that the premises have been used for criminal purposes, its role is solely to determine what steps should be taken in connection with the premises licence, for the promotion of the crime prevention objective. It is important to recognise that certain criminal activity or associated problems may be taking place or have taken place despite the best efforts of the licence holder and the staff working at the premises and despite full compliance with the conditions attached to the licence. In such circumstances, the licensing authority is still empowered to take any appropriate steps to remedy the problems. The licensing authority's duty is to take steps with a view to the promotion of the licensing objectives and the prevention of illegal working in the interests of the wider community and not those of the individual licence holder.
- 11.27 There is certain criminal activity that may arise in connection with licensed premises which should be treated particularly seriously. These are the use of the licensed premises:
 - for the sale and distribution of drugs controlled under the Misuse of Drugs Act 1971 and the laundering of the proceeds of drugs crime;
 - for the sale and distribution of illegal firearms;
 - for the evasion of copyright in respect of pirated or unlicensed films and music, which does considerable damage to the industries affected;
 - for the illegal purchase and consumption of alcohol by minors which impacts on the health, educational attainment, employment prospects and propensity for crime of young people;
 - for prostitution or the sale of unlawful pornography;
 - by organised groups of paedophiles to groom children;
 - as the base for the organisation of criminal activity, particularly by gangs;

- for the organisation of racist activity or the promotion of racist attacks;
- for employing a person who is disqualified from that work by reason of their immigration status in the UK;
- for unlawful gambling; and
- for the sale or storage of smuggled tobacco and alcohol.
- 11.28 It is envisaged that licensing authorities, the police, the Home Office (Immigration Enforcement) and other law enforcement agencies, which are responsible authorities, will use the review procedures effectively to deter such activities and crime. Where reviews arise and the licensing authority determines that the crime prevention objective is being undermined through the premises being used to further crimes, it is expected that revocation of the licence even in the first instance should be seriously considered.

Review of a premises licence following closure order or illegal working compliance order

- 11.29 Licensing authorities are subject to certain timescales, set out in the legislation, for the review of a premises licence following a closure order under section 80 of the Anti-social Behaviour, Crime and Policing Act 2014 or an illegal working compliance order under section 38 of and Schedule 6 to the Immigration Act 2016. The relevant time periods run concurrently and are as follows:
 - when the licensing authority receives notice that a magistrates' court has made a closure order it has 28 days to determine the licence review – the determination must be made before the expiry of the 28th day after the day on which the notice is received;
 - the hearing must be held within ten working days, the first of which is the day after the day the notice from the magistrates' court is received;
 - notice of the hearing must be given no later than five working days before the first hearing day (there must be five clear working days between the giving of the notice and the start of the hearing).

Review of a premises licence following persistent sales of alcohol to children

11.29 The Government recognises that the majority of licensed premises operate responsibly and undertake due diligence checks on those who appear to be under the age of 18 at the point of sale (or 21 and 25 where they operate a Challenge 21 or 25 scheme). Where these systems are in place, licensing authorities may wish to take a proportionate approach in cases where there have been two sales of alcohol within very quick succession of one another (e.g., where a new cashier has not followed policy and conformed with a store's age verification procedures). However, where persistent sales of alcohol to children have occurred at premises, and it is apparent that those managing the premises do not operate a responsible policy or have not exercised appropriate due diligence, responsible authorities should consider taking steps to ensure that a review of the licence is the norm in these circumstances. This is particularly the case where there has been a prosecution for the offence under section 147A or a closure notice has been given under section 169A of the 2003 Act. In determining the review, the licensing authority should consider revoking the licence if it considers this appropriate.

12. Summary reviews

- 12.1 Summary reviews can be undertaken when the police consider that the premises concerned are associated with serious crime or serious disorder (or both). The summary review process, set out under sections 53A-53D of the 2003 Act, allows interim conditions to be quickly attached to a licence and a fast track licence review. The provisions were inserted by section 21 of the Violent Crime Reduction Act 2006 and amended by sections 136-137 of the Policing and Crime Act 2017, including the addition of section 53D.
- 12.2 The powers apply only where a premises licence authorises the sale of alcohol. They do not apply in respect of other premises licences, or to premises operating under a club premises certificate. The powers are aimed at tackling serious crime and serious disorder, in particular (but not exclusively) the use of guns and knives. The powers complement the general procedures in the 2003 Act for tackling crime and disorder associated with licensed premises and should be reserved for the most serious matters which cannot be adequately or otherwise redressed unless urgent action is taken. Separate powers in the Anti-social Behaviour, Crime and Policing Act 2014 provide for the instant closure of premises by the police in some circumstances (in essence, disorder or nuisance). The consequent review of premises licences by the licensing authority is provided for by section 167 of the Licensing At 2003.

Application for summary review

- 12.3 Section 53A of the 2003 Act sets out who may apply for an expedited review and the circumstances in which it can be used. The application is made by, or on behalf of, the chief officer of police and must be made in the form which is set out in Schedule 8A to the Licensing Act 2003 (Premises Licences and Club Premises Certificates) Regulations 2005 (SI 2005/42).
- 12.4 The completed application must be accompanied by a certificate issued by a senior officer of the rank of superintendent or above. The certificate is a formal note which identifies the licensed premises and includes a signed statement by the senior officer that in his/her opinion the premises are associated with serious crime, serious disorder or both. This form is not prescribed in legislation. However, a sample form which forces may wish to adopt is published on GOV.UK.
- 12.5 The tests to determine the kinds of conduct that amount to serious crime are set out in section 81(2) and (3) of the Regulation of Investigatory Powers Act 2000. Those tests are that the conduct:

(a) constitutes an offence for which a person who is 21 years of age or over with no previous convictions could reasonably be expected to be sentenced to imprisonment for 3 years or more; or

(b) Involves the use of violence, results in substantial financial gain or is conducted by a large number of persons in pursuit of a common purpose.

12.6 There is no definitive list of behaviours that constitute serious disorder, and the matter is one for judgment by the local police. The phrase should be given its plain, ordinary meaning, as is the case under section 12 of the Public Order Act 1986 in which it is also used.

- 12.7 In deciding whether to sign a certificate, the senior officer should consider the following (as applicable):
 - The track record of the licensed premises concerned and whether the police have previously had cause to give advice about serious criminal or disorderly conduct (or the likelihood of such conduct) attributable to activities taking place on the premises. It is not expected that this power will be used as a first response to a problem and summary reviews triggered by a single incident are likely to be the exception.
 - The nature of the likely crime and/or disorder is the potential incident sufficiently serious to warrant using this power?
 - Should an alternative power be deployed? Is the incident sufficiently serious to warrant use of the powers in Part 4, Chapter 3 of the Anti-social Behaviour, Crime and Policing Act 2014, or section 38 of and Schedule 6 to the Immigration Act 2016, to close the premises? Or could the police trigger a standard licence review to address the problem? Alternatively, could expedited reviews be used in conjunction with other powers (for example, modifying licence conditions following the use of a closure power)?
 - What added value will use of the expedited process bring? How would any interim steps that the licensing authority might take effectively address the problem?
- 12.8 It is recommended that these points are addressed in the chief officer's application to the licensing authority. In particular, it is important to explain why other powers or actions are not considered to be appropriate. It is up to the police to decide whether to include this information in the certificate or in section 4 of the application for summary review. The police will also have an opportunity later to make representations in relation to the full review. In appropriate circumstances the police might want to make representations to the licensing authority suggesting that they modify the conditions of the premises licence to require searches of customers for offensive weapons upon entry. Under the powers in sections 53A to 53D, this could be done on an interim basis pending a full hearing of the issues within the prescribed 28-day timeframe or for an appropriate period determined by the licensing authority.
- 12.9 Similarly, the power could, where appropriate, be used to reduce the risk of injury caused by glass by requiring the adoption of a safer alternative (but see paragraphs 12.15 and 12.16 below). However, it should always be borne in mind that the aim of the powers is to provide a selective tool, to be used proportionately.

The licensing authority and interim steps pending the review

- 12.10 Within 48 hours of receipt of the chief officer's application, the licensing authority must give the premises licence holder and responsible authorities a notice of the review and should include a copy of the application for review and a copy of the certificate, and must also consider whether it is necessary to take interim steps (place temporary conditions on the licence). When calculating the 48 hour period any non-working day can be disregarded¹¹.
- 12.11 The licensing authority may want to consult the police about the steps that it thinks are

¹¹ This means that, for example, if the application was received at 3pm on a Friday, the 48 hour period would cover the remaining 9 hours on that Friday and the remaining 39 hours starting on the Monday morning (provided it was not a bank holiday). In this case the licensing authority would have to decide on interim steps by 3pm on the Tuesday.

necessary, pending the determination of the review, to address the immediate problems with the premises, in particular the likelihood of serious crime and/or serious disorder. The licensing authority may consider the interim steps without the holder of the premises licence having been given an opportunity to make representations. This does not, of course, mean that the licensing authority *cannot* afford such an opportunity if it thinks it appropriate and feasible to do so in all the circumstances.

- 12.12 The determination of interim steps is not a matter that may be delegated to an officer of the licensing authority. The relevant decisions are likely to be taken by a licensing subcommittee rather than the full committee. It should also be noted that there is no requirement for a formal hearing in order to take interim steps. This means that the relevant sub committee members can communicate by telephone or other remote means in order to reach a decision. A written record should always be produced as soon as possible after a decision is reached.
- 12.13 The interim steps that the licensing authority must consider taking are:
 - the modification of the conditions of the premises licence;
 - the exclusion of the sale of alcohol by retail from the scope of the licence;
 - the removal of the designated premises supervisor from the licence; and
 - the suspension of the licence.

Modification of the conditions of the premises licence can include the alteration or modification of existing conditions or addition of any new conditions, including those that restrict the times at which licensable activities authorised by the licence can take place.

- 12.14 If the licensing authority decides to take steps at the initial interim stage:
 - the decision takes effect immediately, or as soon after it as the licensing authority directs; but
 - the licensing authority must give immediate notice of its decision and its reasons for doing so to the holder of the premises licence and the chief officer of police who made the application. The 2003 Act does not specify that the immediate notice has to be in writing. However, in an individual case the licensing authority may consider that the need for immediate communication at least initially requires a non-written approach, such as a telephone call. This may happen when, for example, the authority decides that the decision should have immediate effect. In such a case, the decision and the reasons for it should be explained clearly and in full to the licenceholder (or someone who may properly act for the licence-holder), and the call followed up as soon as possible with a written version of the decision and the reasons (for example, by email or fax) which is identical to, or not significantly different from, the version given by telephone.
- 12.15 The licensing authority, in deciding when its decision on interim steps should take effect, should consider the practical implications of compliance in relation to the premises. For example to comply with a modification of the conditions of a licence that requires employment of door supervisors, those running the premises may need some time to recruit appropriately qualified and accredited staff.
- 12.16 In addition, very careful consideration needs to be given to interim steps which would require significant cost or permanent or semi-permanent adjustments to premises which would be difficult to remove if the outcome of the subsequent full review was to withdraw

or modify those steps. For example, making structural changes, installing additional CCTV or replacing all glassware with safer alternatives may be valid steps, but might be disproportionate if they are not likely to be deemed necessary following the full review (or any subsequent appeal). The focus for interim steps should be on the immediate measures that are necessary to prevent serious crime or serious disorder occurring.

Making representations against the interim steps

- 12.17 The premises licence holder may make representations against the interim steps taken by the licensing authority. There is no time limit for the premises licence holder to make representations on the interim steps, although in practice this would at some point be superseded by the full review which would have to be completed within 28 days of the application being received by the licensing authority. On receipt of such representations, the licensing authority must (if the representations are not withdrawn) hold a hearing within 48 hours of their receipt. When calculating the 48 hour period, any non-working day can be disregarded. Where the licensing authority has already held a hearing to consider representations against the interim steps, the holder of the licence may only make further representations if there has been a material change in circumstances.
- 12.18 The licensing authority must give advance notice of the hearing to the premises licence holder and the chief officer of police. Given that these measures are designed to deal with serious crime and/or serious disorder on an interim basis only, the process is designed to avoid delay and, as such, significant portions of the Licensing Act 2003 (Hearings) Regulations 2005 (SI 2005/44) (which set out the usual processes governing the conduct of licensing authority hearings) do not apply in order to streamline the hearing process. One result of this is that the licensing authority cannot adjourn the hearing to a later date if the licence holder fails to attend at the scheduled time, as is the case under the normal review procedure. And as is the case with that procedure, the licence holder does not have to be present for the hearing to take place. In addition, there is no timescale for notifying the licence holder of the hearing under the modified process, providing the notification takes place before the hearing is held. However, it is imperative that the licence holder be given as much notice as is possible in the circumstances to afford the holder a maximum practicable opportunity to prepare for and attend the hearing. Licensing authorities should bear in mind that the usual principles of public law decision-making will apply to interim determinations, in a form that has regard to the statutory context of an expedited process.
- 12.19 At the hearing to consider representations against interim steps the licensing authority must:
 - consider whether the interim steps are appropriate for the promotion of the licensing objectives; and
 - determine whether to withdraw or modify the steps taken.
- 12.20 When considering the case the licensing authority must take into account:
 - the senior officer's certificate that accompanied the application;
 - the chief officer's representations (if any); and
 - any representations made by the premises licence holder.
- 12.21 There is no right of appeal to a magistrates' court against the licensing authority's

decision at this stage.

The review of the premises licence under section 53C

- 12.22 The licensing authority must hold a full review of the premises licence and determine the review within 28 days after the day of receipt of the chief officer's application. There can be no adjournment of the hearing or delay in reaching a determination beyond the end of the 28 day period. This must take place even if the chief officer or other responsible authority and/or other person asks to withdraw their application or representation (as the case may be). At the review hearing, the licensing authority must consider what steps are appropriate for the promotion of the licensing objectives taking into account the application and any change in circumstances since any interim steps were imposed, consider any relevant representations, and review the interim steps already taken (if any).
- 12.23 In making its final determination the steps the licensing authority can take are:
 - the modification of the conditions of the premises licence;
 - the exclusion of a licensable activity from the scope of the licence;
 - the removal of the designated premises supervisor from the licence;
 - the suspension of the licence for a period not exceeding 3 months; and
 - the revocation of the licence.
- 12.24 Modification of the conditions of the premises licence can include the alteration or modification of existing conditions or addition of any new conditions, including those that restrict the times at which licensable activities authorised by the licence can take place.
- 12.25 The licensing authority must:
 - advertise the review inviting representations from any persons for no less than seven consecutive days, by notice as described in regulation 38 of the Licensing Act 2003 (Premises Licences and Club Premises Certificates) Regulations 2005 (SI 2005/42) and, if applicable, on the licensing authority's website (see regulation 38(1)(b) of the above). The relevant notices should be published on the day after the day of receipt of the chief officer's application.
 - advertise that any representations which the premises licence holder, responsible authority or any other person want the licensing authority to consider at the review hearing, should be submitted to the licensing authority within 10 working days of the advertisement of the review appearing.
 - give formal notice of the hearing no later than five working days before the day or first day on which the hearing is to be held to the premises licence holder and to every responsible authority.
- 12.26 A party shall give to the licensing authority a notice no later than two working days before the day or the first day on which the hearing is to be held stating
 - whether he intends to attend or be represented at the hearing;
 - whether he considers a hearing to be unnecessary.
 - whether he would like permission for any other person (other than the person he intends to represent him at the hearing) to appear at the hearing and, if so, explain on which points that person will be able to contribute.

- 12.27 The regulations relating to hearings are set out in the Licensing Act 2003 (Hearings) Regulations 2005 (S.I. 2005/44). They apply to final hearings under the section 53A(2)(b) in a similar way to hearings following closure orders under section 167 of the 2003 Act (it should be emphasised that the truncated version of the hearings regulations described in paragraph 12.18 above applies to interim hearings only). The issues they address include who can make representations and what those representations can be about. It is therefore possible for responsible authorities or any other persons to make representations in relation to any of the licensing objectives, not just crime and disorder. Similarly, where it is in the public interest, the regulations relating to the exclusion of individuals from hearings, or conducting the hearing in private, will apply.
- 12.28 The licensing authority must notify its decision and the reasons for making it to:
 - the holder of the premises licence;
 - any person who made relevant representations; and
 - the chief officer of police who made the original application.

Review of the interim steps under section 53D

- 12.29 The licensing authority's determination does not have effect until the end of the 21 day period given for appealing the decision, or until the disposal of any appeal that is lodged (see below information on right of appeal). To ensure that there are appropriate and proportionate safeguards in place at all times, the licensing authority is required to review any interim steps that it has taken that are in place on the date of the hearing and consider whether it is appropriate for the promotion of the licensing objectives for the steps to remain in place, or if they should be modified or withdrawn. The review of the interim steps should take place immediately after the determination under section 53C has been reached. In making its decision, the licensing authority must consider any relevant representations made.
- 12.30 In conducting the review of the interim steps the licensing authority has the power to take any of the steps that were available to it at the initial stage (see paragraph 12.13). Any interim steps taken at the review hearing apply until—

(a) the end of the period given for appealing against a decision made under section 53C (21 days),

(b) if the decision under section 53C is appealed against, the time the appeal is disposed of, or

(c) the end of a period determined by the relevant licensing authority (which may not be longer than the period of time for which such interim steps could apply under (a) or (b) above).

Right of appeal against review of interim steps decision

12.31 The licence holder or the chief officer of police may appeal against the decision made by the licensing authority concerning its review of the interim steps to a magistrates' court. The appeal must be made within 21 days of the appellant being notified of the licensing authority's decision and must be heard in full by the magistrates' court within 28 days beginning with the day on which the appellant lodged the appeal.

Right of appeal against final review decision

- 12.32 An appeal against the final review decision may be made to a magistrates' court within 21 days of the appellant being notified of the licensing authority's determination on the review. An appeal may be made by the premises licence holder, the chief officer of police and/or any other person who made relevant representations.
- 12.33 The decision of the licensing authority, following the review hearing, will not have effect until the end of the period allowed for appeal, or until the disposal of the appeal.
- 12.34 Where appeals are lodged both against the decision following the review of the interim steps and against the final determination, the courts may decide to consider the appeal against the final determination within the 28 day period, allowing the interim steps appeal to be disposed of at the same time.

Flow diagram of the summary review process

12.35 The following flow diagram summarises the process.





¹² Licensing authority

¹³ Only working days count

13. Appeals

13.1 This chapter provides advice about entitlements to appeal in connection with various decisions made by a licensing authority under the provisions of the 2003 Act. Entitlements to appeal for parties aggrieved by decisions of the licensing authority are set out in Schedule 5 to the 2003 Act.

General

- 13.2 With the exception of appeals in relation to closure orders, an appeal may be made to any magistrates' court in England or Wales but it is expected that applicants would bring an appeal in a magistrates' court in the area in which they or the premises are situated.
- 13.3 An appeal has to be commenced by the appellant giving a notice of appeal to the designated officer for the magistrates' court within a period of 21 days beginning with the day on which the appellant was notified by the licensing authority of the decision which is being appealed.
- 13.4 The licensing authority will always be a respondent to the appeal, but in cases where a favourable decision has been made for an applicant, licence holder, club or premises user against the representations of a responsible authority or any other person, or the objections of the chief officer of police, the Home Office (Immigration Enforcement), or local authority exercising environmental health functions, the holder of the premises or personal licence or club premises certificate or the person who gave an interim authority notice or the premises user will also be a respondent to the appeal, and the person who made the relevant representation or gave the objection will be the appellants.
- 13.5 Where an appeal has been made against a decision of the licensing authority, the licensing authority will in all cases be the respondent to the appeal and may call as a witness a responsible authority or any other person who made representations against the application, if it chooses to do so. For this reason, the licensing authority should consider keeping responsible authorities and others informed of developments in relation to appeals to allow them to consider their position. Provided the court considers it appropriate, the licensing authority may also call as witnesses any individual or body that they feel might assist their response to an appeal.
- 13.6 The court, on hearing any appeal, may review the merits of the decision on the facts and consider points of law or address both.
- 13.7 On determining an appeal, the court may:
 - dismiss the appeal;
 - substitute for the decision appealed against any other decision which could have been made by the licensing authority; or
 - remit the case to the licensing authority to dispose of it in accordance with the direction of the court and make such order as to costs as it thinks fit.

All parties should be aware that the court may make an order for one party to pay another party's costs.

On any appeal, the court is not entitled to consider whether the licence holder should have been convicted of an immigration offence or been required to pay an immigration penalty, or whether they should have been granted by the Home Office permission to be in the UK. This is because separate rights exist to appeal these matters or to have an immigration decision administratively reviewed.

Licensing policy statements and Section 182 guidance

- 13.8 In hearing an appeal against any decision made by a licensing authority, the magistrates' court will have regard to that licensing authority's statement of licensing policy and this Guidance. However, the court would be entitled to depart from either the statement of licensing policy or this Guidance if it considered it was justified to do so because of the individual circumstances of any case. In other words, while the court will normally consider the matter as if it were "standing in the shoes" of the licensing authority, it would be entitled to find that the licensing authority should have departed from its own policy or the Guidance because the particular circumstances would have justified such a decision.
- 13.9 In addition, the court is entitled to disregard any part of a licensing policy statement or this Guidance that it holds to be ultra vires the 2003 Act and therefore unlawful. The normal course for challenging a statement of licensing policy or this Guidance should be by way of judicial review, but where it is submitted to an appellate court that a statement of policy is itself ultra vires the 2003 Act and this has a direct bearing on the case before it, it would be inappropriate for the court, on accepting such a submission, to compound the original error by relying on that part of the statement of licensing policy affected.

Giving reasons for decisions

- 13.10 It is important that a licensing authority gives comprehensive reasons for its decisions in anticipation of any appeals. Failure to give adequate reasons could itself give rise to grounds for an appeal. It is particularly important that reasons should also address the extent to which the decision has been made with regard to the licensing authority's statement of policy and this Guidance. Reasons should be promulgated to all the parties of any process which might give rise to an appeal under the terms of the 2003 Act.
- 13.11 It is important that licensing authorities also provide all parties who were party to the original hearing, but not involved directly in the appeal, with clear reasons for any subsequent decisions where appeals are settled out of court. Local residents in particular, who have attended a hearing where the decision was subject to an appeal, are likely to expect the final determination to be made by a court.

Implementing the determination of the magistrates' courts

13.12 As soon as the decision of the magistrates' court has been promulgated, licensing authorities should implement it without delay. Any attempt to delay implementation will only bring the appeal system into disrepute. Standing orders should therefore be in place that on receipt of the decision, appropriate action should be taken immediately unless ordered by the magistrates' court or a higher court to suspend such action (for example, as a result of an on-going judicial review). Except in the case of closure orders, the 2003 Act does not provide for a further appeal against the decision of the magistrates' courts and normal rules of challenging decisions of magistrates' courts will apply.

Provisional statements

13.13 To avoid confusion, it should be noted that a right of appeal only exists in respect of the terms of a provisional statement that is issued rather than one that is refused. This is because the 2003 Act does not empower a licensing authority to refuse to issue a provisional statement. After receiving and considering relevant representations, the licensing authority may only indicate, as part of the statement, that it would consider certain steps to be appropriate for the promotion of the licensing objectives when, and if, an application were made for a premises licence following the issuing of the provisional statement. Accordingly, the applicant or any person who has made relevant representations may appeal against the terms of the statement issued.

14. Statements of licensing policy

Introduction

The Licensing Act 2003

14.1 This chapter provides guidance on the development and preparation of local statements of licensing policy for publication by licensing authorities, the general principles that it is recommended should underpin them, and core content to which licensing authorities are free to add.

General

- 14.2 Section 5 of the 2003 Act requires a licensing authority to prepare and publish a statement of its licensing policy at least every five years. Such a policy must be published before the authority carries out any function in respect of individual applications and notices made under the terms of the 2003 Act. During the five-year period, the policy must be kept under review and the licensing authority may make any revisions to it as it considers appropriate, for instance in the light of feedback from the local community on whether the licensing objectives are being met. If the licensing authority determines and publishes its policy in this way, a new five-year period commences on the date it is published. Previously, licensing authorities were required to determine their licensing policies for each three-year period. Licensing policies published in respect of the three-year period that began on 7 January 2011 are to be treated as though they apply to a period of five years beginning at that date.
- 14.3 Where revisions to the section 182 Guidance are made by the Secretary of State, it will be for the licensing authority to determine whether revisions to its own licensing policy statement are appropriate.

Consultation on policies

- 14.4 Before determining its policy, the licensing authority must consult the persons listed in section 5(3) of the 2003 Act. These are:
 - the chief officer of police for the area;
 - the fire and rescue authority for the area;
 - each local authority's Director of Public Health in England (DPH)¹⁴ or Local Health Board in Wales for an area any part of which is in the licensing authority's area,
 - persons/bodies representative of local premises licence holders;
 - persons/bodies representative of local club premises certificate holders;
 - persons/bodies representative of local personal licence holders; and
 - persons/bodies representative of businesses and residents in its area.
- 14.5 The views of all these persons or bodies should be given appropriate weight when the policy is determined. It is recognised that in some areas, it may be difficult to identify persons or bodies that represent all parts of industry affected by the provisions of the

¹⁴ This change was made as a result of the commencement of measures in the Health and Social Care Act 2012 which amended the 2003 Act and further provision in the NHS Bodies and Local Authorities (Partnership Arrangements, Care Trusts, Public Health and Local Healthwatch) Regulations 2012.

2003 Act, but licensing authorities must make reasonable efforts to do so. Licensing authorities should note that the terms of the 2003 Act do not prevent them consulting other bodies or persons.

- 14.6 Subject to the statutory requirements, it is for each licensing authority to determine the extent of the consultation it should undertake, and whether any particular person or body is representative of the groups described in the 2003 Act. While it is clearly good practice to consult widely, this may not always be necessary or appropriate (for example, where a licensing authority has recently carried out a comprehensive consultation in relation to a revision to its policy made within five years of a full revision to it). As such, it may decide on a simple consultation with those persons listed.
- 14.7 However, licensing authorities should consider very carefully whether a full consultation is appropriate as a limited consultation may not allow all persons sufficient opportunity to comment on and influence local policy (for example, where an earlier consultation was limited to a particular part of the policy).
- 14.8 Fee levels are intended to provide full cost recovery of all licensing functions including the preparation and publication of a statement of licensing policy, but this will be based on the statutory requirements. Where licensing authorities exceed these requirements, they will have to absorb those costs themselves.

Fundamental principles

- 14.9 All statements of policy should begin by stating the four licensing objectives, which the licensing policy should promote. In determining its policy, a licensing authority must have regard to this Guidance and give appropriate weight to the views of consultees.
- 14.10 While statements of policy may set out a general approach to making licensing decisions, they must not ignore or be inconsistent with provisions in the 2003 Act. For example, a statement of policy must not undermine the right of any person to apply under the terms of the 2003 Act for a variety of permissions and to have any such application considered on its individual merits.
- 14.11 Similarly, no statement of policy should override the right of any person to make representations on an application or to seek a review of a licence or certificate where provision has been made for them to do so in the 2003 Act.
- 14.12 Statements of policies should make clear that:
 - licensing is about regulating licensable activities on licensed premises, by qualifying clubs and at temporary events within the terms of the 2003 Act; and
 - conditions attached to various authorisations will be focused on matters which are within the control of individual licence holders and others with relevant authorisations, i.e. the premises and its vicinity.
- 14.13 A statement of policy should also make clear that licensing law is not the primary mechanism for the general control of nuisance and anti-social behaviour by individuals once they are away from the licensed premises and, therefore, beyond the direct control of the individual, club or business holding the licence, certificate or authorisation concerned. Nonetheless, it is a key aspect of such control and licensing law will always be part of a holistic approach to the management of the evening and night-time economy in town and city centres.

Licence conditions

- 14.14 Statements of licensing policy should reflect the general principles regarding licence conditions set out in Chapter 1 of this guidance.
- 14.15 Statements of licensing policy should include a firm commitment to avoid attaching conditions that duplicate other regulatory regimes as far as possible. Chapter 10 provides further detail on this issue.

Enforcement

- 14.16 The Government recommends that licensing authorities should establish and set out joint enforcement protocols with the local police and the other authorities and describe them in their statement of policy. This will clarify the division of responsibilities for licence holders and applicants, and assists enforcement and other authorities to deploy resources more efficiently.
- 14.17 In particular, these protocols should also provide for the targeting of agreed problem and high-risk premises which require greater attention, while providing a lighter touch for low risk premises or those that are well run. In some local authority areas, the limited validity of public entertainment, theatre, cinema, night café and late night refreshment house licences has in the past led to a culture of annual inspections regardless of whether the assessed risks make such inspections necessary. The 2003 Act does not require inspections to take place save at the discretion of those charged with this role. Principles of risk assessment and targeted inspection (in line with the Regulators' Code) should prevail and, for example, inspections should not be undertaken routinely but when and if they are judged necessary. This should ensure that resources are used efficiently and for example, are more effectively concentrated on problem premises. Licensing authorities should also remind operators of licensed premises that it is incumbent on them to provide appropriate training for their staff to ensure the promotion of the licensing objectives.

Entertainment provision

14.18 Statements of licensing policy should set out the extent to which the licensing authority intends to facilitate a broad range of entertainment provision for enjoyment by a wide cross-section of the public. Statements of licensing policy should address what balance is to be struck between promoting the provision of entertainment and addressing concerns relevant to the licensing objectives. Licensing authorities should be conscious that licensing policy may inadvertently deter live music by imposing indirect costs of a disproportionate nature, for example a blanket policy that any pub providing live music entertainment must have door supervisors.

The need for licensed premises

14.19 There can be confusion about the difference between the "need" for premises and the "cumulative impact" of premises on the licensing objectives, for example, on crime and disorder. "Need" concerns the commercial demand for another pub or restaurant or hotel and is a matter for the planning authority and for the market. This is not a matter for a licensing authority in discharging its licensing functions or for its statement of licensing policy.
The cumulative impact of a concentration of licensed premises What is cumulative impact?

- 14.20 The concept of "Cumulative impact" has been described within this guidance and used by licensing authorities within their statements of licensing policy since the commencement of the 2003 Act. 'Cumulative impact assessments' were introduced in the 2003 Act by the Policing and Crime Act 2017, with effect from 6 April 2018. Cumulative impact is the potential impact on the promotion of the licensing objectives of a number of licensed premises concentrated in one area.
- 14.21 In some areas where the number, type or density of licensed premises, such as those selling alcohol or providing late night refreshment, is high or exceptional, serious problems of nuisance and disorder may arise outside or some distance from those premises. Such problems generally occur as a result of large numbers of drinkers being concentrated in an area, for example when leaving premises at peak times or when queuing at fast food outlets or for public transport.
- 14.22 Queuing in itself may lead to conflict, disorder and anti-social behaviour. Moreover, large concentrations of people may also attract criminal activities such as drug dealing, pick pocketing and street robbery. Local services such as public transport, public lavatory provision and street cleaning may not be able to meet the demand posed by such concentrations of drinkers leading to issues such as street fouling, littering, traffic and public nuisance caused by concentrations of people who cannot be effectively dispersed quickly.
- 14.23 Variable licensing hours may facilitate a more gradual dispersal of customers from premises. However, in some cases, the impact on surrounding areas of the behaviour of the customers of all premises taken together will be greater than the impact of customers of individual premises. These conditions are more likely to arise in town and city centres, but may also arise in other urban centres and the suburbs, for example on smaller high streets with high concentrations of licensed premises.

Cumulative impact assessments

- 14.24 A cumulative impact assessment (CIA) may be published by a licensing authority to help it to limit the number or types of licence applications granted in areas where there is evidence to show that the number or density of licensed premises in the area is having a cumulative impact and leading to problems which are undermining the licensing objectives. CIAs relate to applications for new premises licences and club premises certificates and applications to vary existing premises licences and club premises certificates in a specified area.
- 14.25 Section 5A of the 2003 Act sets out what a licensing authority needs to do in order to publish a CIA and review it, including the requirement to consult with the persons listed in section 5(3) of the 2003 Act. The 2003 Act does not stipulate how the CIA should be used once published, because the requirements for determining applications for new licences or variations are the same in areas with a CIA as they are elsewhere, as set out in sections 18, 35, 72 and 85 of the Act. However, any CIA published by a licensing authority must be summarised in its statement of licensing policy. Under section 5(6D) a licensing authority must also have regard to any CIA it has published when determining

or revising its statement of licensing policy.

- 14.26 The CIA must include a statement saying that the licensing authority considers that the number of premises licences and/or club premises certificates in one or more parts of the area described is such that it is likely that granting further licences would be inconsistent with the authority's duty to promote the licensing objectives. As part of the publication a licensing authority must set out the evidential basis for its opinion.
- 14.27 CIAs may relate to premises licensed to carry on any licensable activity, including the sale of alcohol for consumption on or off the premises, and the provision of late night refreshment. This includes late night refreshment providers which are not licensed to sell alcohol. A CIA may relate to all premises licences and club premises certificates in the area described in the assessment or parts thereof, or only to premises of a particular kind described in the assessment. For example, it may be appropriate for the licensing authority to only include off-licences or nightclubs within the scope of its assessment. The licensing authority must make clear, when publishing its CIA, which premises types it applies to. CIAs do not apply to TENs; however it is open to the police and environmental health authority (as relevant persons) to refer to evidence published within a CIA when objecting to a TEN.
- 14.28 While the evidence underpinning the publication of a CIA should generally be suitable as the basis for a decision to refuse an application or impose conditions, it does not change the fundamental way that decisions are made under the 2003 Act. Each decision in an area subject to a CIA therefore still needs to be made on a case-by-case basis and with a view to what is appropriate for the promotion of the licensing objectives. Importantly, the publication of a CIA would not remove a licensing authority's discretion to grant applications for new licences or applications to vary existing licences, where the authority considers this to be appropriate in the light of the individual circumstances of the case.

Evidence of cumulative impact

- 14.29 As noted above, there must be an evidential basis for the decision to publish a CIA. Local Community Safety Partnerships and responsible authorities, such as the police and the environmental health authority, may hold relevant information which would inform licensing authorities when establishing the evidence base for publishing a CIA. Evidence of cumulative impact on the promotion of the licensing objectives needs to relate to the relevant problems identified in the specific area to be covered by the CIA. Information which licensing authorities may be able to draw on includes:
 - local crime and disorder statistics, including statistics on specific types of crime and crime hotspots;
 - statistics on local anti-social behaviour offences;
 - health-related statistics such as alcohol-related emergency attendances and hospital admissions;
 - environmental health complaints, particularly in relation to litter and noise;
 - complaints recorded by the local authority, which may include complaints raised by local residents or residents' associations;
 - residents' questionnaires;
 - evidence from local and parish councillors; and
 - evidence obtained through local consultation.
- 14.30 The licensing authority may consider this evidence, alongside its own evidence of the

impact of licensable activities within its area, and consider in particular the times at which licensable activities are carried on. Information which may inform consideration of these issues includes:

- trends in licence applications, particularly trends in applications by types of premises and terminal hours;
- changes in terminal hours of premises;
- premises' capacities at different times of night and the expected concentrations of drinkers who will be expected to be leaving premises at different times.
- 14.31 Where existing information is insufficient or not readily available, but the licensing authority believes there are problems in its area resulting from the cumulative impact of licensed premises, it can consider conducting or commissioning a specific study to assess the position. This may involve conducting observations of the night-time economy to assess the extent of incidents relating to the promotion of the licensing objectives, such as incidences of criminal activity and anti-social behaviour, examples of public nuisance, specific issues such as underage drinking and the key times and locations at which these problems are occurring.
- 14.32 In order to identify the areas in which problems are occurring, information about specific incidents can be mapped and, where possible, a time analysis undertaken to identify the key areas and times at which there are specific issues.
- 14.33 After considering the available evidence and consulting those individuals and organisations listed in section 5(3) of the 2003 Act and any others, a licensing authority may be satisfied that it is appropriate to publish a CIA. The CIA should also be considered alongside local planning policy and other factors which may assist in mitigating the cumulative impact of licensed premises, as set out in paragraph 14.47.

Steps to publishing a cumulative impact assessment

- 14.34 The steps to be followed in considering whether to publish a CIA are summarised below.
 - Identify concern about crime and disorder; public safety; public nuisance or protection of children from harm in a particular location.
 - Consider whether there is good evidence that crime and disorder or nuisance are occurring, or whether there are activities which pose a threat to public safety or the protection of children from harm.
 - If there is evidence that such problems are occurring, identify whether these problems are being caused by the customers of licensed premises, or that cumulative impact is imminent.
 - Identify the boundaries of the area where problems are occurring (this can involve mapping where the problems occur and identifying specific streets or localities where such problems arise).
 - Consult those specified in section 5(3) of the 2003 Act. As with consultations in respect of the licensing policy statement as a whole, it is for each licensing authority to determine the extent of the consultation it should undertake in respect of a CIA (subject to the statutory requirements).
 - For the purposes of the consultation provide the persons specified in section 5(3) with the following information:
 - the reasons why it is considering publishing a CIA;
 - a general indication of the part or parts of its area which it is considering describing in the assessment;

- whether it considers that the assessment will relate to all premises licence and club premises certificate applications and variation applications, or only to those of a particular kind described.
- Subject to the outcome of the consultation, include and publish details of the CIA, including the evidence in support of the assessment and the particular kinds of premises the assessment relates to. Licensing authorities are not restricted to using general terms such as on-trade, off-trade and late night refreshment providers, and can apply their own descriptions such as vertical-drinking bars and night clubs if appropriate.
- Summarise the licensing authority's opinion in light of the evidence of cumulative impact (or any revision to an existing opinion) in the licensing policy statement and explain within the policy statement how the authority has had regard to any CIAs it has published under section 5A. The summary within the licensing policy statement should include, but is not limited to: the nature of the problems identified and the evidence for such problems; the geographical extent of the area covered by the assessment; the types of premises described in the assessment; and the types of applications for which it would likely be inconsistent with the licensing authority's duty to promote the licensing objectives to grant.

Reviewing the CIA

- 14.35 After publishing a CIA the licensing authority must, within three years, consider whether it remains of the opinion set out in the assessment. In order to decide whether it remains of this opinion it must again consult the persons listed in section 5(3). If having consulted with the statutory list of persons the licensing authority decides that it is no longer of the opinion set out in the CIA, it must publish a statement to that effect. The statement must make clear that any reference to the CIA in its licensing policy statement no longer applies. The licensing authority should remove any reference to the CIA within its licensing policy statement at the earliest opportunity.
- 14.36 If having consulted the licensing authority decides that it remains of the opinion set out in the assessment, it must revise the CIA to include a statement to that effect and set out the evidence as to why it remains of that opinion. It will be important for any evidence included in the revised CIA to be robust and relevant to the current problems described. This is likely to involve the collation of fresh or updated evidence of the kind described in the above section on evidence of cumulative impact. The licensing authority must also at this stage publish any other material change to the assessment. For example if the types of premises or area described in the assessment have changed due to a shift in the nature of the problems being experienced or where there is evidence of the emergence of a new type of problem.
- 14.37 In each case the three year period for reviewing a CIA begins with the original date of the publication of the CIA or the date that a CIA was last revised. Where a licensing policy statement as a whole is due for review, under the five year review period under section 5(4), and this occurs before the end of the three year CIA review period, licensing authorities may wish to use this as an opportunity to carry out a review of the evidence in support of the CIA. However, licensing authorities are free to carry out consultations and reviews of their CIAs (and/or licensing policy statements) at more regular intervals if they consider this to be appropriate.
- 14.38 As Cumulative Impact Policies were not part of the 2003 Act, there are no transitional provisions that apply to CIPs that were in place before 6 April 2018. However, any

existing CIPs should be reviewed at the earliest practical opportunity to ensure they comply with the legislation. It is recommended that the review should take place within three years of the commencement of the legislation on CIAs or when the licensing policy statement is next due for review, whichever is sooner. This will ensure that any CIPs in place before the commencement of the provisions on CIAs adhere to the principles in the legislation (in particular concerning relevant evidence and consultation).

Effect of cumulative impact assessments

- 14.39 When publishing a CIA a licensing authority is required to set out evidence of problems that are being caused or exacerbated by the cumulative impact of licensed premises in the area described. The evidence is used to justify the statement in the CIA that it is likely that granting further premises licences and/or club premises certificates in that area (limited to a kind described in the assessment), would be inconsistent with the authority's duty to promote the licensing objectives.
- 14.40 In publishing a CIA a licensing authority is setting down a strong statement of intent about its approach to considering applications for the grant or variation of premises licences or club premises certificates in the area described. Having published a CIA a licensing authority must have regard to the assessment when determining or revising its statement of licensing policy. It is therefore expected that, in respect of each relevant application in the area concerned, the licensing authority will be considering whether it is appropriate to make a representation to its committee as a responsible authority in its own right. The CIA does not, however, change the fundamental way that licensing decisions are made. It is therefore open to the licensing authority to grant an application where it considers it is appropriate and where the applicant can demonstrate in the operating schedule that they would not be adding to the cumulative impact. Applications in areas covered by a CIA should therefore give consideration to potential cumulative impact issues when setting out the steps that will be taken to promote the licensing objectives. Where relevant representations are received and a licensing authority decides to grant an application it will need to provide the applicant, the chief officer of police and all parties who made relevant representations with reasons for granting the application and this should include any reasons for departing from their own policy.
- 14.41 The CIA must also stress that it does not relieve responsible authorities (or any other persons) of the need to make relevant representations where they consider it appropriate to do so for the promotion of the licensing objectives. Anyone making a representation may base it on the evidence published in the CIA, or the fact that a CIA has been published. It remains incumbent on all responsible authorities and other persons to ensure that their representations can withstand the scrutiny to which they would be subject at a hearing. As with all licensing applications under the 2003 Act, if there are no representations, the licensing authority must grant the application in terms that are consistent with the operating schedule submitted.
- 14.42 The absence of a CIA does not prevent any responsible authority or other person making representations on an application for the grant or variation of a licence on the grounds that the premises will give rise to a negative cumulative impact on one or more of the licensing objectives, However, in each case it would be incumbent on the person making the representation to provide relevant evidence of cumulative impact.

14.43 As noted above, CIAs may apply to the impact of a concentration of any licensed premises, including those licensed for the sale of alcohol on or off the premises, and premises licensed to provide late night refreshment. When establishing its evidence base for publishing a CIA, licensing authorities should be considering the contribution to cumulative impact made by different types of premises within its area, in order to determine the appropriateness of including different types of licensed premises within the CIA.

Limitations on special policies relating to cumulative impact

- 14.44 A CIA should never be absolute. Statements of licensing policy should always allow for the circumstances of each application to be considered properly and for applications that are unlikely to add to the cumulative impact on the licensing objectives to be granted. After receiving relevant representations in relation to a new application for or a variation of a licence or certificate, the licensing authority must consider whether it would be justified in departing from its CIA in the light of the individual circumstances of the case. The impact can be expected to be different for premises with different styles and characteristics. For example, while a large nightclub or high capacity public house might add to problems of cumulative impact, a small restaurant or a theatre may not. If the licensing authority decides that an application should be refused, it will still need to show that the grant of the application would undermine the promotion of one or more of the licensing objectives and that appropriate conditions would be ineffective in preventing the problems involved.
- 14.45 CIAs should never be used as a ground for revoking an existing licence or certificate when representations are received about problems with those premises. Where the licensing authority has concerns about the effect of activities at existing premises between midnight and 6am on the promotion of the licensing objectives in a specific area, it may introduce an Early Morning Alcohol Restriction Order (EMRO) if there is sufficient evidence to do so (see chapter 17). The "cumulative impact" on the promotion of the licensing objectives of a concentration of licensed premises should only give rise to a relevant representation when an application for the grant or variation of a licence or certificate is being considered.
- 14.46 CIAs must not impose quotas based on either the number of premises or the capacity of those premises. This is because quotas that indirectly have the effect of predetermining the outcome of any application would have no regard to the individual characteristics of the premises concerned.

Other mechanisms for controlling cumulative impact

- 14.47 Once away from the licensed premises, a minority of consumers will behave badly. To enable the general public to appreciate the breadth of the strategy for addressing these problems, statements of policy should also indicate the other mechanisms both within and outside the licensing regime that are available for addressing such issues. For example:
 - planning control;
 - positive measures to create a safe and clean town centre environment in partnership with local businesses, transport operators and other departments of the local authority, including best practise schemes such as Best Bar None, Pubwatch or BIDs;

- Community Protection Notices;
- the provision of CCTV surveillance in town centres, taxi ranks, provision of public conveniences open late at night, street cleaning and litter patrols;
- powers of local authorities to designate parts of the local authority area as places where alcohol may not be consumed publicly;
- the confiscation of alcohol from adults and children in designated areas;
- police enforcement of the general law concerning disorder and anti-social behaviour, including the issuing of fixed penalty notices;
- prosecution for the offence of selling alcohol to a person who is drunk (or allowing such a sale);
- Raising a contribution to policing the late night economy through the Late Night Levy.
- Early Morning Alcohol Restriction Orders (see Chapter 17).
- 14.48 As part of its licensing policy statement, the licensing authority may also wish to consider the use of alternative approaches such as fixed closing times, staggered closing times and zoning. Such policy restrictions would need to be evidence-based and would be subject to the merits of each case in accordance with what is appropriate for the promotion of the licensing objectives. The licensing authority would be expected to justify the use of such measures as an appropriate means of managing problems in its area.

Public Spaces Protection Order

- 14.49 The Designated Public Place Order (DPPO) has been replaced by the Public Spaces Protection Order (PSPO) in the Anti-social Behaviour Crime and Policing Act 2014¹⁵. PSPOs can be used to restrict the drinking of alcohol in a public space where this has or is likely to have a detrimental effect on the quality of life on those in the locality, be persistent or continuing in nature, and unreasonable. Before making a PSPO, a council must consult the local police. DPPOs will continue to be valid for a period of three years following commencement of the PSPO in October 2014. Once that three year period expires, they will be treated as a PSPO and enforceable as such. Where a local authority occupies or manages premises, or where premises are managed on its behalf, and it licenses that place for alcohol sales, the PSPO will not apply when the licence is being used for alcohol sales (or 30 minutes after), but the place will be subject to the PSPO at all other times¹⁶. This allows local authorities to promote community events while still using a PSPO to tackle the problems of anti-social drinking.
- 14.50 It should be noted that when one part of a local authority seeks a premises licence of this kind from the licensing authority, the licensing committee and its officers must consider the matter from an entirely neutral standpoint. If relevant representations are made, for example, by local residents or the police, they must be considered fairly by the committee. Anyone making a representation that is genuinely aggrieved by a positive decision in favour of a local authority application by the licensing authority would be entitled to appeal to the magistrates' court and thereby receive an

¹⁵ For full guidance on the PSPO please see the statutory guidance on the 2014 Act:

https://www.gov.uk/government/publications/anti-social-behaviour-crime-and-policing-bill-anti-social-behaviour

¹⁶ Licensed premises in general are exempt from the effect of a PSPO.

independent review of any decision.

Licensing hours

- 14.51 With regard to licensing hours, the Government acknowledges that different licensing approaches may be appropriate for the promotion of the licensing objectives in different areas. The 2003 Act gives the licensing authority power to make decisions regarding licensed opening hours as part of the implementation of its licensing policy statement and licensing authorities are best placed to make such decisions based on their local knowledge and in consultation with other responsible authorities. However, licensing authorities must always consider each application and must not impose predetermined licensed opening hours, without giving individual consideration to the merits of each application.
- 14.52 Statements of licensing policy should set out the licensing authority's approach regarding licensed opening hours and the strategy it considers appropriate for the promotion of the licensing objectives in its area. The statement of licensing policy should emphasise the consideration which will be given to the individual merits of an application. The Government recognises that licensed premises make an important contribution to our local communities, and has given councils a range of tools to effectively manage the different pressures that licensed premises can bring. In determining appropriate strategies around licensed opening hours, licensing authorities cannot seek to restrict the activities of licensed premises where it is not appropriate for the promotion of the licensing objectives to do so.

Children

- 14.53 It is an offence under the 2003 Act to:
 - permit children under the age of 16 who are not accompanied by an adult to be present on premises being used exclusively or primarily for supply of alcohol for consumption on those premises under the authorisation of a premises licence, club premises certificate or where that activity is carried on under the authority of a TEN; and
 - to permit the presence of children under 16 who are not accompanied by an adult between midnight and 5am at other premises supplying alcohol for consumption on the premises under the authority of any premises licence, club premises certificate or TEN.
- 14.54 Outside of these hours, the offence does not prevent the admission of unaccompanied children under 16 to the wide variety of premises where the consumption of alcohol is not the exclusive or primary activity. This does not mean that children should automatically be admitted to such premises and the following paragraphs are therefore of great importance notwithstanding the offences under the 2003 Act. The expression 'exclusively or primarily' should be given its ordinary and natural meaning in the context of the particular circumstances.
- 14.55 Where it is not clear that the business is predominately for the sale and consumption of alcohol, operators and enforcement agencies should seek to clarify the position before enforcement action is taken. Mixed businesses may be more difficult to classify and in such cases operators and enforcement agencies should consult where appropriate about their respective interpretations of the activities taking place on the premises

before any moves are taken which might lead to prosecution.

- 14.56 The 2003 Act does not automatically permit unaccompanied children under the age of 18 to have free access to premises where the consumption of alcohol is not the exclusive or primary activity or to the same premises even if they are accompanied, or to premises where the consumption of alcohol is not involved. Subject only to the provisions of the 2003 Act and any licence or certificate conditions, admission will always be at the discretion of those managing the premises. The 2003 Act includes no presumption of giving children access but equally, no presumption of preventing their access to licensed premises. Each application and the circumstances of individual premises must be considered on their own merits.
- 14.57 A statement of licensing policy should not seek to limit the access of children to any premises unless it is appropriate for the prevention of physical, moral or psychological harm to them (please see Chapter 2). It may not be possible for licensing policy statements to anticipate every issue of concern that could arise in respect of children in relation to individual premises and therefore the individual merits of each application should be considered in each case.
- 14.58 A statement of licensing policy should make clear the range of alternatives which may be considered for limiting the access of children where that is appropriate for the prevention of harm to children. Conditions which may be relevant in this respect are outlined in paragraph 2.27.
- 14.59 Statements of policy should also make clear that conditions requiring the admission of children to any premises cannot be attached to licences or certificates. Where no licensing restriction is appropriate, this should remain a matter for the discretion of the individual licence holder, club or premises user.
- 14.60 Venue operators seeking premises licences and club premises certificates should consider including such prohibitions and restrictions in their operating schedules particularly where their own risk assessments have determined that the presence of children is undesirable or inappropriate.

Responsible authority and children

14.61 A statement of licensing policy should indicate which body the licensing authority judges to be competent to act as the responsible authority in relation to the protection of children from harm. This may be the local authority social services department, the Local Safeguarding Children Board or other competent body as agreed locally. It would be practical and useful for statements of licensing policy to include descriptions of the responsible authorities in any area and appropriate contact details.

Children and cinemas

14.62 The statement of policy should make clear that in the case of premises giving film exhibitions¹⁷, the licensing authority will expect licence holders or clubs to include in their operating schedules arrangements for restricting children from viewing age-restricted films classified according to the recommendations of the British Board of Film

¹⁷ See paragraphs 16.22-16.24 in relation to the licensing exemption for an exhibition of a film on community premises.

Classification or the licensing authority itself (see paragraphs 10.59 to 10.60).

Integrating strategies

- 14.63 It is recommended that statements of licensing policy should provide clear indications of how the licensing authority will secure the proper integration of its licensing policy with local crime prevention, planning, transport, tourism, equality schemes, cultural strategies and any other plans introduced for the management of town centres and the night-time economy. Many of these strategies are not directly related to the promotion of the licensing objectives, but, indirectly, impact upon them. Co-ordination and integration of such policies, strategies and initiatives are therefore important.
- 14.64 Statements of Licensing Policy should consider the prevalence, prevention and reporting of sexual harassment and misconduct and broader violence against women and girls crimes.

Planning and building control

- 14.65 The statement of licensing policy should indicate that planning permission, building control approval and licensing regimes will be properly separated to avoid duplication and inefficiency. The planning and licensing regimes involve consideration of different (albeit related) matters. Licensing committees are not bound by decisions made by a planning committee, and vice versa. However, as set out in chapter 9, licensing committees and officers should consider discussions with their planning counterparts prior to determination with the aim of agreeing mutually acceptable operating hours and scheme designs.
- 14.66 There are circumstances when, as a condition of planning permission, a terminal hour has been set for the use of premises for commercial purposes. Where these hours are different to the licensing hours, the applicant must observe the earlier closing time. Premises operating in breach of their planning permission would be liable to prosecution under planning law. Proper integration should be assured by licensing committees, where appropriate, providing regular reports to the planning committee.
- 14.67 Where there is an application for planning permission, the National Planning Policy Framework expects new development can be integrated effectively with existing businesses and community facilities (such as places of worship, pubs, music venues and sports clubs). Existing businesses and facilities should not have unreasonable restrictions placed on them as a result of development permitted after they were established. Where the operation of an existing business or community facility could have a significant adverse effect on new development (including changes of use) in its vicinity, the applicant (or 'agent of change') should be required by the local planning authority to provide suitable mitigation before the development has been completed.

Promotion of equality

14.68 A statement of licensing policy should recognise that the Equality Act 2010 places a legal obligation on public authorities to have due regard to the need to eliminate unlawful discrimination, harassment and victimisation; to advance equality of opportunity; and to foster good relations, between persons with different protected

characteristics. The protected characteristics are age, disability, gender reassignment, pregnancy and maternity, race, religion or belief, sex, and sexual orientation.

14.69 Public authorities are required to publish information at least annually to demonstrate their compliance with the Equality Duty. The statement of licensing policy should refer to this legislation, and explain how the Equality Duty has been complied with. Further guidance is available from Government Equalities Office and the Equality and Human Rights Commission.

Administration, exercise and delegation of functions

- 14.70 The 2003 Act provides that the functions of the licensing authority (including its determinations) are to be taken or carried out by its licensing committee (except those relating to the making of a statement of licensing policy or where another of its committees has the matter referred to it). The licensing committee may delegate these functions to sub-committees consisting of three members of the committee, or in appropriate cases to officials supporting the licensing authority. Where licensing functions are not automatically transferred to licensing committees, the functions must be carried out by the licensing authority as a whole and not by its executive. Statements of licensing policy should indicate how the licensing authority intends to approach its various functions. Many of the decisions and functions will be purely administrative in nature and statements of licensing policy should underline the principle of delegation in the interests of speed, efficiency and cost-effectiveness.
- 14.71 The 2003 Act does not prevent the development by a licensing authority of collective working practices with other parts of the local authority or other licensing authorities for work of a purely administrative nature, e.g. mail-outs. In addition, such administrative tasks may be contracted out to private businesses. But any matters regarding licensing decisions must be carried out by the licensing committee, its sub-committees or officers.
- 14.72 Where, under the provisions of the 2003 Act, there are no relevant representations on an application for the grant of a premises licence or club premises certificate or police objection to an application for a personal licence or to an activity taking place under the authority of a temporary event notice, these matters should be dealt with by officers in order to speed matters through the system. Licensing committees should receive regular reports on decisions made by officers so that they maintain an overview of the general situation. Although essentially a matter for licensing authorities to determine themselves, it is recommended that delegation should be approached in the following way:

Table: Recommended Delegation of Functions

Matters to be dealt with	Sub-Committee	Officers
Application for personal licence	If a police or immigration enforcement objection	If no objection made
Application for personal licence with unspent convictions	If a police objection	If no objection made
Application for premises licence/club premises certificate	If a relevant representation made	If no relevant representation made
Application for provisional statement	If a relevant representation made	If no relevant representation made
Application to vary premises licence/club premises certificate	If a relevant representation made	If no relevant representation made
Application to vary designated premises supervisor	If a police objection	All other cases
Request to be removed as designated premises supervisor		All cases
Application for transfer of premises licence	If a police objection	All other cases
Applications for interim authorities	If a police objection	All other cases
Application to review premises licence/club premises certificate	All cases	
Decision on whether a representation is irrelevant frivolous vexatious etc		All cases
Decision to object when local authority is a consultee and not the relevant authority considering the application	All cases	
Determination of an objection to a temporary event notice	All cases	
Determination of application to vary premises licence at community premises to include alternative licence condition	If a police objection	All other cases
Decision whether to consult other responsible authorities on minor variation application		All cases
Determination of minor variation application		All cases

15. Licence fees

15.1 The 2003 Act requires a licensing authority to suspend a premises licence or club premises certificate if the annual fee is not paid when it is due. However, this does not apply immediately if the payment was not made before or at the time of the due date because of an administrative error, or because the holder disputed liability for the fee before or at the time of the due date. In either of these cases, there is a grace period of 21 days. This period is intended to allow the licensing authority and the licence or certificate holder an opportunity to resolve the dispute or error. If the dispute or error is not resolved during this 21-day period, the licence will be suspended.

Dispute

15.2 The 2003 Act describes a dispute as one relating to liability to pay the fee at all or relating to its amount. In either case, the licence or certificate holder must notify the licensing authority of the dispute on or before the date on which the fee to which it relates becomes due.

Administrative error

15.3 There is no definition of "administrative error" in the 2003 Act, but it can include an error on the part of the licensing authority, the licence or certificate holder, or any other person. Therefore, "administrative error" will be given its plain, ordinary meaning. An example might be where post has been misdirected.

Suspension

- 15.4 If a licensing authority suspends a licence or certificate, it must notify the holder in writing and specify the date on which the suspension takes effect; this date must be at least two working days after the day the authority gives the notice. It should be noted that this is the minimum period only, and licensing authorities may consider applying longer periods. The authority may wish to inform the police and other responsible authorities that the licence or certificate has been suspended.
- 15.5 A suspension ceases to have effect on the day on which the licensing authority receives payment of the outstanding fee from the licence or certificate holder. To enable the licence holder to demonstrate that the licence has been reinstated, the licensing authority is required to give the holder written acknowledgment of receipt as soon as practicable following receipt, and:

a) if payment was received on a working day, no later than the end of the next working day, or

b) if payment was received on a day when the authority is not working, no later than the end of the second working day after the day on which the fee was received.

15.6 Licensing authorities may wish to consider requesting, in the notice of suspension mentioned above, that subsequent payment of the outstanding fee may be made in such manner as would most expeditiously bring it to the attention of the authority. The licensing authority may also wish to inform the police and other responsible authorities that the licence or certificate has been reinstated.

Effects of suspension

15.7 A premises licence or certificate that has been suspended does not have effect to authorise licensable activities. However, it can for example be subject to a hearing or, in the case of a premises licence, an application for transfer. The licence will nevertheless only be reinstated when the outstanding fee has been paid. Formally, the debt is owed by the holder who held the licence at the time it was suspended. However, it may be more likely in practice that the new holder will actually make the payment. The suspension of licences and certificates is only applicable to unpaid annual fees that become due after sections 55A and 92A of the 2003 Act came into force on 25 April 2012. In the case of a licence or certificate where more than one payment year has been missed (since the coming into force of sections 55A and 92A) payment of the outstanding fee in relation to each year will be required to reinstate the licence.

Additional fees for large scale events

- 15.8 It should be noted that premises licences for large scale events do not automatically attract the higher fee levels set out in the fee regulations made under the 2003 Act, which must be paid in addition to the standard application or variation fees when the premises licence relates to activities attracting the attendance of 5,000 or more. Venues that are permanent or purpose built or structurally altered for the activity are exempt from the additional fee.
- 15.9 Regulations prescribe that the additional fee for large scale events would not be payable where the premises is a structure which is not a vehicle, vessel or moveable structure, and has been constructed or structurally altered to allow:
 - the proposed licensable activities to take place;
 - the premises to be modified temporarily, from time to time, if relevant for the proposed licensable activities;
 - the proposed number of people on the premises at any one time; and
 - the premises to be used in a manner which complies with the operating schedule.
- 15.10 The full details of where the additional fee is applicable can be found in regulations on the Government's legislation website <u>www.legislation.gov.uk</u>.

16. Regulated entertainment

Types of regulated entertainment

- 16.1 Schedule 1 to the 2003 Act sets out what activities are regarded as the provision of regulated entertainment and when they are licensable and those activities which are not and therefore exempt from the regulated entertainment regime. Changes to regulated entertainment took effect on 6 April 2015.
- 16.2 The descriptions of entertainment activities licensable under the 2003 Act are:
 - a performance of a play;
 - an exhibition of a film;
 - an indoor sporting event;
 - a boxing or wrestling entertainment;
 - a performance of live music;
 - any playing of recorded music;
 - a performance of dance; and
 - entertainment of a similar description to a performance of live music, any playing of recorded music or a performance of dance.
- 16.3 To be licensable, one or more of these activities needs to be provided for the purpose (at least partly) of entertaining an audience; has to be held on premises made available for the purpose of enabling that activity; and must also either:
 - · take place in the presence of a public audience, or
 - be provided exclusively for the members of a club or for the club members and their guests, or
 - where that activity takes place in private, be the subject of a charge made with a view to profit.
- 16.4 Guidance around what constitutes audiences and private events is at paragraphs 16.11-16.14.

Overview of circumstances in which entertainment activities are not licensable

- 16.5 There are a number of exemptions that mean that a licence (or other authorisation¹⁸) under the 2003 Act is not required. This Guidance cannot give examples of every eventuality or possible entertainment activity that is not licensable. However, the following activities are examples of entertainment which are not licensable:
 - activities which involve participation as acts of worship in a religious context;
 - activities in places of public religious worship;
 - education teaching students to perform music or to dance;
 - the demonstration of a product for example, a guitar in a music shop;
 - the rehearsal of a play or performance of music for a private audience where no

charge is made with a view to making a profit;

- Morris dancing (or similar)¹⁹;
- Incidental music the performance of live music or the playing of recorded music if it is incidental to some other activity²⁰;
- Incidental film an exhibition of moving pictures if it is incidental to some other activity²¹;
- A spontaneous performance of music, singing or dancing;
- Garden fetes or similar if not being promoted or held for purposes of private gain;
- Films for advertisement, information, education or in museums or art galleries;
- Television or radio broadcasts as long as the programme is live and simultaneous;
- Vehicles in motion at a time when the vehicle is not permanently or temporarily parked;
- Games played in pubs, youth clubs etc. (e.g. pool, darts and table tennis);
- Stand-up comedy; and
- Provision of entertainment facilities (e.g. dance floors)²².
- 16.6 As a result of deregulatory changes that have amended the 2003 Act²³, no licence is required for the following activities:
 - Plays: no licence is required for performances between 08.00 and 23.00 on any day, provided that the audience does not exceed 500.
 - Dance: no licence is required for performances between 08.00 and 23.00 on any day, provided that the audience does not exceed 500²⁴.
 - Films: no licence is required for 'not-for-profit' film exhibition held in community premises between 08.00 and 23.00 on any day provided that the audience does not exceed 500 and the organiser (a) gets consent to the screening from a person who is responsible for the premises; and (b) ensures that each such screening abides by age classification ratings.
 - Indoor sporting events: no licence is required for an event between 08.00 and 23.00 on any day, provided that those present do not exceed 1000.
 - Boxing or wrestling entertainment: no licence is required for a contest, exhibition or display of Greco-Roman wrestling, or freestyle wrestling between 08.00 and 23.00 on any day, provided that the audience does not exceed 1000.
 - Live music: no licence permission is required for:
 - a performance of unamplified live music between 08.00 and 23.00 on any day, on any premises.
 - a performance of amplified live music between 08.00 and 23.00 on any day on premises authorised to sell alcohol for consumption on those premises,

¹⁹ Including any live music or playing of recorded music as an integral part of a performance of Morris dancing, or similar activity. ²⁰ See paragraphs 16.57-16.61

²¹ See paragraphs 16.65-16.68

²² This was previously licensable under the 2003 Act until the commencement of the Live Music Act 2012.

²³ The Live Music Act 2012 ("2012 Act") <u>http://www.legislation.gov.uk/ukpga/2012/2</u>; Licensing Act 2003 (Descriptions of Entertainment) (Amendment) Order 2013 ("2013 Order") <u>http://www.legislation.gov.uk/uksi/2013/1578/contents/made</u>; The Legislative Reform (Entertainment Licensing) Order 2014 ("2014 Order")

http://www.legislation.gov.uk/uksi/2014/3253/introduction/made; and the Deregulation Act 2015 ("2015 Act").

²⁴ But see paragraphs 16.47 -16.48 in relation to dance that is adult entertainment and remains licensable.

provided that the audience does not exceed 500²⁵.

- a performance of amplified live music between 08.00 and 23.00 on any day, in a workplace²⁶ that does not have a licence, provided that the audience does not exceed 500.
- a performance of amplified live music between 08.00 and 23.00 on any day, in a church hall, village hall, community hall, or other similar community premises, that is not licensed by a premises licence to sell alcohol, provided that (a) the audience does not exceed 500, and (b) the organiser gets consent for the performance from a person who is responsible for the premises.
- a performance of amplified live music between 08.00 and 23.00 on any day, at the non-residential premises of (i) a local authority, or (ii) a school, or (iii) a hospital, provided that (a) the audience does not exceed 500, and (b) the organiser gets consent for the performance on the relevant premises from: (i) the local authority concerned, or (ii) the school or (iii) the health care provider for the hospital.
- Recorded Music: no licence permission is required for:
 - any playing of recorded music between 08.00 and 23.00 on any day on premises authorised to sell alcohol for consumption on those premises, provided that the audience does not exceed 500²⁷.
 - any playing of recorded music between 08.00 and 23.00 on any day, in a church hall, village hall, community hall, or other similar community premises, that is not licensed by a premises licence to sell alcohol, provided that (a) the audience does not exceed 500, and (b) the organiser gets consent for the performance from a person who is responsible for the premises.
 - any playing of recorded music between 08.00 and 23.00 on any day, at the non- residential premises of (i) a local authority, or (ii) a school, or (iii) a hospital, provided that (a) the audience does not exceed 500, and (b) the organiser gets consent for the performance on the relevant premises from: (i) the local authority concerned, or (ii) the school proprietor or (iii) the health care provider for the hospital.
- Cross activity exemptions: no licence is required between 08.00 and 23.00 on any day, with no limit on audience size for:
 - any entertainment taking place on the premises of the local authority where the entertainment is provided by or on behalf of the local authority;
 - any entertainment taking place on the hospital premises of the health care provider where the entertainment is provided by or on behalf of the health care provider;
 - any entertainment taking place on the premises of the school where the entertainment is provided by or on behalf of the school proprietor; and
 - any entertainment (excluding films and a boxing or wrestling entertainment)

 ²⁵ Provided that a number of other important conditions are satisfied (see paragraphs 16.326-16.29).
 ²⁶ The Live Music Act 2012 provides that if premises are licensed under the 2003 Act, they cannot also be treated as a workplace for the purpose of the 2012 Act.

²⁷ Provided that a number of other important conditions are satisfied, see paragraphs 16.33.

taking place at a travelling circus, provided that (a) it takes place within a moveable structure that accommodates the audience, and (b) that the travelling circus has not been located on the same site for more than 28 consecutive days.

- 16.7 The deregulatory changes mean that, for example, an indoor sporting event that takes place between 07.00 and 23.30 on a particular day is licensable in respect of activities taking place between 07.00-08.00 and 23.00-23.30. Similarly, where the audience for a performance of dance fluctuates, those activities are licensable if, and for so long as, the number of people in the audience exceeds 500. If organisers are uncertain as to audience sizes or if audience migration is likely²⁸, it might be easier and more flexible to secure an appropriate authorisation. Examples of where a Temporary Event Notice (TEN)²⁹ could still be required include if the activity is the playing of recorded music or the exhibition of a film that requires an authorisation; or if the entertainment is not authorised by an existing licence or certificate and its conditions.
- 16.8 Of course, anyone involved in the organisation or provision of entertainment activities whether or not any such activity is licensable under the 2003 Act must comply with any applicable duties that may be imposed by other legislation relevant to the event (e.g. in areas such as crime and disorder, fire, health and safety, noise, nuisance and planning).³⁰ Any such person should take steps to be aware of relevant best practice, and may find responsible authorities a useful source of expert support and advice.
- 16.9 The various effects of the changes made to entertainment licensing by the set of deregulatory changes between 2012 and 2015³¹ are described in greater detail in subsequent paragraphs:
 - Music entertainment, see in particular paragraphs: 16.20-16.21; 16.26-16.33; and 16.36-16.44;
 - Plays, dance, and indoor sporting events, see in particular paragraphs: 16.34-16.35 and 16.45-16.48;
 - Local authority, hospital and school premises, see in particular paragraphs: 16.16-16.20
 - Community premises, see in particular paragraphs: 16.21-16.24
 - Circuses, see in particular paragraph 16.25
 - Boxing or wrestling entertainment, see in particular paragraphs: 16.49-16.51.

²⁸ See paragraph 16.12

²⁹ See chapter 7

³⁰ See paragraphs 16.70-16.72 in relation to other licensing regimes

³¹ An entertainment activity may meet the conditions of more than one exemption

General circumstances in which entertainment activities are licensable

16.10 An authorisation for regulated entertainment is always required for entertainment activities that take place before 08.00 or after 23.00, unless exempted under any other provision of the 2003 Act, as amended³².

Audience

- 16.11 For the purposes of regulated entertainment, the term "audience" refers to any person for whose entertainment (at least in part) any licensable activities are provided. An audience member need not be, or want to be, entertained: what matters is that an audience is present³³ and that the purpose of the licensable activity is (at least in part) intended to entertain any person present³⁴. The audience will not include performers, together with any person who contributes technical skills in substantial support of a performer (for example, a sound engineer or stage technician), during any associated activities. This includes setting up before the entertainment, reasonable breaks (including intervals) between activities and packing up thereafter. Similarly, security staff and bar workers will not form part of the audience while undertaking their duties.
- 16.12 More than one entertainment activity (or for a single activity, more than one performance or event) can be held concurrently, provided that the audience for each such performance or event does not exceed the threshold at which such a performance or event becomes licensable. In some circumstances, there will be a clear distinction between performances or events; for example, their taking place in separate rooms or on separate floors. However, organisers will have to ensure that audiences do not grow or migrate, so that the audience exceeds the relevant limit for any one performance or event at any time. If there is the possibility of audience migration, it might be easier and more flexible to secure an appropriate authorisation.

Private events

- 16.13 Events held in private are not licensable unless those attending are charged for the entertainment with a view to making a profit (including raising money for charity). For example, where a party is held for friends in a private dwelling featuring amplified live music, if a charge or contribution is made solely to cover the costs of the entertainment, the activity is not regulated entertainment. Similarly, any charge made to the organiser of a private event by musicians, other performers, or their agents does not of itself make that entertainment licensable it would only do so if the guests attending were themselves charged by the organiser for that entertainment with a view to achieving a profit. The fact that this might inadvertently result in the organiser making a profit would be irrelevant, as long as there had not been an intention to make a profit.
- 16.14 Schedule 1 to the 2003 Act also makes it clear that before entertainment is regarded as

³² See examples at paragraph 16.5

³³ In some circumstances, such as un-ticketed live music events, a degree of judgement may be required as to whether persons are part of an audience. Factors to consider could include whether a person is within the perimeter of the premises, the audible range of the performance, and their visibility of the entertainment. In order to meet the definition of an entertainment activity in the 2003 Act, the activity must take place in the presence of an audience and be provided for the purpose, or for purposes which include the purpose of, entertaining that audience.

³⁴ For example, a darts championship competition hosted in part to entertain an audience could be a licensable activity, but a pub game of darts played for the enjoyment of the participants is not licensable.

being provided for consideration, a charge has to be:

- made by or on behalf of a person concerned with the organisation or management of the entertainment; and
- paid by or on behalf of some or all of the persons for whom the entertainment is provided.

Circumstances in which entertainment activities are no longer licensable

16.15 No licence is required for certain entertainment activities on specified premises, as follows:

Local authorities, hospital healthcare providers and school proprietors: cross-entertainment activity exemption

- 16.16 No licence is required for any entertainment provided by or on behalf of a local authority, health care provider, or school proprietor to the extent that it takes place on defined premises, between 08.00-23.00 on any day provided that:
 - for entertainment provided by, or on behalf of, a local authority it takes places on premises in which that authority has a relevant property interest, or is in lawful occupation;
 - for entertainment provided by, or on behalf of, a health care provider it takes place on any premises forming part of a hospital in which the provider has a relevant property interest, or is in lawful occupation; and
 - for entertainment provided by, or on behalf of, a school proprietor it takes place on the premises of the school.
- 16.17 This Guidance cannot give examples of every eventuality where entertainment is not licensable under this exemption through being provided "by or on behalf of". It will depend on the facts in each case. However, the following are examples of activities that are not usually considered to be licensable under this exemption:
 - Any entertainment activity hosted by a local authority on their own premises, where there is a significant relationship between the local authority and the provider of the entertainment (e.g. principal and agent);
 - Any entertainment activity organised on a local authority's behalf on that local authority's premises by a cultural trust in discharge of a local authority's discretionary power to arrange entertainment provision and support for the arts, including festivals and celebrations.
 - Any entertainment activity organised by a healthcare provider on their own hospital premises in partnership with a hospital charity;
 - Any entertainment event on school premises organised by the Parent Teacher Association (PTA) to benefit the school.
- 16.18 It is for the local authority, health care provider or school proprietor to determine whether, and on what basis, they can (or wish) to provide entertainment activity under this exemption, including consideration of issues around fundraising, profit making, governance or use of public funds. However a pure hire of premises by a third party³⁵

does not constitute the provision of an entertainment event "on behalf of" a local authority, healthcare provider, or school proprietor and nor does commercial entertainment which the local authority³⁶ merely facilitates through providing a public space³⁷.

16.19 All the terms used in this exemption, such as "local authority", "health care", "health care provider", "hospital", "school", "school premises", "school proprietor", "domestic premises" and "relevant property interest" are defined in the 2014 Order³⁸.

Local authority, hospital and school premises: third party music entertainment

- 16.20 No licence is required for a performance of live music or the playing of recorded music on local authority, hospital or school premises, that are not domestic premises, between 08.00-23.00 on any day provided that:
 - it is performed in front of an audience of no more than 500 people; and
 - a person concerned in the organisation or management of the music entertainment has obtained the prior written consent³⁹ of the local authority, health care provider or school proprietor (as appropriate) for that entertainment to take place. It is for these "trusted providers" to determine whether, or not, they wish to make their premises available for music entertainment by a 3rd party and on what terms they deem it appropriate.

Community premises: music entertainment

- 16.21 No licence is required for a performance of live music or the playing of recorded music on community premises⁴⁰, between 08.00-23.00 on any day provided that:
 - the community premises are **not** authorised, by a premises licence or club premises certificate, to be used for the supply of alcohol for consumption on the premises⁴¹;
 - the music entertainment is in the presence of an audience of no more than 500 people; and
 - a person concerned in the organisation or management of the music entertainment has obtained the prior written consent⁴² of the management committee of the premises, or if there is no management committee, a person who has control of the premises in connection with the carrying on by that person of a trade, business or other undertaking, or failing that a person with a relevant property interest in the premises.

³⁶ Or healthcare provider or school proprietor.

³⁷ The exemption would similarly not apply, for example, to a commercial company operating on premises belonging to a local authority under a long term lease.

³⁸ See footnote 23

³⁹ This requirement is designed to ensure that those responsible for the premises hosting the entertainment have considered and approved the effect of the event on other users of their premises and the wider community.

⁴⁰ The definition of community premises is covered in paragraphs 4.56-4.60 of this Guidance. A community premises is likely to be multi-functional and 'other similar building' within the definition cannot be stretched to ordinarily include a public house, a bingo hall, or other business premises or private property.

⁴¹ Where a community premises is licensed for the supply of alcohol by a premises licence (or exceptionally a club premises certificate), then any performance of live music or the playing of recorded music on relevant alcohol licensed premises may be subject to the conditional deregulation described in paragraphs 16.26-16.33.

⁴² See footnote 39

Community premises: exhibition of film

- 16.22 No licence⁴³ is required for an exhibition of a film on community premises⁴⁴ between 08.00-23.00 on any day provided that:
 - the film entertainment is not provided with a view to profit⁴⁵;
 - the film entertainment is in the presence of an audience of no more than 500 people;
 - the admission of children is subject to such restrictions as are necessary to comply with the recommendation issued by the BBFC or relevant licensing authority regarding the admission of children; and
 - a person concerned in the organisation or management of the exhibition of the film has obtained the prior written consent⁴⁶ of the management committee of the premises, or if there is no management committee, a person who has control of the premises in connection with the carrying on by that person of a trade, business or other undertaking, or failing that a person with a relevant property interest in the premises.
- 16.23 Under this exemption, one condition is that the film entertainment is not being provided with a view to profit⁴⁷. An entry charge does not of itself make the film entertainment licensable; it is whether the organiser intended to make a profit (that includes raising money for charity). A charge or contribution that is made solely to cover the costs of the film screening⁴⁸ is consistent with 'not being provided with a view to profit'. The 'not with a view to profit' condition applies solely to the activity of exhibiting the film under this exemption. A charge with a view to making a profit may legitimately be levied for any other activity or event that is distinct from film admission, such as the provision of refreshments, film talks, or a social event.
- 16.24 This community film exemption is also conditional on those responsible having in place operating arrangements that ensure that the age rating for the film is implemented by means of a suitable child admission policy⁴⁹. How this is achieved is a matter for the organisation or social group exhibiting the film. For example, they may operate a membership subscription scheme which pays for entry to all titles in a season and is limited to adults. It could be a children's film club with a policy of only showing films that are suitable for all by being rated 'U' by the BBFC. Alternatively, the organisers could sell tickets to the public and ensure that children are only permitted to attend in accordance with any age rating for the film i.e. a door admissions policy linked to proof of age.

⁴³ However, see paragraph 16.70 in relation to copyright

⁴⁴ See footnote 40

⁴⁵ See paragraph 16.23

⁴⁶ See footnote 39

⁴⁷ 'not provided with a view to profit' is the inverse of 'with a view to profit' mentioned in paragraph 16.13

⁴⁸ Legitimate costs of a film screening would include overheads directly relevant to providing the film entertainment (e.g. premises hire, film hire, equipment etc.)

⁴⁹ See 3rd bullet point in paragraph 16.22

Travelling circuses

- 16.25 Where types of entertainment are present in a performance by a travelling circus⁵⁰ they will not be licensable provided that certain qualifying conditions are met⁵¹. The qualifying conditions are that:
 - the entertainment is not an exhibition of a film or a boxing or wrestling entertainment;
 - the entertainment takes place between 08.00 and 23.00 on the same day;
 - the entertainment takes place wholly within a moveable structure and the audience present is accommodated wholly inside that moveable structure; and
 - the travelling circus has not been located on the same site for more than 28 consecutive days.

Live music

- 16.26 Live music is licensable:
 - where a performance of live music whether amplified or unamplified takes place before 08.00 or after 23.00 on any day;
 - where a performance of amplified live music does not take place either on relevant licensed premises, or at a workplace that is not licensed other than for the provision of late night refreshment;
 - where a performance of amplified live music takes place at a time when the relevant licensed premises are not open for the purposes of being used for the sale or supply of alcohol for consumption on the premises⁵²;
 - where a performance of amplified live music takes place at relevant licensed premises, or workplaces⁵³, in the presence of an audience of more than 500 people⁵⁴; or
 - where a licensing authority intentionally removes the effect of the deregulation provided for by the 2003 Act when imposing a condition on a premises licence or club premises certificate as a result of a licence review⁵⁵.
- 16.27 In any of the above circumstances, unless the performance of live music is appropriately authorised by a premises licence, club premises certificate or TEN, allowing it to take place could lead to enforcement action and, where relevant, a review of the alcohol licence or certificate.
- 16.28 A public performance of live unamplified music that takes place between 08.00 and 23.00 on the same day no longer requires a licence under the 2003 Act in any location. An exception to this is where a specific condition related to live music is included following a review of the premises licence or club premises certificate in respect of relevant licensed premises.

⁵⁰ 'Travelling circus' is defined in the 2014 Order as meaning a circus which travels from site to site for the purpose of giving performances. Musical entertainment at a travelling fairground is likely to be incidental to the main attractions and rides that are not themselves regulated entertainment.

⁵¹ There is no audience limit for this exemption, but the conditions are designed to ensure that deregulation does not have unintended consequences for the licensing objectives - e.g. only bona fide travelling circuses qualify.

⁵² See Chapter 3 of this Guidance

⁵³ See paragraph 16.31

 $^{^{\}rm 54}$ The 2014 Order substituted "500" for "200" that was in the 2012 Act

⁵⁵ See paragraphs 16.55-16.56

16.29 As a result of the amendments to the 2003 Act, section 177 of the 2003 Act now only applies to performances of dance⁵⁶.

Key terms used in relation to live music

- 16.30 Under the live music provisions, "music" includes vocal or instrumental music or any combination of the two. "Live music" is a performance of live music in the presence of an audience which it is intended to entertain. While a performance of live music can include the playing of some recorded music, 'live' music requires that the performance does not consist entirely of the playing of recorded music without any additional (substantial and continual) creative contribution being made. So, for example, a drum machine or backing track being used to accompany a vocalist⁵⁷ or a band would be part of the performance of amplified live music. The performance of a DJ who is merely playing tracks would not be classified as live music, but it might if he or she was performing a set which largely consisted of mixing recorded music in a live performance to create new sounds⁵⁸. There will inevitably be a degree of judgement as to whether a performance is live music (or recorded music) and organisers of events should check with their licensing authority if this consideration is relevant to whether the activity is authorised by a licence or certificate. In the event of a dispute about whether a performance is live music or not, it will be for the licensing authority initially and ultimately, for the courts to decide in the individual circumstances of any case.
- 16.31 A "workplace" is as defined in regulation 2(1) of the Workplace (Health, Safety and Welfare) Regulations 1992 and is anywhere that is made available to any person as a place of work. It is a very wide term which can include outdoor spaces, as well as the means of entry and exit.
- 16.32 A "relevant licensed premises" for the purposes of this chapter is one which is authorised to sell or supply alcohol for consumption on the premises by a premises licence or club premises certificate. Premises cannot benefit from the deregulation introduced by the 2012 Act by virtue of holding an authorisation for the sale or supply of alcohol under a TEN.⁵⁹

Recorded music

- 16.33 No licence is required for recorded music where it takes place on premises which are authorised by a premises licence or club premises certificate to be used for the supply of alcohol for consumption on the premises. However, recorded music remains licensable:
 - where the playing of recorded music takes places before 08.00 or after 23.00 on any day;
 - where the playing of recorded music takes place at a time when the relevant licensed premises are not open for the purposes of being used for the sale or supply of alcohol for consumption on the premises;⁶⁰
 - where the playing of recorded music takes place at relevant licensed premises in the

⁵⁶ See paragraph 2.13. Post the 2013 Order, Section177 can be relevant to a performance of dance after 23.00 on any day

 $^{^{\}rm 57}$ Karaoke is generally classed as a performance of live music

⁵⁸ This would include 'scratching'

⁵⁹ TENs are covered in chapter 7

⁶⁰ See Chapter 3 of this Guidance

presence of an audience of more than 500 people; and

 where a licensing authority intentionally removes the effect of the deregulation provided for by the 2003 Act (as amended).⁶¹

Plays and dance

- 16.34 No licence is required for a performance of a play or dance to the extent that certain qualifying conditions⁶² are satisfied. However a performance of a play or dance remains licensable:
 - where the performance takes places before 08.00 or after 23.00 on any day; or
 - where the performance takes place in the presence of an audience of more than 500 people.

Indoor Sport

- 16.35 No licence is required for an indoor sporting event to the extent that certain qualifying conditions⁶³ are satisfied. However an indoor sporting event remains licensable:
 - where the event takes places before 08.00 or after 23.00 on any day;
 - where the event takes place in the presence of more than 1000 spectators.

Licence conditions

Live Music or recorded music

- 16.36 Any existing licence conditions⁶⁴ (or conditions added on a determination of an application for a premises licence or club premises certificate⁶⁵) which relate to live music or recorded music remain in place, but are **suspended** between the hours of 08.00 and 23.00 on the same day where the following conditions are met:
 - at the time of the music entertainment, the premises are open for the purposes of being used for the sale or supply of alcohol for consumption on the premises;
 - if the music is amplified, it takes place before an audience of no more than 500 people; and
 - the music takes place between 08.00 and 23.00 on the same day.
- 16.37 Whether a licence condition relates to live or recorded music will be a matter of fact in each case. In some instances, it will be obvious that a condition relates to music and will be suspended, for example "during performances of live music all doors and windows must remain closed". In other instances, it might not be so obvious: for example, a condition stating "during performances of regulated entertainment all doors and windows must remain closed" would be suspended insofar as it relates to music between 08.00 and 23.00 on the same day to an audience of up to 500, but the condition would continue to apply if there was regulated entertainment after 23.00.
- 16.38 More general licence conditions (e.g. those relating to overall management of potential noise nuisance) that are not specifically related to the provision of entertainment (e.g.

⁶¹ This would be by way of imposing a condition on a premises licence or club premises certificate as a result of a licence review, see paragraphs 16.55-16.56

⁶² See paragraphs 16.6 and 16.45-16.48

⁶³ See paragraph 16.6.

⁶⁴ In relation to relevant licensed premises, see paragraph 16.32

⁶⁵ See paragraphs 16.39-16.40

signage asking patrons to leave quietly) will continue to have effect.

- 16.39 Chapter 9 of this Guidance sets out how a licensing authority must determine applications for a new licence or to vary an existing premises licence. Licence conditions imposed, in accordance with paragraphs 9.42 to 9.44, for live or recorded music activities will only apply if the activity meets the criteria of having more than 500 people present, and/or the activities are taking place between 23.00 and 08.00.
- 16.40 These conditions will, in effect, be suspended between 08.00 and 23.00 if a performance of live music or the playing of recorded music takes place before an audience of 500 people or fewer, but will remain on the face of the licence for when these activities may take place under other circumstances.
- 16.41 Where a performance of live music or the playing of recorded music on relevant licensed premises is not licensable, it remains possible for anyone to apply for a review of a licence or certificate, if there are appropriate grounds to do so.⁶⁶

Beer gardens

- 16.42 Beer gardens are often included as part of a premises licence or club premises certificate. Live amplified music that takes place in a beer garden is exempt from licensing requirements, provided the beer garden is included in the licence or certificate applying to the relevant licensed premises, and the performance takes place between 08.00 and 23.00 on the same day before an audience of 500 people or fewer.
- 16.43 Where a beer garden does not form part of the relevant licensed premises and so is not included in plans attached to a premises licence or club premises certificate, it is nevertheless very likely that it will be a workplace⁶⁷. Paragraph 12B of Schedule 1 to the 2003 Act says that a performance of live music in a workplace that does not have a licence (except to provide late night refreshment) is not regulated entertainment if it takes place between 08.00 and 23.00 on the same day in front of an audience of no more than 500 people. Note that the exemption in paragraph 12B does not apply to the playing of recorded music.
- 16.44 However, a licensing authority may, where justified⁶⁸, impose a licence condition that relates to the performance of live music in an unlicensed beer garden being served by any associated premises licence or club premises certificate. Provided such a condition is lawfully imposed, it takes effect in accordance with its terms.

Plays, dance and indoor sport

- 16.45 Where qualifying conditions are satisfied⁶⁹, any current licence condition that relates to a performance of a play or dance, or an indoor sporting event for which a licence is no longer required will (except in the circumstances described in the next paragraph) have no effect.
- 16.46 Where, however, these non-licensable activities take place at the same time as other

⁶⁶ See paragraphs 16.55-16.56 and chapter 11.

⁶⁷ A beer garden is one example of a potential workplace, see paragraph 16.31. Whether other outdoor spaces (such as seating adjacent to a premises, a smoking shelter, or a car park) constitute a workplace, part of the licensed premises, or neither, will be a matter of fact in each case.

⁶⁸ Including on a licence review

⁶⁹ See paragraph 16.6

activities for which a licence is required (e.g. the sale or supply of alcohol for consumption on the premises), conditions included in a licence may nevertheless apply to the non-licensable activities in the circumstances set out in paragraphs 16.52 and 16.53 (conditions relating to other non-licensable activities).

- 16.47 Dance that is sufficiently sexual in nature continues to be regulated. Performances of dance which are "relevant entertainment" within the meaning of the Local Government (Miscellaneous Provisions) Act 1982 ("the 1982 Act") are not deregulated, regardless of the size of the audience or the time of day. "Relevant entertainment" is defined in the 1982 Act as a live performance or live display of nudity that, ignoring financial gain, can be assumed to be provided solely or principally for the purpose of sexually stimulating any member of the audience.⁷⁰
- 16.48 In almost all cases where a performance of dance is potentially licensable as both the provision of **relevant** entertainment (under the 1982 Act) and **regulated** entertainment (under the 2003 Act), the 1982 Act disapplies the entertainment licensing regime in the 2003 Act in favour of its stricter regime for the control of sex establishments. However, an authorisation under the 2003 Act rather than the 1982 Act will continue to be required where:
 - the premises are not licensed as a sex entertainment venue under the 1982 Act, and
 - relevant entertainment has been provided at those premises on no more than 11 occasions in any 12 month period, with none of those occasions lasting longer than 24 hours or taking place within a month of any other such occasion.

Boxing or wrestling entertainment and conditions relating to combined fighting sports

- 16.49 An indoor boxing or wrestling entertainment cannot also be an indoor sporting event, and any contest, exhibition or display that combines boxing or wrestling with one or more martial arts ('combined fighting sports') is whether indoors or not a boxing or wrestling entertainment.
- 16.50 Where a premises licence or club premises certificate purports to authorise a boxing or wrestling entertainment or combined fighting sports as an 'indoor sporting event', the 2013 Order provides that the authorisation will be treated as having authorised those activities as a boxing or wrestling entertainment. Those activities will continue to be subject to any relevant conditions attached to that authorisation.
- 16.51 A contest, exhibition or display of Greco-Roman wrestling, or of freestyle wrestling, between two participants (regardless of their sex) does not require a licence provided that certain qualifying conditions are met. They are that:
 - it takes place in the presence of no more than 1,000 spectators;
 - it takes place between 08.00 and 23.00 on the same day; and
 - it take place wholly inside a building and the spectators present at that entertainment are accommodated wholly inside that building.

Conditions relating to other non-licensable activities

- 16.52 If appropriate for the promotion of the licensing objectives, and if there is a link to remaining licensable activities, conditions that relate to non-licensable activities can be added to or altered on that premises licence or club premises certificate at review following problems occurring at the premises. This has been a feature of licence conditions since the 2003 Act came into force. A relevant example could be the use of conditions relating to large screen broadcasts of certain sporting events which, combined with alcohol consumption, could create a genuine risk to the promotion of the licensing objectives. It is also not uncommon for licence conditions relating to the sale of alcohol to restrict access to outside areas, such as unlicensed beer gardens, after a certain time.
- 16.53 Similarly, while karaoke no longer needs licensing as the provision of entertainment facilities⁷¹ (and will generally be classed as a performance of live music⁷²) it might, for example, be possible on review to limit the use or volume of a microphone made available for customers on an 'open-mic' night (which encompasses more than just live music), if a problem had occurred because of customers purchasing alcohol for consumption on the premises becoming louder and less aware of causing noise nuisance later in the evening. Another example might be a condition restricting access to a dance floor at certain times, where the presence of customers in close proximity who had been consuming alcohol on the premises had led to serious disorder. In the first instance it is for the licensing authority to satisfy itself that a particular condition is appropriate and lawful in each case.

Removing licence conditions

16.54 A licence holder who wishes to remove conditions relating to activities that are no longer licensable⁷³ may apply to the licensing authority for a licence variation. In the course of considering such applications, licensing authorities should remove such conditions unless there are sufficiently serious and specific concerns about the effects of hosting deregulated entertainment activities along with the remaining licensable activities taking place in the premises.

Licence reviews: Live and recorded music

16.55 On a review of a premises licence or club premises certificate, section 177A(3) of the 2003 Act permits a licensing authority to lift the suspension74 and give renewed effect to an existing condition relating to music. Similarly, under section 177A(4), a licensing authority may add a condition relating to music as if music were regulated entertainment, and as if that premises licence or club premises certificate licensed the music. In both instances the condition should include a statement that Section 177A does not apply to the condition.

⁷¹ As a result of the 2012 Act

⁷² Karaoke is generally classed as a 'performance of live music', and provided that it is genuinely taking place, it is not likely to be classed as the 'playing of recorded music'

⁷³ A licence holder may favour removal if the previously licensable activity has "grandfather" conditions that are out of date, or unclear. Where such a condition in relation to live or recorded music is suspended, then the licence holder may wish to avoid any prospect of the condition being given renewed effect (were a suspension to be lifted following a licence review).
⁷⁴ See paragraph 16.36.

16.56 An application for a review in relation to relevant premises can be made by a licensing authority, any responsible authority or any other person. Applications for review must still be relevant to one or more of the licensing objectives and meet a number of further requirements⁷⁵.

Incidental music

- 16.57 The performance of live music or playing of recorded music is not regulated entertainment under the 2003 Act if it is 'incidental' to another activity "which is not itself a description of entertainment falling within paragraph 2" of Schedule 1 to the 2003 Act.
- 16.58 The incidental music exemption can apply to an indoor sporting event or a performance of a play or dance for which no licence is required, as it takes place between 08.00 and 23.00 on the same day and before an audience which does not exceed the relevant limit. This is because such an activity is no longer a description of entertainment within the meaning of paragraph 2 of Schedule 1 to the 2003 Act. This means that, while a performance of live music or the playing of recorded music cannot be incidental to a boxing or wrestling entertainment⁷⁶ such music may be within the scope of the incidental music exemption for an indoor sporting event or performance of a play or dance for which no licence is required.
- 16.59 Whether or not music is "incidental" to another activity will depend on the facts of each case. In considering whether or not live or recorded music is incidental, one relevant factor could be whether, against a background of the other activities already taking place, the addition of music will create the potential to undermine the promotion of one or more of the four licensing objectives of the 2003 Act. Other factors might include some or all of the following:
 - Is the music the main, or one of the main, reasons for people attending the premises and being charged?
 - Is the music advertised as the main attraction?
 - Does the volume of the music disrupt or predominate over other activities, or could it be described as 'background' music?
- 16.60 Conversely, factors which would not normally be relevant in themselves include:
 - the number of musicians, e.g. an orchestra providing incidental music at a large exhibition;
 - whether musicians are paid;
 - whether the performance is pre-arranged; and
 - whether a charge is made for admission to the premises.
- 16.61 In any disputed case, it will be for the licensing authority initially and, ultimately, for the courts to consider whether music is "incidental" in the individual circumstances of any case.

⁷⁵ See Chapter 11 for more information about reviews under the 2003 Act.

⁷⁶ And as such, the music entertainment needs to be authorised under the 2003 Act. This would include music during a Greco-Roman or freestyle wrestling entertainment. While, depending on the circumstances, the Greco-Roman or freestyle wrestling may, or may not be licensable, it is still within the 'description' of a 'wrestling entertainment'.

Busking

- 16.62 Busking or street performance is the practice of performing in public spaces for money. Performances are not limited to music or singing and can take the form of a wide range of activities that people find entertaining.
- 16.63 Busking is generally **not** licensable under the 2003 Act as:
 - it often occurs in a place that is not a premises made available (at least in part) for the purposes of providing entertainment⁷⁷;
 - the entertainment is usually incidental to another activity, such as shopping or sightseeing, as there are few circumstances in which anyone would go out specifically to watch buskers; and
 - any unamplified live music is not licensable between 08.00 and 23.00⁷⁸.
- 16.64 Local authorities may have policies on busking, including codes of conduct or permit regimes and occasionally byelaws and legislation specific to a local authority although many localities have no policy or restrictions.

Incidental film

- 16.65 An exhibition of a film within the meaning of paragraph 15 of Schedule 1 to the 2003 Act is not regulated entertainment if it is 'incidental' to another activity "which is not itself a description of entertainment falling within paragraph 2" of Schedule 1 to the 2003 Act.
- 16.66 The incidental film exemption can apply to an indoor sporting event or a performance of a play or dance for which no licence is required as it takes place between 08.00 and 23.00 on the same day before an audience which does not exceed the relevant limit. Such activities would no longer be a description of entertainment within the meaning of paragraph 2 of Schedule 1 to the 2003 Act. This means that, while any exhibition of moving pictures cannot be incidental to a boxing or wrestling entertainment⁷⁹, such film displays may be within the scope of the incidental film exemption for an indoor sporting event or performance of a play or dance for which no licence is required.⁸⁰
- 16.67 Whether or not an exhibition of moving pictures is "incidental" to another activity will depend on the facts of each case. In considering whether or not film is incidental, one relevant factor could be whether, against a background of the other activities already taking place, the addition of an exhibition of moving images will create the potential to undermine the promotion of one or more of the four licensing objectives of the 2003 Act. This would mean that if the BBFC or the relevant licensing authority has given an age rating to a film, video, or music video, then to qualify for the "incidental film" licensing exemption, the admission of children to the premises will need to be restricted in accordance with the appropriate age rating. But that is one aspect of one relevant factor. Other factors to consider in assessing whether film is incidental might include some or all of the following:

⁷⁷ See paragraph 16.3

⁷⁸ See paragraph 16.28

⁷⁹ And as such, the film display needs to be authorised under the 2003 Act. This would include moving pictures during a Greco-Roman or freestyle wrestling entertainment. While, depending on the circumstances, the Greco- Roman or freestyle wrestling may, or may not be licensable, it is still within the 'description' of a 'wrestling entertainment'.

⁸⁰ For indoor sporting events, it takes place between 08.00 and 23.00 in front of an audience which does not exceed 1,000.

- Is the film the main, or one of the main, reasons for people attending the premises and being charged?
- Is the film advertised as the main attraction?
- Does the screening of the film predominate over other activities, or could it be described as 'background' images?
- Does the appearance of moving pictures within another entertainment activity, for which no licence is required (e.g. a performance of a play or dance⁸¹), undermine the promotion of the licensing objectives?
- 16.68 In any disputed case, it will be for the licensing authority initially and, ultimately, for the courts to consider whether film is "incidental" in the individual circumstances of any case.

Entertainment activity provided as part of childcare

16.69 Entertainment activity that is provided as part of childcare will generally not be licensable. This includes entertainment activity in a nursery or private home. In addition, paragraph 5 of Schedule 1 to the 2003 Act includes a licensing exemption for an exhibition of a film where the main purpose is to provide education. Education will generally include all forms of pre-school child and day care. Furthermore, an exhibition of a film, or the playing of live or recorded music, will generally be incidental to the activity of childcare and so the incidental music and film exemption in paragraph 7 of Schedule 1 will also apply⁸². This will generally be the case for any entertainment activity organised as part of wraparound childcare⁸³, including breakfast clubs, after school clubs or holiday clubs linked to the child's school or based in the local community.

Other Licensing regimes

Copyright

16.70 The deregulation of licensing for the provision of entertainment under the 2003 Act does not remove the requirement for licences for the use of copyright works. Entertainment activities as described in paragraph 16.2 may require music and screening licences for example. The acquisition of such licences will make the entertainment compliant with the Copyright, Designs and Patents Act 1988.⁸⁴

Leafleting

16.71 The deregulation of entertainment licensing does not remove the prohibition on the unauthorised distribution of free printed matter in an area that has been designated under Schedule 3A of the Environmental Protection Act 1990⁸⁵. The organisers of the event or entertainment may need to obtain consent from the relevant principal litter authority before giving out free printed promotional material (leaflets, flyers, cards etc.) in a public place in certain areas.

⁸¹ See paragraph 16.34

⁸² See paragraphs 16.57-16.61 and 16.65-16.68

⁸³ Childcare for school aged children around traditional school hours

⁸⁴ Further information concerning how to acquire such licences can be found on at <u>https://www.gov.uk/licence-to-play-live-or-recorded-music</u>

⁸⁵ http://www.legislation.gov.uk/ukpga/1990/43/schedule/3A

Child performers

16.72 Child performance legislation⁸⁶ requires that a licence must be obtained from a child's home local authority before a child can take part in certain types of performance and activities. A licence may be required whether or not any payment is made for the child to perform. The deregulation of entertainment licensing does not alter the regulations on when children can take part in performances⁸⁷.

⁸⁶ Children and Young Persons Acts 1933 and 1963; The Children (Performances and Activities) (England) Regulations 2014 and The Children (Performances and Activities) (Wales) Regulations 2015

17. Early morning alcohol restriction orders

General

- 17.1 This chapter provides guidance to licensing authorities about Early Morning Alcohol Restriction Orders ("EMROs"). The power conferred on licensing authorities to make, vary or revoke an EMRO (or propose to take any of these steps) is set out in sections 172A to 172E of the 2003 Act. The exercise of the licensing authority's functions may be delegated by its committee to a sub-committee, other than the decision to make, vary or revoke an EMRO (which is exercised by its full council). This power enables a licensing authority to prohibit the sale of alcohol for a specified time period between the hours of 12am and 6am in the whole or part of its area, if it is satisfied that this would be appropriate for the promotion of the licensing objectives.
- 17.2 EMROs are designed to address recurring problems such as high levels of alcoholrelated crime and disorder in specific areas at specific times; serious public nuisance; and other instances of alcohol-related anti-social behaviour which is not directly attributable to specific premises.
- 17.3 An EMRO:
 - applies to the supply of alcohol authorised by premises licences, club premises certificates and temporary event notices;
 - applies for any period beginning at or after 12am and ending at or before 6am. It does not have to apply on every day of the week, and can apply for different time periods on different days of the week;
 - applies for a limited or unlimited period (for example, an EMRO could be introduced for a few weeks to apply to a specific event);
 - applies to the whole or any part of the licensing authority's area;
 - will not apply to any premises on New Year's Eve (defined as 12am to 6am on 1 January each year);
 - will not apply to the supply of alcohol by those who provide hotel or similar accommodation to their residents between 12 am and 6am, provided the alcohol is sold at those times only through mini-bars and/or room service; and
 - will not apply to a relaxation of licensing hours by virtue of an order made under section 172 of the 2003 Act.

The EMRO process

- 17.4 An EMRO can apply to the whole or part of the licensing authority's area. The area may, for example, comprise a single floor of a shopping complex or exclude premises which have clearly demonstrated to the licensing authority that the licensable activities carried on there do not contribute to the problems which form the basis for the proposed EMRO.
- 17.5 If the licensing authority has already published a Cumulative Impact Assessment ("CIA") (see Chapter 14 of this Guidance), it should consider the relationship between the CIA and proposed EMRO area, and the potential overall impact on its local licensing policy.

17.6 Introducing an EMRO is a licensing function. Therefore, this is not the responsibility of a council's executive. The final decision to make an EMRO (or to vary or revoke one) must be made by the full council of the licensing authority. Any preceding steps, including advertising the proposed EMRO, holding hearings and making a determination to put before the full council for its final decision, are for the licensing committee of the licensing authority. The licensing committee may delegate these steps to the licensing sub- committee or officers as it sees fit.

Evidence

- 17.7 When establishing its evidence base for making an EMRO, a licensing authority⁸⁸ may wish to consider the approach set out in paragraphs 14.29 to 14.33 of this Guidance which includes indicative types of evidence, although this should not be considered an exhaustive list of the types of evidence which may be relevant. These matters are not necessarily determinative. They include but are not necessarily limited to:
 - local crime and disorder statistics, including statistics on specific types of crime and crime hotspots, statistics on local anti-social behaviour offences,
 - environmental health complaints, particularly in relation to litter and noise;
 - complaints recorded by the local authority, which may include complaints raised by local residents or residents' associations;
 - residents' questionnaires;
 - trends in licence applications, particularly trends in applications by types of premises and terminal hours;
 - · changes in terminal hours of premises;
 - capacities of different premises at different times of night and the expected concentrations of drinkers who will be expected to be leaving premises at different times.
- 17.8 Before a licensing authority makes a determination to recommend to the full council that it makes a proposed EMRO, it should be satisfied that it has sufficient evidence to demonstrate that making the EMRO would be appropriate for the promotion of the licensing objectives. The requirement to take an evidence-based decision to promote the licensing objectives should enable licensing authorities to draw on their experience from other licensing decisions they make under the 2003 Act, such as the determination of applications for the grant of premises licences. The licensing authority should consider evidence from partners, including from responsible authorities and local Community Safety Partnerships, alongside its own evidence, to determine whether an EMRO would be appropriate for the promotion of the licensing objectives.

Introducing an EMRO

17.9 An EMRO is a powerful tool which will prevent licensed premises in the area to which the EMRO relates from supplying alcohol during the times at which the EMRO applies. The licensing authority should consider whether other measures may address the problems that they have identified as the basis for introducing an EMRO. As set out in

⁸⁸ The final decision to make an EMRO (or to vary or revoke one) must be made by the full council of the licensing authority. However, all preceding steps, including advertising the proposed EMRO, holding hearings and making a determination to put before the full council for its final decision, are for the licensing committee of the licensing authority. The licensing committee may delegate these steps to the licensing sub-committee or officers as it sees fit.

paragraphs 9.42-9.44 of this Guidance, when determining whether a step is appropriate to promote the licensing objectives, a licensing authority is not required to decide that no lesser step will achieve the aim. They should, however, consider whether taking that step is reasonable, justified and proportionate. The introduction of an EMRO may have far-reaching, wider impacts on the socio-economic circumstances in an area. In considering whether the introduction of an EMRO is an appropriate step to promote the licensing objectives, based on whether this is reasonable, justified and proportionate, a licensing authority may hold informal discussions early in the process with a range of interested partners; these include, but are not limited to, premises that may be affected by the introduction of the EMRO. Other measures that could be taken instead of making an EMRO might include:

- working in partnership with licensed premises on voluntary measures and encouraging the creation of business-led best practice schemes in the area;
- reviewing licences of specific problem premises;
- introducing a CIA;
- use of the new closure power in the Anti-social Behaviour, Crime and Policing Act 2014 which replaces section 161 of the 2003 Licensing Act. This new closure power can be used to protect victims and communities by quickly closing premises that are causing nuisance or disorder. Further guidance on this power can be found on the gov.uk website, under the Anti-social Behaviour, Crime and Policing Act: anti-social behaviour guidance;
- use of other mechanisms such as those set out in paragraph 14.47 of this Guidance
- 17.10 If the licensing authority has identified a problem in a specific area attributable to the supply of alcohol at two or more premises in that area, and has sufficient evidence to demonstrate that it is appropriate for the promotion of the licensing objectives, it can propose making an EMRO. The licensing authority should first decide on the matters which must be the subject of the proposal. These are:
 - the days (and periods on those days) on which the EMRO would apply;
 - the area to which the EMRO would apply;
 - the period for which the EMRO would apply (if it is a finite period); and
 - the date from which the proposed EMRO would apply.

In relation to the date when it plans to introduce the EMRO, the licensing authority should note that this may change when it is specified in the final order without the need to formally consult on the new date (as if it was a new proposal), provided it does not adversely affect any person as described in paragraph 17.21.

Advertising an EMRO

17.11 The proposed EMRO must be advertised. The licensing authority should include a short summary of the evidence and the manner in which representations can be made in the document, as well as the details of the proposed EMRO. The proposal must be advertised for at least 42 days (a reference in this Chapter to a period of "days" means a period comprising calendar days and not only working days). The licensing authority must publish the proposal on its website and in a local newspaper. If no newspaper exists, it must be published in a local newsletter, circular or similar document. The licensing authority must also send a notice of the proposal to all affected people in its

area. They are:

- holders of (and applicants for) premises licences or club premises certificates to which the proposed EMRO would apply;
- premises users in relation to TENs to which the proposed EMRO would apply;
- those who have received a provisional statement in respect of a premises to which the proposed EMRO would apply.
- 17.12 Licensing authorities must, moreover, display a notice of the proposal in the area to which the EMRO would apply, in a manner which is likely to bring the proposal to the attention of those who may have an interest in it.
- 17.13 The licensing authority should also inform responsible authorities in its area and neighbouring licensing authorities of its proposal to make an EMRO. It may also like to consider what further steps could be taken, in any particular case, to publicise the proposal in order to draw it to the wider attention of any other persons who are likely to have an interest in it.

Representations

- 17.14 Those who are affected by a proposed EMRO, responsible authorities or any other person have 42 days (starting on the day after the day on which the proposed EMRO is advertised) to make relevant representations. To be considered a relevant representation, a representation must:
 - be about the likely effect of the making of the EMRO on the promotion of the licensing objectives;
 - be made in writing in the prescribed form and manner, setting out the EMRO to which it relates and the nature of the representation;
 - be received within the deadline; and
 - if made by a person other than a responsible authority, not be frivolous or vexatious.

Chapter 9 of this Guidance gives further advice on determining whether a representation is frivolous or vexatious. Representations can be made in relation to any aspect of the proposed EMRO. If a licensing authority decides that a representation is not relevant, it should consider informing the person who has made that representation.

- 17.15 Responsible authorities may wish to make representations, as may affected persons (as set out in the above paragraph).
- 17.16 Others may also wish to make representations about the proposed EMRO. These persons could include, but are not limited to:
 - residents;
 - employees of affected businesses;
 - owners and employees of businesses outside the proposed EMRO area; and
 - users of the late night economy.

Hearings

17.17 If a relevant representation or representations are received, the licensing authority must hold a hearing to consider them (unless the authority and anyone who has made representations agree that this is unnecessary). The licensing authority should consider,

based on the number of relevant representations received by it and any other circumstances it considers appropriate, whether to hold the hearing over several days, which could be arranged to take place other than on consecutive working days.

- 17.18 As described in paragraph 17.6, a hearing to consider representations in relation to an EMRO may be held by the licensing committee, the licensing sub-committee or an officer of the licensing authority. It is recommended, however, that such hearings be conducted by the licensing committee or sub-committee.
- 17.19 Licensing committees or sub-committees⁸⁹ should be familiar with the hearing process as it has similarities with other processes under the 2003 Act. Further guidance on hearings can be found in Chapter 9 of this Guidance (paragraphs 9.31 to 9.41). However, licensing authorities should note the following key points in relation to a hearing about a proposed EMRO:
 - the hearing must be commenced within 30 working days, beginning with the day after the end of the period during which representations may be made;
 - the hearing does not have to take place on consecutive working days, if an authority considers this to be necessary to enable it to consider any of the representations made by a party or if it considers this approach to be in the public interest;
 - a licensing committee or sub-committee must make its determination within 10 working days of the conclusion of the hearing;
 - the licensing committee or sub-committee is not required to notify those making representations of its determination; and
 - the licensing authority may give notices in relation to a hearing by electronic means provided it is satisfied that the text of the notice is capable of being accessed by the recipient, it is legible in all material respects and is capable of being reproduced in written form (e.g. printed by the recipient).
- 17.20 The licensing committee or sub-committee will determine the manner in which the hearing will be conducted in accordance with the Licensing Act 2003 (Hearings) Regulations 2005. If a licensing committee or sub-committee determines that a representation is frivolous or vexatious, it must notify in writing the person who made the representation.
- 17.21 As a result of the hearing, the licensing committee or sub-committee has three options:
 - to determine that the proposed EMRO is appropriate for promotion of the licensing objectives;
 - to determine that the proposed EMRO is not appropriate for the promotion of the objectives and therefore that the process should be ended;
 - to determine that the proposed EMRO should be modified.

In the final case, if the authority proposes that the modified EMRO should differ from the initial proposal in relation to the area specified, any day not in the initial proposal or the period of any day specified, the authority should advertise what is in effect a new proposal to make an EMRO in the manner described above, so that further representations may be made. However, there will be cases in which it may be possible

to modify the terms of a proposed EMRO without being required to formally consult. This may arise where the modified terms would not have a more adverse effect on any person (primarily, if not solely, licensed premises which will be subject of the EMRO) than the EMRO in the terms in which it was originally proposed. Such cases may include the following:

- the modification may shorten the period during which the EMRO would apply or reduce the number of days on which it applies, provided these periods were a part of the original proposal. For example, a change to an EMRO applying on Fridays and Saturdays to just applying on Saturdays may not require re advertisement and consultation of the EMRO design, whereas changing the days the EMRO applies on from Fridays and Saturdays to just Thursdays might reasonably be expected to require further consultation;
- the date on which the EMRO commences is later than that described in the original proposal;

Licensing authorities should consider very carefully in each case (including in relation to legality) whether further consultation on a proposed modification to the EMRO is necessary.

Final EMRO

- 17.22 If the licensing authority determines that the proposed EMRO is appropriate for the promotion of the licensing objectives, its determination must be put to the full council for its final decision. There is no time specified in legislation by which the full council must make this decision. This is intended to reflect the fact that the licensing authority may only meet in full council infrequently.
- 17.23 The matters set out in the final order must be no different from the matters set out in the proposal to make the order, subject to the caveat described above in paragraph 17.21. The order must be set out in the prescribed form and contain the prescribed content.
- 17.24 No later than 7 days after the day on which the EMRO is made, the licensing authority must send a notice to all affected persons of the EMRO, and make the order available for at least 28 days on its website and by displaying a notice in the EMRO area. A licensing authority should retain details of the EMRO on its website for as long as the EMRO is in force. It is recommended that the licensing authority advises neighbouring licensing authorities and the Secretary of State that the order has been made, the nature of the order and when (and for how long) it will take effect.
- 17.25 The licensing authority should monitor the effectiveness of the EMRO to ensure it continues to be appropriate for the promotion of the licensing objectives and periodically review whether it is appropriate to continue to apply it. The licensing authority should consider setting out its policy in relation to reviewing EMROs (if any) in its statement of licensing policy.
- 17.26 The variation or revocation of an order requires the licensing authority to undertake the same process as that which applied on its introduction; that is after gathering the appropriate evidence, it advertises its new EMRO proposal, following the process set out above so that those affected and anyone else can make representations.
- 17.27 If an order applies for a finite period, the order will cease to apply on its last day. If the licensing authority wishes to introduce a further (new) EMRO, it must follow the full

process for proposing a new EMRO.

17.28 Licensing authorities should update their statement of licensing policy (in accordance with section 5 of the 2003 Act) to include reference to the EMRO as soon as reasonably possible.

Exceptions to an EMRO

17.29 EMROs will not apply on New Year's Eve in recognition of its status as a national celebration. The supply of alcohol to residents through mini-bars and room service in premises with overnight accommodation will also not be subject to an EMRO.

Enforcement of EMROs

- 17.30 The sale or supply of alcohol in contravention of an EMRO is an 'unauthorised licensable activity' which is an offence under section 136 of the 2003 Act. Moreover, it may result in a closure notice being served on the premises under section 76 of the Anti-social Behaviour, Crime and Policing Act 2014 as a precursor to an application for a closure order under section 80 of that Act (which requires the constable or authority that issued the notice to apply to a magistrates' court not later than 48 hours after the service of the closure notice). This may alternatively, result in the licence being reviewed on crime prevention grounds. Further information on reviews can be found in Chapter 11 of this Guidance.
- 17.31 An EMRO overrides all authorisations to supply alcohol under the 2003 Act (including temporary event notices). It is immaterial whether an authorisation was granted before or after an EMRO was made as there are no authorisations that have the effect of authorising the sale of alcohol during the EMRO period, with the only exception being a licensing hours order made under section 172 of the 2003 Act.

EMRO Process Flowchart



Annex – Counter Terrorism

We have collected here for ease of reference a number of resources relating to public safety at venues and events which licensing authorities may find helpful when considering applications where enhanced security concerns may be relevant.

LGA Councillors Handbook

https://www.local.gov.uk/publications/licensing-act-2003-councillors-handbook-england-andwales-0

Case study from Manchester relating to use of licensing conditions

https://www.local.gov.uk/case-studies/manchester-city-council-security-and-licensing

Supporting links to wider counter terrorism measures which may be helpful for licenced premises and Local Authorities.

https://www.protectuk.police.uk/first-aid-preparedness-organisations

https://www.protectuk.police.uk/advice-and-guidance/response/supporting-information-publicaccess-trauma-first-aid-kits